EXISTENCE OF GENERALIZED BUSEMANN FUNCTIONS AND GIBBS MEASURES FOR RANDOM WALKS IN RANDOM POTENTIALS

SEAN GROATHOUSE, CHRISTOPHER JANJIGIAN, AND FIRAS RASSOUL-AGHA

ABSTRACT. We establish the existence of generalized Busemann functions and Gibbs-Dobrushin-Landford-Ruelle measures for a general class of lattice random walks in random potentials with finitely many admissible steps. This class encompasses directed polymers in random environments, first- and last-passage percolation, and elliptic random walks in both static and dynamic random environments in all dimensions and with minimal assumptions on the random potential.

1. INTRODUCTION

The model of a random walk interacting with a random potential (RWRP) has been a major topic of research in probability over the last half-century. Through various specializations, it encompasses random walks in both static and dynamic random environments, directed polymers in random environments, as well as zero-temperature models of deterministic walks in random environments like directed and undirected firstand last-passage percolation. The objects we investigate, called generalized Busemann functions, have been previously studied in specific instances through the structure of positive harmonic functions [69-71] and the associated Martin boundary [9], infinite volume Gibbs-Dobrushin-Landford-Ruelle measures and geodesics [1, 4-6, 8, 14, 16, 23, 24, 27, 39, 41, 54, 63], increment stationary distributions of directed polymers and stochastic Hamilton-Jacobi equations [1, 4-6, 8, 14, 17, 40, 41, 63], solutions to variational formulas for the limiting free energy and shape function [9, 57, 60, 61, 70], and, in the related stochastic Hamilton-Jacobi setting, correctors for the stochastic homogenization problem [15, 41].

In this introductory section, we will motivate the model and questions we consider informally, ignoring technical complications. A careful treatment follows in the main body of the text. Our model begins with a time-homogeneous reference random walk with finitely many admissible steps on an integer lattice \mathbb{Z}^d . We denote the set of admissible steps by \mathcal{R} , the associated transition mass function by p, and the path law started at $x \in \mathbb{Z}^d$ by P_x . The random walk interacts with a random potential $\{V(\omega, z) : z \in \mathcal{R}\}$ through its position and its increment.

At positive temperature, the quenched (unrestricted-length) point-to-point polymer measure in environment ω is a Gibbsian measure on paths X. which emanate from a site x, defined via the Radon-Nikodym derivative

(1.1)
$$\frac{d\mathbf{Q}_{x,y}^{\beta,\omega}}{d\mathbf{P}_x}(X_{\bullet}) = \frac{e^{-\beta\sum_{k=0}^{\tau_y-1}V(T_{X_k}\omega,X_{k+1}-X_k)}}{Z_{x,y}^{\beta,\omega}}\mathbb{1}_{\{\tau_y<\infty\}}(X_{\bullet}).$$

In the previous expression, $T_x \omega$ is a shift of the environment ω by x, τ_y is the first strictly positive time the path reaches site y, and the *partition function* is

(1.2)
$$Z_{x,y}^{\beta,\omega} = \mathbf{E}_x \left[e^{-\beta \sum_{k=0}^{\tau_y - 1} V(T_{X_k}\omega, X_{k+1} - X_k)} \mathbb{1}_{\{\tau_y < \infty\}} \right]$$

Here E_x is the expectation with respect to P_x , and $\beta \in (0, \infty)$ is interpreted as the *inverse temperature*.

Date: January 16, 2025.

²⁰²⁰ Mathematics Subject Classification. 60K35, 60K37.

Key words and phrases. Busemann functions, cocycle, directed polymer in a random environment, first-passage percolation, Gibbs measure, harmonic function, last-passage percolation, Martin boundary, random polymer, random walk in a random environment, random growth, shape theorem.

S. Groathouse and F. Rassoul-Agha were partially supported by National Science Foundation grants DMS-1811090 and DMS-2054630.

C. Janjigian was partially supported by National Science Foundation grant DMS-2125961.

F. Rassoul-Agha was partially supported by MPS-Simons Fellowship grant 823136.

A straightforward computation checks that these measures are Markovian with one-step transition probabilities from $x \neq y$ to $x + z, z \in \mathcal{R}$, given by

(1.3)
$$p(z)e^{-\beta V(T_x\omega,z)}\frac{Z_{x+z,y}^{\beta,\omega}}{Z_{x,y}^{\beta,\omega}} = p(z)e^{-\beta (V(T_x\omega,z)+\frac{1}{\beta}(\log Z_{x,y}^{\beta,\omega}-\log Z_{x+z,y}^{\beta,\omega}))}.$$

As $\beta \to \infty$, one should expect the measure in (1.1) to converge to a distribution supported on paths which minimize the potential along the path; i.e., the measure is asymptotically supported on paths which optimize

(1.4)
$$-\min_{x,\in \text{supp } P_x} \left\{ \sum_{k=0}^{\tau_y - 1} V(T_{x_k}\omega, x_{k+1} - x_k) \right\} = F_{x,y}^{\infty,\omega} = \lim_{\beta \to \infty} \frac{1}{\beta} \log Z_{x,y}^{\beta,\omega}.$$

This "zero temperature" RWRP model is known in the literature as first-passage percolation (FPP), with the standard undirected Euclidean model arising if P_x is any random walk with the same admissible steps as simple symmetric random walk on \mathbb{Z}^d . In that setting $F_{x,y}^{\infty,\omega}$ is called the passage time. The statement that the transition probabilities in (1.3) sum to one becomes the following local control problem, which can be used to construct minimizing paths, called geodesics in FPP:

$$0 = \min_{z \in \mathcal{R}} \{ V(T_x \omega, x + z) + F_{x,y}^{\infty, \omega} - F_{x+z,y}^{\infty, \omega} \}$$

If one replaces stopping on reaching y with stopping after n steps, then the positive-temperature polymer measure defined through (1.1) is what we call the *quenched restricted-length point-to-level polymer measure*, which is defined via the Radon-Nikodym derivative

(1.5)
$$\frac{d\mathbf{Q}_{x,n}^{\beta,\omega}}{d\mathbf{P}_{x,n}}(X_{\bullet}) = \frac{e^{-\beta\sum_{k=0}^{n-1}V(T_{X_k}\omega, X_{k+1} - X_k)}}{Z_{x,n}^{\beta,\omega}}$$

In the previous expression, $P_{x,n}$ is the restriction of P_x to the first n steps of the walk and

$$Z_{x,n}^{\beta}(\omega) = \mathbb{E}_{x,n} \bigg[e^{-\beta \sum_{k=0}^{n-1} V(T_{X_k}\omega, X_{k+1} - X_k)} \bigg].$$

Again, these measures are Markovian with one-step transition probabilities from x to x + z, where $z \in \mathcal{R}$, given by

(1.6)
$$p(z)e^{-\beta V(T_x\omega,z)}\frac{Z_{x+z,n-1}^{\beta}(\omega)}{Z_{x,n}^{\beta}(\omega)} = p(z)e^{-\beta (V(T_x\omega,z)+\frac{1}{\beta}(\log Z_{x,n}^{\beta}-\log Z_{x+z,n-1}^{\beta}))}.$$

Denote by \mathbb{P} the law of the random environment ω . In general, the above quenched point-to-level polymer measures are not Kolmogorov-consistent as n varies unless it happens to be the case that \mathbb{P} -almost surely for all x and all n,

(1.7)
$$Z_{x,n}^{\beta}(\omega) = 1$$
 or, equivalently, $\sum_{z \in \mathcal{R}} p(z) e^{-\beta V(T_x \omega, z)} = 1.$

If this holds, then the restricted point-to-level model reduces to that of a random walk in a random environment (RWRE).

As previously noted, except in the special case where the model is an RWRE, the measures $(Q_{x,n}^{\beta,\omega})_n$ and $(Q_{x,y}^{\beta,\omega})_y$ discussed above are not in general Kolmogorov-consistent. The domain Markov property satisfied by another family of measures (restricted length point-to-point, defined further down in (2.3)) gives Dobrushin-Landford-Ruelle (DLR) equations which characterize infinite volume Gibbs measures in this setting. One typically expects such measures to arise in the *thermodynamic limit* as the terminal condition (y or n, respectively) tends to infinity. Such a limit is equivalent to the convergence of the transition probabilities defined through (1.3) (or (1.6)). Writing $\beta^{-1} \log Z_{x,y}^{\beta,\omega} = F_{x,y}^{\beta,\omega}$, this, in turn, is equivalent to the convergence of limits of the form

(1.8)
$$B^{\beta,\omega}(x,y) = \lim_{u} \left\{ F^{\beta,\omega}_{x,u} - F^{\beta,\omega}_{y,u} \right\}$$

where the limit is taken as $u \to \infty$ in an appropriate sense. In the zero-temperature setting of FPP, the analogous limits

(1.9)
$$B^{\infty,\omega}(x,y) = \lim_{u} \left\{ F^{\infty,\omega}_{x,u} - F^{\infty,\omega}_{y,u} \right\}$$

are known as *Busemann functions*, which is a name inherited from their interpretation in metric geometry. We keep this terminology in the general model.

The existence of such limits is highly non-trivial to prove in general, with most arguments, even in the planar d = 2 setting, relying on strong and generally unproven hypotheses about the limiting *free energy* $(\beta < \infty)$ or *limit shape* $(\beta = \infty)$. These quantities are defined respectively (in the point-to-point setting) through the limits

$$\Lambda^{\beta}(\xi) = \lim_{n \to \infty} \frac{1}{n} F_{\mathbf{0}, x_n}^{\beta, \omega} \quad \text{and} \quad \Lambda^{\infty}(\xi) = \lim_{n \to \infty} \frac{1}{n} F_{\mathbf{0}, x_n}^{\infty, \omega}$$

where $x_n/n \to \xi \in \mathbb{R}^d$. In the setting of RWRE, the corresponding object has also been studied extensively in the guise of the quenched large deviation rate function for the time *n* position of the path.

If the limits in (1.8) and (1.9) exist, then these Busemann functions satisfy four key properties, the first three of which are inherited immediately from the pre-limit structure of the model:

(i) (Recovery) For $\beta < \infty$,

$$1 = \sum_{z \in \mathcal{R}} p(z) e^{-\beta(V(\omega, z) + \frac{1}{\beta} B^{\beta, \omega}(\mathbf{0}, z))}$$

and for $\beta = \infty$,

$$0 = \min_{z \in \mathcal{P}} \{ V(\omega, z) + B^{\infty, \omega}(\mathbf{0}, z) \}.$$

(ii) (Cocycle)

$$B^{\beta,\omega}(x,y) + B^{\beta,\omega}(y,z) = B^{\beta,\omega}(x,z)$$

(iii) (Shift covariance)

$$B^{\beta,T_z\omega}(x,y) = B^{\beta,\omega}(x+z,y+z).$$

(iv) (Duality) There exists $m \in \partial \Lambda^{\beta}(\xi)$ for some ξ for which

(1.10)
$$\mathbb{E}[B^{\beta,\omega}(x,y)] = m \cdot (x-y).$$

In the duality expression, $\partial \Lambda^{\beta}(\xi)$ denotes the superdifferential of the concave function Λ^{β} . With some convex analysis, one can generally show that duality is actually a consequence of the previous three conditions. We include this condition as part of our (informal) definition to simplify the discussion below. Random fields which satisfy these four properties are called *generalized Busemann functions*. Where such objects exist, they can serve essentially the same role as true Busemann functions defined through the limits in (1.8) and (1.9). In particular, in the setting of polymer models (resp. FPP/LPP), the recovery property of the generalized Busemann functions can be used to construct semi-infinite Markovian path measures (resp. paths). Then, the cocycle property implies that these measures (resp. paths) satisfy the aforementioned DLR equations (resp. are geodesics). Gibbs measures (resp. geodesics) built in this way have extra structure and can be shown to also be consistent with the unrestricted-length point-to-point measures (resp. geodesics) which are discussed above.

As the reader will see in the body, because of how generally we work in this project, the careful statements of our main results are fairly technical. For this reason, we give informal statements in this introduction, with precise statements to follow. Our main technical result, stated carefully in Theorem 4.5 below, is that generalized Busemann functions essentially always exist.

Informal Theorem 1.1. For appropriate choices of the distribution \mathbb{P} of the environment ω and all choices of the reference walk with finitely many admissible steps, for each ξ and $m \in \partial \Lambda^{\beta}(\xi)$, there exist random variables $B^{\beta,\omega}(x,y)$ satisfying the duality (1.10) and the recovery, cocycle, and covariance properties above.

In the statement above, what is meant by "appropriate" is that the environment needs to be appropriately ergodic or mixing and satisfy mild moment hypotheses. Our hypotheses include most models with finitely many admissible steps studied in the literature, with two significant exceptions being degenerate (i.e. not elliptic) random walks in random environments and walks on percolation clusters.

The fourth condition in our informal definition of the generalized Busemann function was called duality because it expresses a Legendre-Fenchel duality (through the concave function Λ^{β}) between the mean vector m of the Busemann function and a direction ξ for which $m \in \partial \Lambda^{\beta}(\xi)$ which appears when one uses the Busemann function to construct Gibbs measures or infinite geodesics. The study of the Gibbs measures and semi-infinite geodesics has been a focus of significant recent attention. We discuss some of these related works below and refer the reader to the introduction of [42] for a more comprehensive discussion. The main result of this paper establishes the existence of such measures and geodesics in a broad setting, while also detailing some of their fundamental properties. The precise statement of our existence result is Theorem 3.14 below. Informally, we can summarize that result as follows.

Informal Theorem 1.2. Under the same hypotheses as Informal Theorem 1.1, for each $x \in \mathbb{Z}^d$, ξ , and $m \in \partial \Lambda^{\beta}(\xi)$, there exists a probability measure $Q_x^{\beta,m,\omega}$ on infinite paths starting at x. This measure satisfies Gibbs consistency with the restricted-length point-to-point measures and, in an appropriate sense, with the unrestricted-length measures. Furthermore, it corresponds to the tilt m through the following dualities:

For almost all ω , $Q_x^{\beta,m,\omega}$ -almost surely, the limit points of $X_n/|X_n|_1$ lie in the set of directions ζ such that $m \in \partial \Lambda^{\beta}(\zeta)$. Similarly, the limit points of X_n/n lie in the set of velocities satisfying the analogous condition with respect to the restricted-length free energy.

The need for understanding consistency with the unrestricted-length measures "in an appropriate sense" in the statement above is because the unrestricted-length measures are not in general self-consistent in the Gibbs sense. We discuss this point further in Appendix B.

Appendix A.1 of the Ph.D. thesis [32] contains some further preliminary results on the general structure of infinite volume Gibbs measures in this general setting, including their equivalence to recovering cocycles.

What we are calling generalized Busemann functions have previously appeared with other interpretations in other settings. Suppose that the model is an irreducible RWRE as in (1.7) and that h solves for all $x \in \mathbb{Z}^d$ and some $c \in \mathbb{R}$,

(1.11)
$$h(x) = \sum_{z} p(z) e^{-V(T_x \omega, z) - c} h(x + z).$$

Then the above equation says that $h(X_n)e^{cn}$ is a (quenched) martingale for the RWRE. When c = 0, this becomes the usual notion of a harmonic function. We call the $c \neq 0$ case a *time-dependent* harmonic function.

If B satisfies the recovery and cocycle properties discussed above, then the function $h(x) = e^{B^{1,\omega}(0,x)}$ is harmonic (with c = 0). More generally, we will see that the (space-time) generalized Busemann functions (see Remark 4.2 below for a careful definition) coming from the restricted-length measures in (1.5) define time-dependent harmonic functions as in (1.11). Such time-dependent harmonic functions have previously been used to study RWRE in [69–71]. One could conversely start with a covariant positive harmonic function with c = 0 and construct a generalized Busemann function by setting $B(x, y) = \log h(x) - \log h(y)$. A similar statement holds for space-time generalized Busemann functions in the case of $c \neq 0$.

Through this connection, generalized Busemann functions can be seen to be closely related to the Martin boundary theory of the RWRE and generalizations to the RWRP model. In this interpretation, the infinite volume polymer measures discussed in Informal Theorem 1.2 are the Markov processes generated by Doob h-transforms of the RWRE with respect to this harmonic h. Through this connection, it is our hope that the study of the objects constructed in this work may shed some light on the long-open questions concerning zero-one laws for RWRE models previously studied in [43, 56, 64, 67, 73, 74].

As mentioned in the opening paragraph of this manuscript, many other connections are present in related settings. Next, we briefly comment on two of these connections. Generalized Busemann functions have been shown in [40] to be equivalent to translation invariant stationary distributions for directed polymer models (in the sense that a realization of one induces a realization of the other). See also [41]. These translation-invariant stationary distributions play a prominent role in the KPZ scaling theory, which predicts the values of the non-universal constants needed to center and scale to see the universal distributions in the KPZ class [46, 65].

In the closely related setting of stochastic Hamilton-Jacobi equations, generalized Busemann functions correspond to globally defined solutions to the corrector equation, as discussed in [15, 41]. The construction of such objects is a key step in one approach to the problem of proving stochastic homogenization for such models, beginning with the pioneering work [50]. We refer the reader to the discussion in [7] as well as that in [41] for connections between the problems studied here and the general stochastic Hamilton-Jacobi setting.

Before concluding this section on motivation, it is important to clarify that our focus in this work lies in the simultaneous construction of the generalized Busemann functions for a countable dense set of vectors in the superdifferential of the limiting free energy. Our scope does not encompass the complete construction of the Busemann process, meaning a stochastic process indexed by the whole superdifferential. Thus far,

such a process has only been successfully constructed generally in two-dimensional directed nearest-neighbor settings [39, 42], where the path structure imposes a monotonicity that allows for the construction of the Busemann process based on its values at a dense countable set of directions.

We close this introductory section by remarking that this work leaves open many of the usual questions, such as those considered in Newman's seminal work [54], regarding the structure of the semi-infinite Gibbs measures and geodesics. In particular, it is natural to wonder whether all semi-infinite Gibbs measures and geodesics are directed into faces of the unrestricted-length limit shape. One might also ask for conditions under which one can show the uniqueness of these semi-infinite measures and geodesics or the non-existence of bi-infinite measures and geodesics. See [19, 20] for some recent relevant results. The existence of the full Busemann process and its connection to the non-uniqueness of Gibbs measures and infinite geodesics as in [39, 42] would also be of interest.

Another open problem of interest is to determine the conditions under which paths have an asymptotic law of large numbers (LLN) velocity. In the context of standard FPP, this is commonly referred to as the problem of asymptotic geodesic length. Our results show that a sufficient condition for paths to possess an almost sure asymptotic velocity is for the restricted-length limiting free energy to be strictly concave. However, it is known that this condition does not hold universally, as exemplified by marginally nestling RWRE, where the limiting free energy features a linear segment [68, Theorems 7.4 and 8.1]. Nevertheless, even for such models, it has been proven that in certain special cases the paths do have an asymptotic velocity [22, 55, 66], but the general scenario remains unresolved. The results in [10] and [45] suggest that in the standard FPP model, for a given asymptotic direction, there is a large class of weight distributions for which there are multiple asymptotic speeds and, conversely, there is a large class of weight distributions for which there is a unique asymptotic speed. Consequently, it is natural to question whether or not having positive temperature forces the LLN to hold, as is conjectured, for example, in the case of a uniformly elliptic RWRE with i.i.d. transition probabilities.

1.1. Methods and related work. In the polymer and percolation literature, two main approaches have been widely utilized to establish the existence of Busemann functions. The earliest approach can be traced back to the pioneering work of Newman and collaborators [35, 48, 54] on first-passage percolation.

In this approach, quantitative estimates on the strict convexity (or concavity, depending on sign conventions) of the limiting shape are used to prove geodesic coalescence, from which the existence of Busemann functions follows as a consequence. These ideas were later carried over to other percolation [4, 6, 17, 18] and polymer [8] models.

However, it is worth noting that obtaining the required curvature bounds for this approach, except for a few specific cases where the shape function can be explicitly computed, remains a major open problem in the field. More importantly, from our perspective, such conditions are provably false in full generality. It was shown by Häggstrom and Meester [33] that in two dimensions, every compact convex shape with all the symmetries of the lattice appears as the limit shape of some stationary FPP model. See also the polygonal examples constructed in [2, 13]. Moreover, even in the FPP setting, geodesic coalescence in dimensions three or higher continues to be an unresolved problem, with even its validity in high dimensions remaining unclear.

The approach we employ in this work is based on the connection between generalized Busemann functions and invariant measures. It broadly follows the idea of Césaro averaging distributions of Markov processes to produce stationary distributions, a concept that traces back at least to the classical Krylov-Bogoliouboff theorem [47]. The essential technical difficulty in this approach in RWRP models is constructing generalized Busemann functions that satisfy the duality condition (1.10) for a rich set of vectors m. There are two main steps to such an argument: constructing tight approximate Busemann functions with means converging to m and then establishing uniform integrability.

A method for proving the existence of generalized Busemann functions with the correct mean structure was developed (mostly in the context of FPP) over a series of works tracing from Liggett's proof of the subadditive ergodic theorem [49, Theorem 1.10], to Garet and Marchand [25] to Gouéré [30], Hoffman [34], and finally culminating with Damron and Hanson [23]. This approach was subsequently applied by Cardaliaguet and Souganidis [15] to construct correctors in a class of stochastic Hamilton-Jacobi equations.

The uniform integrability requirement is simplified in settings like FPP where loops are possible and easy upper and lower bounds on the approximate Busemann functions immediately imply the result. In settings where loops are not allowed in some directions, only one-sided bounds follow quickly from the model structure, and one needs more involved tools to obtain the mean convergence. In the queueing literature, this issue was also encountered in the construction of stationary $\cdot/G/1/\infty$ queues in [51], which also follows the general Krylov-Bogoliouboff approach mentioned above. This construction was later applied in [28] to show the existence of Busemann functions in the two-dimensional directed last-passage percolation model. A generally applicable method was introduced in [39] in the context of a 1 + 1 dimensional nearest-neighbor directed polymer in an i.i.d. random environment, where a variational representation of Λ^{β} , originally proven in that setting in [26], was used to show the uniform integrability.

Implementing these previous ideas in this general setting is complicated by the variety of admissible paths and the structure of the limiting free energy and shape function in the range of models we study. Our approach is a hybrid of that of [23] and [39], but with some significant differences from both. In [23], the approximate Busemann functions corresponding to a direction ξ are increments of point-to-(randomized-) hyperplane passage times, where the hyperplane is chosen to be tangent to the limit shape in direction ξ . These point-to-hyperplane passage times can be viewed as a random discrete approximation to the dual norm of the limit shape. One sees immediately from this construction how the relationship to a vector in the subdifferential of the (convex) limit shape arises. In the restricted-length setting of [39], by contrast, the approximate Busemann functions are increments of point-to-(randomized-)time-level passage times, with an external field added to the potential. These tilted point-to-level passage times are a random discrete approximation of the Legendre transform of the shape function. Standard convex analytic arguments then connect the external field to the superdifferential of the (concave) limiting free energy.

In our setting, which encompasses models with characteristics of both directed and undirected models, neither of these approaches appears to work on its own. Notably, there are cases where the tangent to the limiting free energy can be the zero vector, or the limiting free energy itself can be zero in non-trivial directions (e.g., in certain RWRE models), which poses challenges in implementing the approach of [23] in a general setting. Furthermore, when loops are permitted in the model, introducing external fields to unrestricted-length point-to-level models may lead to infinite partition functions. To address these challenges, we employ a hybrid approximation utilizing a point-to-hyperplane free energy with an external field, which proves successful except in purely directed cases where all hyperplanes become the zero hyperplane. In such instances, our approach closely resembles the one adopted in [39].

With generalized Busemann functions constructed, the existence of Gibbs-Dobrushin-Landford-Ruelle measures follows essentially immediately from the cocycle and recovery properties. We show that these measures are simultaneously consistent with both the restricted and unrestricted length point-to-point measures, as well as with the Green's function. The next natural questions are if the paths under these measures are directed (meaning that $X_n/|X_n|_1$ converges) and if they satisfy a law of large numbers (meaning that X_n/n converges).

The argument that the paths under the semi-infinite measures have an asymptotic direction requires some care because our models allow for the possible existence of traps that constrain paths to bounded sets. We give an exact characterization of when this occurs and provide easy-to-check conditions on the environment that ensure the paths do not become trapped. We then show that in the absence of traps, the paths are strongly directed into the set of directions that are in a Legendre-Fenchel duality (defined through the concave function Λ^{β}) with the mean of the Busemann function. This is the set of directions ξ such that $m \in \partial \Lambda^{\beta}(\xi)$, where m is the mean vector of the Busemann function. This essentially follows by combining the recovery property of the generalized Busemann functions with the shape theorems for both the free energy and the generalized Busemann functions themselves. This argument is similar to the approach taken in [23, 27], with certain details modified in the positive temperature case.

We also prove a large deviation principle for the velocity of the path under the semi-infinite polymer measure, using methods similar to [39, 58]. The possible law of large numbers limiting velocities are then included in the zero set of the rate function, which turns out to be the set of vectors ξ that are dual to the mean of the Busemann function through the concave function Λ^{β} , i.e., such that $m \in \partial \Lambda^{\beta}(\xi)$.

We conclude this section with a brief technical remark. For simplicity, the preceding discussion focused solely on the free energy Λ^{β} and its superdifferential as a function on the cone generated by the admissible steps \mathcal{R} . In actuality, when working with a direction ξ in this cone, we restrict attention to the unique face of the cone containing ξ in its relative interior. This restriction arises because paths with ξ as their asymptotic direction can only take steps within this face. Another subtle technical point is that the free energy Λ^{β} is not generally known to be continuous up to the boundary of the face. To address this, we instead consider the unique continuous extension of Λ^{β} from the interior to the entire face, which happens to be given by the upper-semicontinuous regularization of the restriction of Λ^{β} to the interior of the face. The Gibbs measures (and geodesics) we construct are then supported on semi-infinite paths that are constrained to steps lying within the corresponding face of the cone.

Outline of paper. The outline of the remainder of the paper is as follows. Section 2 describes our setting, defining the relevant path spaces and RWRP measures, then collects results about the free energies, and states the shape theorems. Section 3 collects the technical conditions for our results and states our main result concerning the existence of Gibbs measures and geodesics. Section 4 includes both the statement and proof of our results concerning the existence of the generalized Busemann functions. Section 5 contains the construction of the associated semi-infinite Gibbs measures. Section 6 then concludes the proofs of the results stated in Section 3. Appendix A collects some basic facts we use from linear algebra and convex analysis. Appendix B discusses Gibbs consistency and inconsistency of the various RWRP models we consider. Appendix C proves the shape theorems we state in Section 2. Appendix D relates the restricted-and unrestricted-length limiting free energies.

Notation. \mathbb{R} denotes the real numbers and \mathbb{Z} denotes the integers. For $a \in \mathbb{R}$, $\mathbb{R}_{\geq a} = [a, \infty)$, $\mathbb{R}_{>a} = (a, \infty)$, $\mathbb{Z}_{\geq a} = \mathbb{R}_{\geq a} \cap \mathbb{Z}$, and $\mathbb{R}_{\leq a}$, $\mathbb{R}_{< a}$, $\mathbb{Z}_{< a}$, $\mathbb{Z}_{< a}$ are similar. For $x \in \mathbb{R}^d$, let $|x|_1$ denote the ℓ^1 norm of x, and $|x|_{\infty}$ denote the ℓ^{∞} norm of x. A sequence $(x_k)_{n \leq k \leq m}$ is denoted by $x_{n:m}$. Similarly, $x_{-\infty:n} = (x_k)_{k \leq n}$, $x_{n:\infty} = (x_k)_{k \geq n}$, and $x_{-\infty:\infty} = (x_k)_{k \in \mathbb{Z}}$. When the index set is understood, we denote the sequence by x_{\bullet} . For $a \in \mathbb{R}$, $a^+ = \max(a, 0)$ and $a^- = -\min(a, 0)$.

Throughout, we use the convention that $\inf \emptyset = \infty$. ri A denotes the relative interior of a set $A \subset \mathbb{R}^d$. ext A denotes the set of extreme points of a convex set $A \subset \mathbb{R}^d$.

Given a metrizable topological space \mathcal{X} , we denote by $\mathcal{B}(\mathcal{X})$ the Borel σ -algebra of \mathcal{X} . The collection of probability measures on $(\mathcal{X}, \mathcal{B}(\mathcal{X}))$ is denoted by $\mathcal{M}_1(\mathcal{X})$ and equipped with the weak topology.

The symbol \triangle marks the end of a numbered remark.

2. Setting

In this section, we describe the models of random walks in random potential that we study in this work. The examples at the end of Section 2.2 show how a large number of models studied elsewhere in the literature are special cases of the models we consider.

2.1. The admissible paths. We begin by defining the path spaces used by the various models. Let $d \ge 1$ be a positive integer and consider a non-empty finite subset $\mathcal{R} \subset \mathbb{Z}^d$ containing at least two points. This is the set of admissible steps. \mathcal{R} may contain the zero vector $\mathbf{0} \in \mathbb{Z}^d$.

We denote the additive group and semi-group generated by \mathcal{R} by

$$\mathcal{G} = \left\{ \sum_{z \in \mathcal{R}} b_z z : b_z \in \mathbb{Z} \right\} \quad \text{and} \quad \mathcal{G}^+ = \left\{ \sum_{z \in \mathcal{R}} b_z z : b_z \in \mathbb{Z}_{\ge 0} \right\}.$$

For each $n \in \mathbb{Z}_{\geq 0}$, the set of points that are accessible from **0** in exactly n admissible steps is

$$D_n = \left\{ x \in \mathcal{G}^+ : \exists (b_z)_{z \in \mathcal{R}} \in \mathbb{Z}_{\geq 0}^{\mathcal{R}} \text{ with } \sum_{z \in \mathcal{R}} b_z = n \text{ and } x = \sum_{z \in \mathcal{R}} b_z z \right\}.$$

We now define several collections of admissible paths. The set of all admissible paths of length $n \in \mathbb{Z}_{\geq 0}$ from $x \in \mathbb{Z}^d$ is denoted by

$$\mathbb{X}_x^n = \left\{ x_{0:n} : x_0 = x, x_i - x_{i-1} \in \mathcal{R} \text{ for all } 1 \leq i \leq n \right\}$$

For integers $j \leq k$, the set of admissible paths from $x \in \mathbb{Z}^d$ at time j to $y \in \mathbb{Z}^d$ at time k is denoted by

$$\mathbb{X}_{x,y}^{j,k} = \left\{ x_{j:k} : x_j = x, x_k = y, x_{i+1} - x_i \in \mathcal{R} \text{ for all } j \leq i \leq k-1 \right\}$$

This set is empty unless $y - x \in D_{k-j}$. When j = 0, we abbreviate the set of admissible paths from x to y of length n by

$$\mathbb{X}_{x}^{n} = \mathbb{X}_{x}^{0,n}$$

The set of all admissible paths from $x \in \mathbb{Z}^d$ to $y \in \mathbb{Z}^d$ is

$$\mathbb{X}_{x,y} = \bigcup_{n=0}^{\infty} \mathbb{X}_{x,y}^n$$

This set is empty unless $y - x \in \mathcal{G}^+$. Define the space of paths from $x \in \mathbb{Z}^d$ that reach $y \in \mathbb{Z}^d$ for the first time in exactly $k \in \mathbb{Z}_{\geq 0}$ steps

$$\overline{\mathbb{X}}_{x,y}^k = \left\{ x_{0:k} \in \mathbb{X}_x^k : x_i \neq y \text{ for all } 0 \leq i < k \text{ and } x_k = y \right\}.$$

Define the space of killed paths from $x \in \mathbb{Z}^d$ to $y \in \mathbb{Z}^d$, i.e. paths of arbitrary length that start at x and end when they reach y:

$$\overline{\mathbb{X}}_{x,y} = \bigcup_{k=0}^{\infty} \overline{\mathbb{X}}_{x,y}^{k}$$

The set of all semi-infinite paths rooted at $x \in \mathbb{Z}^d$ is denoted by

$$\mathbb{X}_x = \{ x_{0:\infty} : x_0 = x, x_i - x_{i-1} \in \mathcal{R} \text{ for all } i \ge 1 \}.$$

 \mathbb{X}_x and the spaces above, which can be embedded in it, come equipped with the natural filtration generated by the coordinate projections. We denote the natural stochastic process on these path spaces by X_{\bullet} . For $y \in \mathbb{Z}^d$, define the stopping time $\tau_y = \inf\{k \ge 0 : X_k = y\}$.

All of the path spaces discussed above are compact metrizable spaces in the product-discrete topology.

When we wish to emphasize the dependence on the set \mathcal{R} of admissible steps, we will write $D_n(\mathcal{R})$, $\mathbb{X}_x^n(\mathcal{R})$, $\mathbb{X}_x(\mathcal{R})$, and so on.

Let $p : \mathcal{R} \to (0, 1]$ be a probability kernel, i.e. $\sum_{z \in \mathcal{R}} p(z) = 1$. We assume, without loss of generality, p(z) > 0 for all $z \in \mathcal{R}$. Let P_x (with expectation E_x) be the classical random walk starting at x and having i.i.d. increments with distribution p. This random walk is referred to as the *reference walk*, and P_x is the *reference measure*.

2.2. Random walks in random potentials. Let (Ω, \mathfrak{S}) be a Polish space endowed with its Borel σ algebra. We assume this measurable space is equipped with a commutative group of continuous bijections $T = \{T_x : \Omega \to \Omega : x \in \mathbb{Z}^d\}$ such that T_0 is the identity map, and for all $x, y \in \mathbb{Z}^d$, $T_x \circ T_y = T_y \circ T_x = T_{x+y}$. A generic element in Ω will be denoted by ω and is referred to as an *environment*. Let \mathbb{P} (with expectation \mathbb{E}) be a probability measure on (Ω, \mathfrak{S}) , called the *environment measure*. We assume \mathbb{P} is invariant under T_x for all $x \in \mathbb{Z}^d$. For a subset $S \subset \mathbb{Z}^d$, we say $(\Omega, \mathfrak{S}, \mathbb{P}, \{T_z : z \in S\})$ is ergodic if for every event $A \in \mathfrak{S}$ satisfying $T_z^{-1}A = A$ for all $z \in S$, $\mathbb{P}(A) \in \{0, 1\}$.

Let $V : \Omega \times \mathcal{R} \to \mathbb{R}$ be a measurable function, which we will call a *potential*. For $q \ge 1$, we say $V \in L^q$ if $\mathbb{E}[|V(\omega, z)|^q] < \infty$ for all $z \in \mathcal{R}$.

Let β be a positive real number, called the *inverse temperature*. For $x, y \in \mathbb{Z}^d$ such that $y - x \in \mathcal{G}^+$, $n \in \mathbb{Z}_{>0}$, and $u, v \in \mathbb{Z}^d$ and $n \in \mathbb{N}$ such that $v - u \in D_n$, the *restricted and unrestricted-length point-to-point* partition functions at inverse temperature β are, respectively,

(2.1)
$$Z_{u,v,n}^{\beta}(\omega) = \mathbf{E}_{u} \left[e^{-\beta \sum_{k=0}^{n-1} V(T_{X_{k}}\omega, X_{k+1} - X_{k})} \mathbb{1}_{\{X_{n} = v\}} \right] \text{ and }$$

(2.2)
$$Z_{x,y}^{\beta}(\omega) = \mathbf{E}_{x} \bigg[e^{-\beta \sum_{k=0}^{\tau_{y}-1} V(T_{X_{k}}\omega, X_{k+1}-X_{k})} \mathbb{1}_{\{\tau_{y}<\infty\}} \bigg].$$

Throughout the paper, depending on what notation is most convenient in a particular expression, we will at times place the dependence on ω into a superscript, writing expressions like $Z_{u,v,n}^{\beta,\omega}$ or omit the dependence on ω entirely and write expressions like $Z_{u,v,n}^{\beta}$. Since \mathcal{R} is finite, $Z_{u,v,n}^{\beta}(\omega)$ is always finite. $Z_{x,y}^{\beta}(\omega)$, on the other hand, can be infinite if the potential V can take negative values and the paths can make loops. When loops are present, (3.1) below suffices to ensure that $Z_{x,y}^{\beta}(\omega)$ is finite. Define $Z_{u,v,n}^{\beta}(\omega) = 0$ when $v - u \notin D_n$ and $Z_{x,y}^{\beta}(\omega) = 0$ when $y - x \notin \mathcal{G}^+$.

The corresponding restricted and unrestricted-length point-to-point free energies are, respectively,

$$F_{u,v,n}^{\beta}(\omega) = \frac{1}{\beta} \log Z_{u,v,n}^{\beta}(\omega) \text{ and } F_{x,y}^{\beta}(\omega) = \frac{1}{\beta} \log Z_{x,y}^{\beta}(\omega).$$

If $v - u \notin D_n$, define $F_{u,v,n}^{\beta}(\omega) = -\infty$. Similarly, if $y - x \notin \mathcal{G}^+$, then $F_{x,y}^{\beta}(\omega) = -\infty$.

Remark 2.1. A more physically correct definition of the free energy includes a minus sign. Many mathematics papers, including many of those on which we rely in this work, omit this sign to avoid proliferating minus signs in the various computations. To maintain consistency and avoid conflicting notation with most of the earlier works we cite, we have also omitted the minus sign. \triangle

The corresponding quenched point-to-point polymer measures are the probability measures on $\mathbb{X}_{u,v}^{0,n}$ and $\overline{\mathbb{X}}_{x,y}$, respectively, such that

$$\frac{d\mathbf{Q}_{u,v,n}^{\beta,\omega}}{d\mathbf{P}_{u}}(X\boldsymbol{\cdot}) = \frac{e^{-\beta\sum_{k=0}^{n-1}V(T_{X_{k}}\omega,X_{k+1}-X_{k})}\mathbb{1}_{\{X_{n}=v\}}}{Z_{u,v,n}^{\beta}(\omega)} \quad \text{and}$$
$$\frac{d\mathbf{Q}_{x,y}^{\beta,\omega}}{d\mathbf{P}_{x}}(X\boldsymbol{\cdot}) = \frac{e^{-\beta\sum_{k=0}^{\tau_{y}-1}V(T_{X_{k}}\omega,X_{k+1}-X_{k})}\mathbb{1}_{\{\tau_{y}<\infty\}}}{Z_{x,y}^{\beta}(\omega)}.$$

For a tilt (or external field or force) $h \in \mathbb{R}^d$ and an integer $n \ge 1$, define the tilted *n*-step partition function

$$Z_{x,n}^{\beta,h}(\omega) = \mathbf{E}_x \left[e^{-\beta \sum_{k=0}^{n-1} V(T_{X_k}\omega, X_{k+1} - X_k) + \beta h \cdot (X_n - x)} \right]$$

and the corresponding free energy

(2.3)

$$F_{x,n}^{\beta,h}(\omega) = \frac{1}{\beta} \log Z_{x,n}^{\beta,h}(\omega)$$

Similarly to the point-to-point case, $Z_{x,n}^{\beta,h}(\omega)$ is always finite, since \mathcal{R} is finite. The corresponding quenched polymer measure is the probability measure on \mathbb{X}_x^n such that

$$\frac{d\mathbf{Q}_{x,n}^{\beta,h,\omega}}{d\mathbf{P}_x}(X_{\bullet}) = \frac{e^{-\beta\sum_{k=0}^{n-1}V(T_{X_k}\omega,X_{k+1}-X_k)+\beta h\cdot(X_n-x)}}{Z_{x,n}^{\beta,h}(\omega)}$$

Through most of the manuscript, we work with the unrestricted-length model. This is without loss of generality because a restricted-length model can be rewritten as an unrestricted-length model, as explained in the following remark.

Remark 2.2. (Restricted-length models as unrestricted-length models) Consider a restricted-length model in \mathbb{Z}^d with steps $z \in \mathcal{R}$, random walk probability kernel p, and potential function $V(\omega, z)$. This model can be written as an unrestricted-length model in \mathbb{Z}^{d+1} as follows. Write $\mathbb{Z}^{d+1} = \mathbb{Z}^d \times \mathbb{Z}$ and denote elements of \mathbb{Z}^{d+1} by $\langle z, n \rangle$, where $z \in \mathbb{Z}^d$ and $n \in \mathbb{Z}$. The set of admissible steps of the new model is $\overline{\mathcal{R}} = \{\langle z, 1 \rangle : z \in \mathcal{R}\}$. Shifts in the (d + 1)-st dimension are defined to be the identity shift, i.e., $\overline{T}_{\langle z,n \rangle} = T_z$. The probability kernel is $\overline{p}(\langle z, 1 \rangle) = p(z)$, and the potential function is $\overline{V}(\omega, \langle z, 1 \rangle) = V(\omega, z)$. Then the restricted-length point-to-point partition function and quenched polymer measure from x to y in n steps are equivalent to the corresponding unrestricted-length objects from $\langle x, 0 \rangle$ to $\langle y, n \rangle$. Theorem 2.17 and Remarks 4.6 and 5.9 are examples of how results for the restricted-length model can be obtained from ones for the unrestricted-length model. Remark 3.19 explains how other quantities for the restricted-length model transfer to ones for the unrestricted-length model. \triangle

Taking $\beta \to \infty$ in the above definitions leads to polymer models at zero temperature, where the free energy becomes a passage time. For $x, y \in \mathbb{Z}^d$ with $y - x \in \mathcal{G}^+$ and an integer $k \ge 0$, the restricted and unrestricted-length point-to-point passage times are given by

$$F_{x,y,k}^{\infty} = \sup_{x_{0:k} \in \mathbb{X}_{x,y}^{k}} \sum_{i=0}^{k-1} (-V(T_{x_{i}}\omega, x_{i+1} - x_{i})) \text{ and}$$
$$F_{x,y}^{\infty} = \sup_{k \in \mathbb{Z}_{\geq 0}} \sup_{x_{0:k} \in \mathbb{X}_{x,y}^{k}} \sum_{i=0}^{k-1} (-V(T_{x_{i}}\omega, x_{i+1} - x_{i})).$$

Similar to the positive temperature situation, while $F_{x,y,k}^{\infty}$ is always finite, condition (3.1) below suffices to ensure the finiteness of $F_{x,y}^{\infty}$ if loops are allowed.

At zero temperature, the quenched measures are replaced by (restricted and unrestricted-length) geodesics, which are optimal paths that achieve the passage time. Between two points, there may be multiple geodesics. One may choose an ordering on \mathcal{R} to order all paths lexicographically, obtaining an ordering of geodesics. Then, for example, the first geodesic in this ordering is unique.

Next, we give examples of models that appear in the literature and are covered by our setting.

Example 2.3 (Edge and vertex weights). We can represent random weights assigned to the vertices of \mathbb{Z}^d by taking $\Omega = \mathbb{R}^{\mathbb{Z}^d}$ and $V(\omega) = \omega_0$. To represent directed edge weights, take $\Omega_0 = \mathbb{R}^{\mathcal{R}}$, $\Omega = \Omega_0^{\mathbb{Z}^d}$. Then for each

 $x \in \mathbb{Z}^d$, $\omega_x = (\omega_{(x,x+z)})_{z \in \mathcal{R}}$ is the vector of edge weights out of x. The potential function is $V(\omega, z) = \omega_{(0,z)}$. The shifts are $(T_v \omega)_{(x,x+z)} = \omega_{(x+v,x+v+z)}$ for $v \in \mathbb{Z}^d$.

For undirected edge weights, assume $\mathbf{0} \notin \mathcal{R}$ and define the set of edges $\mathcal{E} = \{\{x, y\} \subset \mathbb{Z}^d : y - x \in \mathcal{R} \text{ or } x - y \in \mathcal{R}\}$. Let $\Omega = \mathbb{R}^{\mathcal{E}}$ and $V(\omega, z) = \omega_{\{\mathbf{0}, z\}}$. The shifts are $(T_v \omega)_{\{x, x+z\}} = \omega_{\{x+v, x+v+z\}}$ for $v \in \mathbb{Z}^d$.

One can also have a hybrid of all three versions by letting $\Omega = \mathbb{R}^{\mathbb{Z}^d} \times (\mathbb{R}^{\mathcal{R}})^{\mathbb{Z}^d} \times \mathbb{R}^{\mathcal{E}}$. For each $x \in \mathbb{Z}^d$, the weights at x are given by $\omega_x = \{\alpha_x, \gamma_{x,x+z}, \delta_{\{x,x+z'\}} : z, z' \in \mathcal{R}\}$ and for $z \in \mathcal{R}$, $V(\omega, z) = \alpha_0 + \gamma_{0,z} + \delta_{\{0,z\}}$.

Example 2.4 (Product environment). Let Ω_0 be Polish and $\Omega = \Omega_0^{\mathbb{Z}^d}$ equipped with the product topology and Borel σ -algebra. We denote points $\omega = (\omega_x)_{x \in \mathbb{Z}^d}$. The translations are $(T_x \omega)_y = \omega_{x+y}$. \mathbb{P} is a product measure if $\{\omega_x\}_{x \in \mathbb{Z}^d}$ are i.i.d. random variables under \mathbb{P} . \mathbb{P} has a *finite range of dependence* if there exists an $r_0 \ge 0$ such that for any $A, B \subset \mathbb{Z}^d$ where $|x - y|_1 \ge r_0$ for all $x \in A$ and $y \in B$, $\{\omega_x : x \in A\}$ and $\{\omega_y : y \in B\}$ are independent. V is *local* if there exists an $L \ge 0$ such that for all $z \in \mathcal{R}$, $V(\omega, z)$ is measurable with respect to $\sigma\{\omega_x : |x|_1 \le L\}$. In other words, V only depends on finitely many coordinates ω_x .

Example 2.5. Our general setting covers the following models.

(1) First- and last-passage percolation. At zero temperature, directed last-passage percolation is obtained by taking $\mathcal{R} = \{e_1, \ldots, e_d\}, \ \Omega = \mathbb{R}^{\mathbb{Z}^d}$, setting $V(\omega, z) = -\omega_0$, and taking $\beta = \infty$. The last-passage time is

$$F_{x,y}^{\infty} = \sup_{k \in \mathbb{Z}} \sup_{x_{0:k} \in \mathbb{X}_{x,y}^{k}} \sum_{i=0}^{k-1} \omega_{x_{i}}.$$

Standard first-passage percolation is obtained by taking $\mathcal{R} = \{\pm e_1, \ldots, \pm e_d\}$, putting weights on the edges as explained in Example 2.3, then taking $\beta = \infty$ and putting a minus sign in front of the free energy. The first-passage time is

$$-F_{x,y}^{\infty} = \inf_{k \in \mathbb{Z}} \inf_{x_{0:k} \in \mathbb{X}_{x,y}^{k}} \sum_{i=0}^{k-1} \omega_{\{x_{i},x_{i+1}\}}$$

- (2) Directed polymers in random environments (DPRE). This is the case where $\beta < \infty$ and there exists a vector $\hat{u} \in \mathbb{R}^d$ with $z \cdot \hat{u} > 0$ for all $z \in \mathcal{R}$ or, equivalently, where 0 does not lie in the convex hull of \mathcal{R} . Thus, the polymer always moves strictly forward in direction \hat{u} ; by [60, Corollary A.2], this is equivalent to the condition that there are no loops, i.e., $\mathbb{X}_{0,0} = \{\mathbf{0}\}$. A commonly studied special case, which we call a space-time directed polymer, is one where there exists a vector $\hat{u} \in \mathbb{R}^d$ such that $z \cdot \hat{u} = 1$ for all $z \in \mathcal{R}$. Then $(X_n - X_0) \cdot \hat{u} = n$ plays the role of a time coordinate. There is a large literature studying this model; see [21] and the references therein. By Lemma A.2, this is satisfied if and only if, for any two points x and y in \mathbb{Z}^d such that $y - x \in \mathcal{G}^+$, all paths from x to y take the same number of steps. A common example of this is when $\mathcal{R} = \{e_1, e_2, \dots, e_d\}$.
- (3) Random walk in a random environment (RWRE). Let \mathcal{R} be general and $\beta = 1$. Let $\pi_z : \Omega \to [0, 1]$, $z \in \mathcal{R}$, be measurable such that $\sum_{z \in \mathcal{R}} \pi_z = 1$, \mathbb{P} -almost surely. Let $V(\omega, z) = -\log \pi_z(\omega) + \log p(z)$. Then $Z_{x,n}^{1,\omega} = 1$ and the measures $\{Q_{x,n}^{1,0,\omega} : n \in \mathbb{N}\}$ are consistent, being the marginal distributions of the Markov chain X_n starting at x which uses transition probabilities $\pi_{y,y+z} = \pi_z(T_y\omega)$. Under the path measure this family induces by Kolmogorov's extension theorem, $Z_{x,y}^{\beta}(\omega)$ is the probability that the path ever reaches y if started at x and $Q_{x,y}^{1,0,\omega}$ is the distribution of this Markov chain conditioned on reaching y, provided that this probability is positive. A space-time directed polymer which is also a random walk in a random environment is called a random walk in a dynamic random environment.

An RWRE is called *elliptic* if $\pi_z > 0$ for all $z \in \mathcal{R}$ and *uniformly elliptic* if there exists a deterministic c > 0 such that $\pi_z \ge c$ almost surely for all $z \in \mathcal{R}$. If the model is not elliptic, we say it is *degenerate*; such models are not covered by the results of this paper. Note that uniform ellipticity is equivalent to the potential V being bounded.

An RWRE is called *nestling* if **0** lies in the relative interior of the convex hull of the support of the random vector $\sum_{z \in \mathcal{R}} \pi_z z$, marginally nestling if **0** lies on the relative boundary of this convex hull and non-nestling if **0** does not lie in the convex hull. This concept features in some of the discussion above and in the discussion below.

- (4) Stretched polymers. The term $h \cdot (X_n x)$ in the definition of $Q_{x,n}^{\beta,h,\omega}$ can be interpreted as an external force that stretches the polymer. A special case that appears in the literature is when $\mathcal{R} = \{\pm e_1, \pm e_2, \dots, \pm e_d\}, \Omega = \mathbb{R}^{\mathbb{Z}^d}$, and $V(\omega, z) = \omega_0$. This model is studied in [36], for example.
- (5) Killed polymers. Let $\beta < \infty$ and consider an unrestricted-length model with $\Omega = \mathbb{R}^{\mathbb{Z}^d}$, $\mathcal{R} = \{\pm e_1, \pm e_2, \dots, \pm e_d\}$, and $V(\omega, z) = \omega_0$. The Green's function g(x, y), defined in (3.7), can be interpreted as the expected number of visits of the reference random walk to the site y before it is "killed" by the potential. See, for example, [72, p. 249].

2.3. Limiting free energies. Before we state our main result, we need to record a few limit theorems for the various free energies defined in Section 2.2. These results require the following definition.

Definition 2.6. For a finite subset $\mathcal{R} \subset \mathbb{Z}^d$ and $z \in \mathcal{R} \setminus \{0\}$, a non-negative measurable function $g : \Omega \to \mathbb{R}$ is in class $\mathcal{L}_{z,\mathcal{R}}$ if

$$\overline{\lim_{\varepsilon \searrow 0} \lim_{n \to \infty}} \max_{\substack{x \in \mathcal{G}^+(\mathcal{R}) \\ \|x\|_1 \le n}} n^{-1} \sum_{0 \le k \le \varepsilon n} g(T_{x+kz}\omega) = 0 \quad for \ \mathbb{P}\text{-}a.e. \ \omega.$$

Membership in class $\mathcal{L}_{z,\mathcal{R}}$ depends on a trade-off between the moments of g and the degree of mixing of \mathbb{P} . For example, if (\mathbb{P}, T_z) is exponentially mixing, then $g \in L^q(\mathbb{P})$ for some q > d is sufficient. For a general \mathbb{P} , if g is bounded, then $g \in \mathcal{L}_{z,\mathcal{R}}$. See [60, Lemma A.4] for a precise formulation of sufficient conditions to ensure $g \in \mathcal{L}_{z,\mathcal{R}}$.

For inverse temperature $\beta \in (0, \infty]$ and velocity $\xi \in \mathcal{U}$ the restricted-length limiting quenched point-to-point free energy is defined by

(2.4)
$$\Lambda_{\operatorname{res}}^{\beta}(\xi) = \lim_{n \to \infty} n^{-1} F_{\mathbf{0}, \hat{x}_n(\xi), n}^{\beta},$$

where $\hat{x}_n(\xi) \in D_n$ is a lattice point approximation of $n\xi$ as defined in [58, (2.1)].

It was shown in Theorem 2.2 in [58] and Theorem 2.4 in [26] that the limit (2.4) exists in $(-\infty, \infty]$, P-almost surely, for all $\xi \in \mathcal{U}$ simultaneously. By [58, Theorem 2.6] and the inequalities in [26, (2.11)], $\Lambda_{\text{res}}^{\beta}(\xi)$ is either identically infinite for all $\xi \in \text{ri}\mathcal{U}$ and $\beta \in (0, \infty]$ or $\Lambda_{\text{res}}^{\beta}$ is bounded on \mathcal{U} , for each $\beta \in (0, \infty]$. In the latter case, [58, Theorem 2.6] and [26, Remark 2.5] state that with P-probability one, $\Lambda_{\text{res}}^{\beta}$ is lower semicontinuous on \mathcal{U} and concave and continuous on ri \mathcal{U} . Furthermore, the upper semicontinuous regularization of $\Lambda_{\text{res}}^{\beta}$ and its unique continuous extension from ri \mathcal{U} to \mathcal{U} are equal. The conditions of Theorem 2.17 below ensure that $\Lambda_{\text{res}}^{\beta}$ is finite on the relative interior of the face being considered in that result.

Remark 2.7. In [26] and [58], the authors assume that \mathbb{P} is ergodic, but the results we mention above continue to hold for a stationary \mathbb{P} by the ergodic decomposition theorem. The authors of these papers also assume that for each $z' \in \mathcal{R}$, $|V(\omega, z')| \in L^1(\mathbb{P})$ and belongs to $\mathcal{L}_{z,\mathcal{R}}$ for all $z \in \mathcal{R} \setminus \{0\}$. However, upon closer examination of their proofs, it can be observed that the condition on V in Theorem 2.17 below is sufficient for their results to hold. This can be seen, for instance, in the proof of Lemma 5.20, which demonstrates how the arguments can be adjusted to work under the weaker assumptions.

Remark 2.8. In general, $\Lambda_{\text{res}}^{\beta}(\xi)$ may not be deterministic. For example, if **0** is an extreme point of \mathcal{U} , then $\Lambda_{\text{res}}^{\beta}(\mathbf{0}) = -V(\omega, \mathbf{0}) + \beta^{-1} \log p(\mathbf{0})$. A sufficient condition to ensure that it is deterministic is to have the ergodicity of $(\Omega, \mathfrak{S}_{\mathcal{R}'}^V, \mathbb{P}, \{T_z : z \in \mathcal{R}'\})$, where $\mathcal{R}' = \mathcal{U}' \cap \mathcal{R}, \mathcal{U}'$ is the unique face of \mathcal{U} such that $\xi \in \mathrm{ri}\mathcal{U}', \mathfrak{S}_{\mathcal{R}'}^V = \sigma(V(T_x\omega, z) : x \in \mathcal{G}', z \in \mathcal{R}')$, and \mathcal{G}' is the additive group generated by \mathcal{R}' .

The following example provides a simple model where Λ_{res}^1 can be computed via simulations. We will also refer to this example throughout the paper to illustrate key results and definitions.

Example 2.9 (Nearest-neighbor RWRE on Z). Let p(1) = p(-1) = 1/2 so that $\mathcal{R} = \{-1, 1\}$ and $\mathcal{U} = [-1, 1]$. Consider the i.i.d. RWRE model as in Example 2.5(3). Greven and den Hollander [31] gave a detailed description of $-\Lambda_{\text{res}}^1$, which is the large deviation rate function for the position of the particle. They demonstrated that this function is always differentiable away from 0 but may have a corner at 0. It also may have two linear pieces around the corner at 0 but is strictly convex otherwise. Figure 2.1 shows the results of simulations for the model with $\pi_1 = 1 - \pi_{-1} = a_1$ with probability q and $\pi_1 = 1 - \pi_{-1} = a_2$ with probability 1 - q, for the various regimes of the parameters $q, a_1, a_2 \in (0, 1)$. N, in the caption of the figure, is the number of steps we used for the random walk in the simulation used to approximate $-\Lambda_{\text{res}}^1$, which is plotted in blue. The unrestricted-length rate function $-\Lambda^1$, defined in (2.6) below, is plotted in red. See Remark 2.11 for the details involved in the computation of $-\Lambda^1$.



A: (Recurrent) q = 1/3, $a_1 = 1/5$, $a_2 = 2/3$, N = 2,000





B: (Transient with 0 velocity) q = 2/3, $a_1 = 3/4$, $a_2 = 3/10$, N = 20,000



C: (Marginal nestling) $q=3/5,\;a_1=1/2,\;a_2=95/100,\;N=1000,000$

D: (Non-nestling) q = 4/5, $a_1 = 9/10$, $a_2 = 7/10$, N = 200,000

FIGURE 2.1. The thicker blue curves show plots of the approximation of $-\Lambda_{\rm res}^1(t)$ from Example 2.9 by $-N^{-1}F_{0,|Nt|,N}^1$ for $t \in [-1,1]$ in simulations. In the recurrent example in Figure 2.1A, $-\Lambda_{\rm res}^1$ is differentiable and strictly convex. In the transient example with 0 velocity in Figure 2.1B, $-\Lambda_{\rm res}^1$ is strictly convex and differentiable except at 0. In the ballistic marginal nestling example in Figure 2.1C, $-\Lambda_{\rm res}^1$ has a flat segment and a linear segment, meeting at a corner at 0. Elsewhere, it is differentiable and strictly convex. (The ballistic nestling case looks qualitatively the same.) In the non-nestling example in Figure 2.1D, $-\Lambda_{\rm res}^1$ has two linear segments that meet at a corner at 0. Elsewhere, it is differentiable and strictly convex. The thinner red piecewise linear graphs show plots of $-\Lambda^1(t)$ for $t \in [-1, 1]$. In all cases, $\Lambda^1(t) = 0$ for $t \ge 0$. In Figure 2.1A, $\Lambda^1(t) = 0$ for $t \le 0$ as well. In Figures 2.1B and 2.1C, the slope of $-\Lambda^1$ on t < 0 matches the slope of the linear segment of $-\Lambda_{\rm res}^1$ left of 0. In Figure 2.1D, $-\Lambda^1$ is tangent to $-\Lambda_{\rm res}^1$ at a unique point in (-1, 0).

Define the cone

$$\mathcal{C} = \Big\{ \sum_{z \in \mathcal{R}} b_z z : b_z \in \mathbb{R}_+ \Big\}.$$

For a face \mathcal{A} of \mathcal{C} , let $\mathcal{R}_{\mathcal{A}} = \mathcal{R} \cap \mathcal{A}$, $\mathcal{U}_{\mathcal{A}} = \mathcal{U} \cap \mathcal{A}$, and let $\mathcal{G}_{\mathcal{A}}$ and $\mathcal{G}_{\mathcal{A}}^+$ denote, respectively, the group and the semigroup generated by $\mathcal{R}_{\mathcal{A}}$. For the definition of a face of a convex set, see Appendix A. Define

(2.5)
$$\mathcal{R}_{\mathcal{A}}^{\mathrm{id}} = \{ z \in \mathcal{R}_{\mathcal{A}} : T_z \text{ is the identity map} \}$$

13

to be the set of steps for which the associated shift is the identity. If $\mathbf{0} \in \mathcal{R}_{\mathcal{A}}$, then $\mathbf{0} \in \mathcal{R}_{\mathcal{A}}^{id}$. Non-zero steps can also be in this set. As an example, in the restricted-length setting of Remark 2.2, if $\mathbf{0}$ is an admissible step, $\overline{T}_{\langle \mathbf{0}, 1 \rangle}$ is the identity. It may also be the case that $\mathcal{R}_{\mathcal{A}}^{id} = \emptyset$. Let \mathbb{A} be the set of faces \mathcal{A} of \mathcal{C} such that $\mathcal{R}_{\mathcal{A}} \neq \mathcal{R}_{\mathcal{A}}^{id}$. Note that if $\mathbf{0}$ is an extreme point of \mathcal{U} , then the face $\{\mathbf{0}\}$ is not in \mathbb{A} .

Given $\mathcal{A} \in \mathbb{A}$, $\xi \in \operatorname{ri} \mathcal{A}$, and an inverse temperature $\beta \in (0, \infty]$, the unrestricted-length limiting quenched point-to-point free energy is defined by

(2.6)
$$\Lambda^{\beta}(\xi) = \lim_{n \to \infty} n^{-1} F_{\mathbf{0}, x_n}^{\beta}.$$

where the sequence $x_n \in \mathcal{G}^+_{\mathcal{A}}$ satisfies $x_n/n \to \xi$. In [37, Theorem 2.8] it is shown that the limit exists \mathbb{P} almost surely for all $\xi \in \mathcal{C}$ simultaneously, and is finite, positively 1-homogeneous, and lower semicontinuous on \mathcal{C} , and concave and continuous on ri \mathcal{C} . As in Remark 2.7, although [37] assumes stronger conditions on class \mathcal{L} membership, minor adjustments of their proofs (along the lines of the proof of our Lemma 5.20) give the same results under the assumptions of Theorem 2.15 below.

Remark 2.10. Similarly to the restricted-length case, a sufficient condition to guarantee that $\Lambda^{\beta}(\xi)$ is deterministic is to have the ergodicity of $(\Omega, \mathfrak{S}^{V}_{\mathcal{R}'}, \mathbb{P}, \{T_z : z \in \mathcal{R}'\})$, where $\mathcal{R}' = \mathcal{A} \cap \mathcal{R}, \ \mathcal{A} \in \mathbb{A}$ is the unique face of \mathcal{C} such that $\xi \in \mathrm{ri} \mathcal{A}$, and $\mathfrak{S}^{V}_{\mathcal{R}'}$ is as in Remark 2.8.

Remark 2.11 (RWRE on \mathbb{Z}). We sketch the computation of Λ^{β} in the case of the one-dimensional RWRE models from Example 2.9 which were discussed in Figure 2.1. Here, $\mathcal{C} = \mathbb{R}$ and it is the only face.

In this model, $Z_{0,x}^{1,\omega}$ represents the probability that the RWRE, starting at 0, eventually reaches x. In the recurrent case (Figure 2.1A), $Z_{0,x}^{1,\omega} = 1$ almost surely for all $x \in \mathbb{Z}$, which implies that Λ^1 is identically 0. Similarly, in the other (transient-to-the-right) cases, $\Lambda^1(t) = 0$ for all $t \ge 0$. By homogeneity, $\Lambda^1(t) = -t\Lambda^1(-1)$ for all t < 0. Using the results in [70], we identify $-\Lambda^1(-1)$, which corresponds to $\overline{\lambda}(0)$ in the notation of that paper.

In that work, the rate function I corresponds to our $-\Lambda_{\rm res}^1$, and $\bar{\lambda}(0)$ there matches $-\Lambda^1(-1)$ in our framework. For the transient nestling cases (Figures 2.1B and 2.1C), Theorem 1.8 of [70] shows that, in the notation of that paper, $r_c = I(0) = -\Lambda_{\rm res}^1(0) = 0$. The arguments in the proof of that theorem (page 1052 of the paper) further establish that the left derivative of $-\Lambda_{\rm res}^1$ at 0 equals $\bar{\lambda}(0) = -\Lambda^1(-1)$. This explains the behavior depicted in Figures 2.1B and 2.1C.

In the non-nestling case (Figure 2.1D), we have $r_c = I(0) = -\Lambda_{\rm res}^1(0) > 0$, and there exists a point $\bar{\xi}_c \in (-1,0)$ such that $r(\bar{\xi}_c) = r_c$, where $r(\xi)$ is defined in the middle of page 1050 in [70], substituting $\bar{\lambda}$ for λ . By [70, Theorem 1.8], $\bar{\xi}_c$ is the left endpoint of the interval $[\bar{\xi}_c, 0]$ where $I = -\Lambda_{\rm res}^1$ is linear, with the derivative of $-\Lambda_{\rm res}^1 = I$ at $\bar{\xi}_c$ matching its left derivative at 0. Then, referring again to the formula on page 1052 in [70], we get that the left derivative of $-\Lambda_{\rm res}^1 = I$ at 0 equals $\bar{\lambda}(r(\bar{\xi}_c)) = \bar{\lambda}(I(0)) > \bar{\lambda}(0)$. In particular, $-\Lambda^1(-1) < 0$, and the slope of the line representing $-\Lambda^1$ on $(-\infty, 0]$ is steeper than that of the linear segment of $-\Lambda_{\rm res}^1$ intersect at a unique point in $(-1, \bar{\xi}_c)$. This explains the behavior illustrated in Figure 2.1D. See Remark 4.7 for more.

The above limits can be strengthened to uniform limits, commonly referred to as "shape theorems." The shape theorem for the unrestricted-length model is established in [37, Theorem 3.10], where the uniform limit holds within the interior of each face. Theorem 2.15 extends this result to the entire face, while Theorem 2.17 presents the shape theorem for the restricted-length model. Detailed proofs are provided in Appendix C. A key observation is that the shape theorem is stated face-by-face, with the proper centering achieved through the upper-semicontinuous regularization of the limiting shape restricted to the interior of each face. The restriction to faces arises because paths starting at the origin and terminating at a specific point remain within the face containing the endpoint in its relative interior (see Lemma A.4). The primary technical point, however, is the necessity of the upper-semicontinuous regularization. This stems from the fact that the continuity of the limiting shape up to the boundaries of the faces is not generally known. Thus, the correct centering requires using the unique continuous regularization.

Recall the upper semicontinuous regularization of a function $f: \mathcal{X} \to [-\infty, \infty)$ defined as

$$f^{\text{usc}}(x) = \inf \left\{ \sup_{y \in G} f(y) : G \ni x \text{ and } G \text{ is open} \right\}$$

Given a face \mathcal{A} of \mathcal{C} , let $\Lambda_{\mathcal{A}}^{\beta}$ be the function that equals Λ^{β} on \mathcal{A} and equals $-\infty$ outside of \mathcal{A} . Denote the upper semicontinuous regularization of $\Lambda_{\mathcal{A}}^{\beta}$ by $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$. Then, under the conditions of Theorem 2.15 below, the restriction of $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$ to \mathcal{A} is the unique continuous extension of $\Lambda_{\mathcal{A}}^{\beta}$ from ri \mathcal{A} to all of \mathcal{A} . In particular, $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$ is concave and positively 1-homogeneous.

Remark 2.12. The shape Λ^{β} is expected to be continuous on all of C in a wide class of natural models. In such cases, the restrictions and upper semi-continuous extensions discussed above can be dispensed with. The example following this remark gives examples where this continuity is known to hold.

In the example in Remark 2.8, if the distribution of a zero step is continuous, then Λ^{β} cannot be continuous at zero and so considering passage times seen from the relative interior of a face is necessary.

Example 2.13 (i.i.d. directed LPP and RWRP). In i.i.d. directed LPP on \mathbb{Z}^d with $\mathcal{R} = \{e_1, e_2, \ldots, e_d\}$ (discussed in Example 2.5(1)) under a moment hypothesis slightly weaker than the weights having > d moments, it is shown in [53] that the shape function Λ^{∞} is continuous on all of $\mathcal{C} = \mathbb{R}^d_+ = \{(x_1, \ldots, x_d) : x_1, \ldots, x_d \ge 0\}$.

Continuity of Λ^1_{res} on \mathcal{U} is proven for a wide collection of RWRP models in [58, Theorem 3.2].

We will reference the next example frequently in the discussion that follows to illustrate our results and the definitions.

Example 2.14 (FPP on \mathbb{Z}^3 with a forbidden step). Consider the standard FPP on \mathbb{Z}^3 with i.i.d. Exponential(1) weights, but where paths are not allowed to use the $-e_3$ step. We can model this as a zero temperature model with hybrid edge weights as in Example 2.3. In the notation of that example, let $\mathcal{R} = \{e_1, -e_1, e_2, -e_2, e_3\}$. For $x \in \mathbb{Z}^3$ and $z \in \{-e_1, e_1, -e_2, e_2\}$ let $\alpha_x = \gamma_{x,x+z} = \delta_{\{x,x+e_3\}} = 0$ and let $\{\gamma_{x,x+e_3}, \delta_{\{y,y+e_1\}}, \delta_{\{u,u+e_2\}} : x, y, u \in \mathbb{Z}^d\}$ be i.i.d. Exponential(1) random variables. The first-passage time is given by $-F_{x,y}^{\infty}$. Then $\mathcal{C} = \{(x, y, z) \in \mathbb{R}^3 : z \ge 0\}$ has two non-empty faces: \mathcal{C} and its boundary, $\partial \mathcal{C} = \{(x, y, 0) : x, y \in \mathbb{R}\}$. Restricted to $\mathcal{A} = \partial \mathcal{C}$, this model becomes the classical planar standard FPP with i.i.d. Exponential(1) edge-weights and the upper semi-continuous regularization of $-\Lambda_{\mathcal{A}}^{\infty}$ is the usual shape function of that planar FPP. See Figure 2.2 for simulations of the limit shape in this model.

As explained above, it is enough to study the model from the perspective of the relative interior of each face \mathcal{A} of \mathcal{C} . That there is a locally uniform version of the limit (2.6) defining the shape function is the content of the following "shape theorem".

For $x \in \mathcal{G}^+$ let \mathcal{I}_x be the σ -algebra generated by $A \in \mathfrak{S}$ such that $T_x^{-1}A = A$.

Theorem 2.15. Let $\beta \in (0, \infty]$. Fix a face $\mathcal{A} \in \mathbb{A}$ of \mathcal{C} (possibly \mathcal{C} itself). Assume $V^+(\omega, z) \in L^1(\mathbb{P})$ for each $z \in \mathcal{R}_{\mathcal{A}}$, and $V^+(\omega, z) \in \mathcal{L}_{z,\mathcal{R}_{\mathcal{A}}}$ for each $z \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}$. For each $z \in \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}$, assume that there exists a $\hat{z} \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}$ such that $V^+(\omega, z) \in \mathcal{L}_{\hat{z},\mathcal{R}_{\mathcal{A}}}$. Assume that $\mathbb{E}[\sup_{n \geq 1} n^{-1}\mathbb{E}[F_{\mathbf{0},nx}^{\beta} | \mathcal{I}_x]] < \infty$ for all $x \in \mathcal{G}_{\mathcal{A}}^+ \setminus \{\mathbf{0}\}$. Then \mathbb{P} -almost surely, $\Lambda_{\mathcal{A}}^{\beta}$ is finite on \mathcal{A} , $\Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}$ is the unique continuous extension of $\Lambda_{\mathcal{A}}^{\beta}$ from ri \mathcal{A} to \mathcal{A} ,

(2.7)
$$\lim_{\substack{|x|_1 \to \infty \\ x \in \mathcal{G}_{+}^{4}}} \frac{F_{\mathbf{0},x}^{\beta} - \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(x)}{|x|_1} \leq 0$$

and, simultaneously for all $\xi \in \mathcal{A}$ with $|\xi|_1 = 1$ and any sequence $v_n \in \mathcal{G}^+_{\mathcal{A}}$ with $|v_n|_1 \to \infty$ and $v_n/|v_n|_1 \to \xi$, we have

(2.8)
$$\Lambda^{\beta}(\xi) \leq \lim_{n \to \infty} |v_n|_1^{-1} F_{\mathbf{0}, v_n}^{\beta} \leq \overline{\lim}_{n \to \infty} |v_n|_1^{-1} F_{\mathbf{0}, v_n}^{\beta} \leq \Lambda^{\beta, \text{usc}}_{\mathcal{A}}(\xi)$$

If furthermore $(\Omega, \mathfrak{S}_{\mathcal{R}_{\mathcal{A}}}^{V}, \mathbb{P}, \{T_{z} : z \in \mathcal{R}_{\mathcal{A}}\})$ is ergodic, then $\Lambda_{\mathcal{A}}^{\beta, \text{usc}}$ is deterministic on \mathcal{A} .

In the statement of the previous result, the first three conditions concerning the integrability of V^+ and membership of potential steps in \mathcal{L} classes are minor regularity assumptions which are satisfied in the typical applications. The last condition, on the integrability of conditional expectations of $n^{-1}F_{0,nx}^{\beta}$, is addressed in the following remark.

Remark 2.16. [37, Theorem 2.8] provides common cases which guarantee the hypothesis

$$\mathbb{E}\left[\sup_{n\geq 1}n^{-1}\mathbb{E}[F_{\mathbf{0},nx}^{\beta}\,|\,\mathcal{I}_{x}]\right] < \infty$$





A: Level 75 sub-level set of an approximation of $-\Lambda^{\infty}_{\mathcal{A}}$, $\mathcal{A} = \mathcal{C} = \{(x, y, z) \in \mathbb{R}^3 : z \ge 0\}$. Darker colors represent points with lower z coordinates.

B: Level 75 sub-level set of an approximation of $-\Lambda_{\mathcal{A}}^{\infty}$, where $\mathcal{A} = \partial \mathcal{C} = \{(x, y, 0) : x, y \in \mathbb{R}\}$. Darker colors represent points closer to the origin.

FIGURE 2.2. A simulation of the ball of radius 75, $\{\xi : -\Lambda^{\infty}(\xi) \leq 75\}$, in Example 2.14 obtained by taking sub-level sets of the average of 100 samples of passage times from the origin. Both figures use the same data, so the pane on the right is the figure on the left viewed from below (with different coloring). We expect that Λ^{∞} is continuous on C.

holds for all $x \in \mathcal{G}^+ \setminus \{0\}$. Particularly this hypothesis is satisfied under Condition 3.2 below. See [37, Lemmas 3.11 and 3.12] for the proofs of these implications.

Naturally, there is a version of the previous result in the special case of a restricted-length model, which we record below for use later in the paper. Given a face \mathcal{U}' of \mathcal{U} , let $\Lambda^{\beta}_{\mathcal{U}', \text{res}}$ be the function that equals $\Lambda^{\beta}_{\text{res}}$ on \mathcal{U}' and equals $-\infty$ outside of \mathcal{U}' . Denote the upper semicontinuous regularizations of $\Lambda^{\beta}_{\mathcal{U}', \text{res}}$ by $\Lambda^{\beta, \text{usc}}_{\mathcal{U}', \text{res}}$. Then, under the conditions of Theorem 2.17, the restriction of $\Lambda^{\beta, \text{usc}}_{\mathcal{U}', \text{res}}$ to \mathcal{U}' is the unique continuous extension of $\Lambda^{\beta}_{\mathcal{U}', \text{res}}$ from ri \mathcal{U}' to all of \mathcal{U}' . In particular, $\Lambda^{\beta, \text{usc}}_{\mathcal{U}', \text{res}}$ and $\Lambda^{\beta, \text{usc}}_{\mathcal{A}}$ are concave and $\Lambda^{\beta, \text{usc}}_{\mathcal{A}}$ is positively 1-homogeneous.

Theorem 2.17. Let $\beta \in (0, \infty]$. Fix a face $\mathcal{U}' \neq \{\mathbf{0}\}$ of \mathcal{U} (possibly \mathcal{U} itself). Let \mathcal{A} be the cone generated by \mathcal{U}' . Assume $\mathcal{R}_{\mathcal{A}} \neq \mathcal{R}^{\mathrm{id}}_{\mathcal{A}}$. Assume $V^+(\omega, z) \in L^1(\mathbb{P})$ for each $z \in \mathcal{R}_{\mathcal{A}}$, and $V^+(\omega, z) \in \mathcal{L}_{z,\mathcal{R}_{\mathcal{A}}}$ for each $z \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}^{\mathrm{id}}_{\mathcal{A}}$. For each $z \in \mathcal{R}^{\mathrm{id}}_{\mathcal{A}}$, assume that there exists a $\hat{z} \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}^{\mathrm{id}}_{\mathcal{A}}$ such that $V^+(\omega, z) \in \mathcal{L}_{\hat{z},\mathcal{R}_{\mathcal{A}}}$. Take $\beta \in (0,\infty]$. Assume that for all $k \in \mathbb{Z}_{>0}$ and $x \in D_k(\mathcal{R} \cap \mathcal{U}')$, $\mathbb{E}[\sup_{n \geq 1} n^{-1}\mathbb{E}[F^{\beta}_{\mathbf{0},nx,nk} \mid \mathcal{I}_x]] < \infty$. Then \mathbb{P} -almost surely, $\Lambda^{\beta}_{\mathcal{U}',\mathrm{res}}$ is finite on \mathcal{U}' , $\Lambda^{\beta,\mathrm{usc}}_{\mathcal{U}',\mathrm{res}}$ is the unique continuous extension of $\Lambda^{\beta}_{\mathcal{U}',\mathrm{res}}$ from $\mathrm{ri}\,\mathcal{U}'$ to \mathcal{U}' ,

(2.9)
$$\overline{\lim_{n \to \infty} \max_{x \in n \mathcal{U}' \cap D_n}} \frac{F^{\beta}_{\mathbf{0},x,n} - \Lambda^{\beta,\text{usc}}_{\mathcal{U}',\text{res}}(x)}{n} \leqslant 0.$$

and, simultaneously for all $\xi \in \mathcal{U}'$ and any sequence $v_n \in \mathcal{G}^+_{\mathcal{A}}$ with $v_n/n \to \xi$ we have

(2.10)
$$\Lambda^{\beta}_{\mathcal{U}', \mathrm{res}}(\xi) \leq \lim_{n \to \infty} n^{-1} F^{\beta}_{\mathbf{0}, v_n, n} \leq \overline{\lim_{n \to \infty}} n^{-1} F^{\beta}_{\mathbf{0}, v_n, n} \leq \Lambda^{\beta, \mathrm{usc}}_{\mathcal{U}', \mathrm{res}}(\xi).$$

If furthermore $(\mathbb{P}, \mathfrak{S}_{\mathcal{R}'}^V, \mathbb{P}\{T_z : z \in \mathcal{R}_{\mathcal{A}}\})$ is ergodic, then $\Lambda_{\mathcal{U}', res}^{\beta, usc}$ is deterministic on \mathcal{U}' .

Remark 2.18. Condition 3.3 below implies

$$\mathbb{E}\Big[\sup_{n\geq 1} n^{-1} \mathbb{E}[F_{\mathbf{0},nx,nk}^{\beta} \,|\, \mathcal{I}_x]\Big] < \infty$$

holds for all $k \in \mathbb{Z}_{>0}$ and $x \in D_k(\mathcal{R} \cap \mathcal{U}')$. This can be shown by transforming the restricted-length model into an unrestricted-length one, as described in Remark 2.2, using (3.10), and following the argument in Remark 2.16 within the context of the unrestricted-length version.

3. The main result

Recall that \mathcal{R} denotes the set of admissible steps, its convex hull \mathcal{U} , and the cone $\mathcal{C} = \{\sum_{z \in \mathcal{R}} b_z z : b_z \in \mathbb{R}_+\}$. By [60, Corollary A.2], $\mathbf{0} \in \mathcal{U}$ is equivalent to the existence of loops, i.e. $\mathbb{X}_{\mathbf{0},\mathbf{0}} \neq \{\mathbf{0}\}$. By [60, Corollary A.3], $\mathbf{0} \in \mathrm{ri}\mathcal{U}$ is equivalent to the existence of admissible paths from $\mathbf{0}$ to each $x \in \mathcal{G}$, i.e. $\mathcal{G}^+ = \mathcal{G}$. See Figure 3.1 for an illustration of possible cases of \mathcal{R} and \mathcal{U} and Figure 3.2 for illustrations of \mathcal{C} . Note that because this cone is finitely generated, it is either equal to \mathbb{R}^d (which happens if $\mathbf{0} \in \mathrm{ri}\mathcal{U}$) or it is an intersection of finitely many half spaces whose boundary hyperplanes pass through the origin (see [62, Theorem 19.1]).



FIGURE 3.1. The different panels depict various possible cases of \mathcal{R} and \mathcal{U} . In each panel, the large balls represent $z \in \mathcal{R}$ that are extreme in \mathcal{U} , the small ball is **0**, and \mathcal{U} is the lightly shaded figure. Left to right: $\exists \hat{u} : \forall x \in \mathcal{U}, x \cdot \hat{u} = 1; \mathbf{0} \notin \mathcal{U}; \mathbf{0} \in \text{ext} \mathcal{U}; \mathbf{0} \in \mathcal{U} \setminus \text{ri} \mathcal{U}.$

Given a proper concave function $f : \mathbb{R}^d \to [-\infty, \infty)$, we will denote by ∂f the superdifferential of f. For $\xi \in \mathbb{R}^d$,

$$\partial f(\xi) = \{ v \in \mathbb{R}^d : f(\zeta) \leq f(\xi) + v \cdot (\zeta - \xi) \quad \forall \zeta \in \mathbb{R}^d \}.$$

By the argument in [62, p. 15], $\partial f(\xi)$ is a closed convex set. The image of $A \subset \mathbb{R}^d$ is denoted by $\partial f(A)$. If f is positively 1-homogeneous, then by the Fenchel-Young inequality, if $m \in \partial f(\xi)$ then $f(\xi) = m \cdot \xi$. See Lemma A.1.

For each face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself), let $W_{\mathcal{A}}$ be the linear subspace of \mathbb{R}^d generated by $\mathcal{R}_{\mathcal{A}}$. Since $\Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\zeta) = -\infty$ when $\zeta \notin W_{\mathcal{A}}$ we have

$$W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi) = \{ v \in W_{\mathcal{A}} : \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\zeta) \leq \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi) + v \cdot (\zeta - \xi) \quad \forall \zeta \in W_{\mathcal{A}} \}.$$

The set on the right-hand side is the superdifferential at ξ of the restriction of $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$ to $W_{\mathcal{A}}$. If $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}(\xi)$ is finite for all $\xi \in \mathcal{A}$, then [62, Theorem 23.4] says that this set is non-empty and bounded. It then has an extreme point, e.g., by the Krein–Milman theorem. Remark 3.6 explains how Conditions 3.2 and 3.5 below guarantee the finiteness of $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$. See Figure 3.2 for an illustration of some possible cones and super-level sets of $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$. Figure 2.2 already gives an example of the case where **0** lies on the relative boundary of \mathcal{C} .

Remark 3.1 (FPP on \mathbb{Z}^3 with a forbidden step). In the model described in Example 2.14 above, the cone has two faces $\mathcal{C} = \{(x, y, z) : x, y \in \mathbb{R}, z \ge 0\}$ and $\partial \mathcal{C} = \{(x, y, 0) : x, y \in \mathbb{R}\}$. If $\mathcal{A} = \mathcal{C}$, then $W_{\mathcal{A}} = \mathbb{R}^3$ and the elements of $W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\infty, \text{usc}}(\xi)$ correspond to 3-dimensional hyperplanes (2-dimensional planes) which are tangent to $\Lambda_{\mathcal{A}}^{\infty, \text{usc}}$ at ξ . See Figure 2.2A. If $\mathcal{A} = \partial \mathcal{C}$, then $W_{\mathcal{A}} = \{(x, y, 0) : x, y \in \mathbb{R}\}$ and the elements of $W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\infty, \text{usc}}(\xi)$ correspond to 2-dimensional hyperplanes (1-dimensional lines) on $\{(x, y, 0) : x, y \in \mathbb{R}\}$ which are tangent to $\Lambda_{\mathcal{A}}^{\infty, \text{usc}}$ at ξ . See Figure 2.2B. Δ

To ensure that the unrestricted-length polymer measures in the next theorem are well defined, we need the following conditions.

Condition 3.2 (Unrestricted-length conditions). Given a face \mathcal{A} of \mathcal{C} (possibly \mathcal{C} itself), assume that one of the following holds:



FIGURE 3.2. The different panels depict various possible cases of the super-level sets $B_t = \{\xi \in \mathcal{C} : \Lambda^{\beta, \text{usc}}(\xi) \ge t\}$, which are depicted darkly shaded. When $\mathbf{0} \in \text{ri}\mathcal{U}$, the cone \mathcal{C} is the whole of \mathbb{R}^d . When $\mathbf{0} \notin \text{ri}\mathcal{U}$, the cone \mathcal{C} is depicted lightly shaded. These cones are unbounded but truncated in the drawings and hence appear as pyramids. The faces of each of these cones are the cone itself, the two-dimensional faces that are truncated in the drawing and appear as triangles on the side of the pyramid, the lines that are the boundaries of the two-dimensional faces, and the point $\mathbf{0}$. The super-level sets on the first row are all bounded. Left to right: $\mathbf{0} \in \text{ri}\mathcal{U}$, t < 0, and $\Lambda^{\beta, \text{usc}}$ is differentiable; $\mathbf{0} \in \text{ri}\mathcal{U}$, t < 0, and B_t is polygonal; $\mathbf{0} \notin \mathcal{U}$, $\Lambda^{\beta, \text{usc}}(\xi) > 0$ $\forall \xi \neq \mathbf{0}$, t < 0, and B_t is polygonal. The shapes on the second row are all unbounded and are truncated in the drawings. Left to right: $\mathbf{0} \notin \mathcal{U}$, $\Lambda^{\beta, \text{usc}}$ takes both < 0 and ≥ 0 values, t < 0, and B_t is polygonal; $\mathbf{0} \notin \mathcal{U}$, $\Lambda^{\beta, \text{usc}}(\xi) > 0 \ \forall \xi \neq \mathbf{0}$, t < 0, $\forall \xi \neq \mathbf{0}$, t > 0, and $\Lambda^{\beta, \text{usc}}$ takes both < 0 and ≥ 0 values, t < 0, and B_t is polygonal; $\mathbf{0} \notin \mathcal{U}$, $\Lambda^{\beta, \text{usc}}(\xi) > 0 \ \forall \xi \neq \mathbf{0}$, t > 0, and $\Lambda^{\beta, \text{usc}}$ takes both < 0 and $A^{\beta, \text{usc}}$ is differentiable; $\mathbf{0} \notin \mathcal{U}$, $\Lambda^{\beta, \text{usc}}(\xi) > 0 \ \forall \xi \neq \mathbf{0}$, t > 0, and $\Lambda^{\beta, \text{usc}}$ is differentiable.

(a) The setting is undirected (meaning $\mathbf{0} \in \mathcal{U}_{\mathcal{A}}$) and the potential satisfies

$$(3.1) \qquad \qquad \mathbb{P}\{V(\omega, z) \ge 0\} = 1 \quad for \ all \ z \in \mathcal{R}_{\mathcal{A}}$$

and $V^+(\omega, z) \in L^1(\mathbb{P})$ for each $z \in \mathcal{R}_A$.

(b) The setting is directed (meaning $\mathbf{0} \notin \mathcal{U}_{\mathcal{A}}$), there exists $c \in \mathbb{R}$ such that

$$(3.2) \qquad \qquad \mathbb{P}\{V(\omega, z) \ge c\} = 1 \quad \text{for all } z \in \mathcal{R}_{\mathcal{A}},$$

and $V^+(\omega, z) \in L^1(\mathbb{P})$ for each $z \in \mathcal{R}_A$.

(c) The setting of Example 2.4 holds, $\mathbf{0} \notin \mathcal{U}_{\mathcal{A}}$, V is local, \mathbb{P} has a finite range of dependence r_0 , and for some q > d, $V^+(\omega, z) \in L^1(\mathbb{P})$ and $V^-(\omega, z) \in L^q(\mathbb{P})$ for all $z \in \mathcal{R}_{\mathcal{A}}$.

When considering a restricted-length model, one can convert it to an unrestricted-length model as in Remark 2.2. When viewing the model without doing this, parts (b) and (c) of the above condition then specialize to the condition below, which we present without going through unrestricted-length measures for the convenience of the reader.

Condition 3.3 (Restricted-length conditions). Given a face \mathcal{U}' of \mathcal{U} (possibly \mathcal{U} itself), assume that one of the following holds:

(a) There exists $c \in \mathbb{R}$ such that

 $\mathbb{P}\{V(\omega, z) \ge c\} = 1 \quad for \ all \ z \in \mathcal{R} \cap \mathcal{U}'$

and $V^+(\omega, z) \in L^1(\mathbb{P})$ for each $z \in \mathcal{R} \cap \mathcal{U}'$.

(b) The setting of Example 2.4 holds, $\mathbf{0} \notin \mathcal{U}'$, V is local, \mathbb{P} has a finite range of dependence r_0 , and for some q > d, $V^+(\omega, z) \in L^1(\mathbb{P})$ and $V^-(\omega, z) \in L^q(\mathbb{P})$ for all $z \in \mathcal{R} \cap \mathcal{U}'$.

Before continuing, we comment on the strength and generality of these hypotheses.

Remark 3.4. The moment hypotheses in Conditions 3.2 and 3.3 hold in typical applications but are also often not quite sharp in particular examples. For example, in the standard FPP with i.i.d. edge weights, one can often weaken the moment hypothesis to one involving the minimum of the edge weights at a site. If $\mathbf{0} \in \mathcal{U}_{\mathcal{A}}$, then (3.1) in Condition 3.2(a) is necessary for the finiteness of $Z_{x,y}^{\beta}$. Otherwise loops can be repeated indefinitely, causing $Z_{x,y}^{\beta}(\omega) = \infty$ for $x, y \in \mathbb{Z}^d$ with $y - x \in \mathcal{G}_{\mathcal{A}}^+$, with positive probability. The standard FPP gives a frequently studied example of a model satisfying this condition. Condition 3.2(b) (and 3.3(a)) corresponds to a physically reasonable assumption that energies are bounded from below. Among many other examples, this allows us to consider well-studied bounded potential models like uniformly elliptic RWRE models in ergodic environments, without needing independence. The last conditions allow us to cover examples which are unbounded below with a finite range of dependence. Natural examples of such models not covered by the first two cases include i.i.d. Gaussian last-passage percolation and directed polymers as well as (not necessarily uniformly) elliptic RWREs in independent environments which satisfy our moment conditions.

Recall the set $\mathcal{R}^{id}_{\mathcal{A}}$, defined in (2.5). To apply the shape theorems for the limiting free energy we need the following mild technical condition. As explained after Definition 2.6, this condition is a trade-off between moment and mixing assumptions and is typically valid in practice.

Condition 3.5. Given a face \mathcal{A} of \mathcal{C} (possibly \mathcal{C} itself), assume $V^+(\omega, z) \in L^1(\mathbb{P})$ for each $z \in \mathcal{R}_{\mathcal{A}}$, and $V^+(\omega, z) \in \mathcal{L}_{z,\mathcal{R}_{\mathcal{A}}}$ for each $z \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}$. For each $z \in \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}$, assume that there exists a $\hat{z} \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}$ such that $V^+(\omega, z) \in \mathcal{L}_{\hat{z},\mathcal{R}_{\mathcal{A}}}$.

Remark 3.6. By [37, Theorem 2.8], if Conditions 3.2 and 3.5 are satisfied, then we have, \mathbb{P} -almost surely, $|\Lambda_{\mathcal{A}}^{\beta}(\xi)| < \infty$ for all $\xi \in \mathrm{ri} \mathcal{A}$, and $|\Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(\xi)| < \infty$ for all $\xi \in \mathcal{A}$. As mentioned earlier, the version of Condition 3.5 assumed by [37] is stronger than the one we assume here. However, their results still hold under our weaker condition with minor adjustments in the proofs. For detailed information on these adjustments, see the proof of Lemma 5.20.

A primary objective in this work is to construct polymer measures that are supported on the set of directed semi-infinite paths. To that end, we now define directedness.

Definition 3.7. For a set $A \subset U$, the sequence x_n is directed into A if $|x_n|_1 \to \infty$ and all limit points of $x_n/|x_n|_1$ are contained in A.

The directedness of the polymers we construct requires the following extra assumption that avoids degenerate situations. Let \mathcal{U}_0 be the unique face of \mathcal{U} that contains 0 in its relative interior. Let $\mathcal{R}_0 = \mathcal{R} \cap \mathcal{U}_0$. If $0 \notin \mathcal{U}$, then $\mathcal{U}_0 = \mathcal{R}_0 = \emptyset$. Let

$$\mathcal{A}_{\mathbf{0}} = \Big\{ \sum_{z \in \mathcal{R}_{\mathbf{0}}} b_z z : b_z \in \mathbb{R}_+ \Big\}.$$

Condition 3.8. Assume that

(3.4)
$$\mathbb{P}\{V(\omega, z) \ge 0\} = 1 \quad \text{for all } z \in \mathcal{R}_0$$

and if $\beta < \infty$, then assume also that either the reference walk P₀ is transient or it is recurrent and

$$\mathbb{P}\left\{\exists x \in \mathcal{G}_{\mathcal{A}_{\mathbf{0}}}^{+}, \exists z \in \mathcal{R}_{\mathbf{0}} : V(T_{x}\omega, z) > 0\right\} = 1.$$

Remark 3.9. Condition (3.4) states that the potential V must be non-negative along any loops that may exist. This is implied by Condition 3.2, which we already assume to ensure the well-definedness of the unrestricted-length polymer measures with which we work. By Lemma 5.14 below, Condition 3.8 is equivalent to the positive-temperature semi-infinite polymers we construct being transient, which is a necessary precondition for them to move in a preferred direction. \triangle

For $\beta \in (0, \infty]$, a face $\mathcal{A} \in \mathbb{A}$ (including \mathcal{C}), and $m \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\text{ri } \mathcal{A})$, define

(3.5)
$$\mathcal{F}_{m,\mathcal{A}}^{\beta} = \{\xi \in \mathcal{A} : |\xi|_1 = 1 \text{ and } \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(\xi) = m \cdot \xi\}$$

This set is not empty because $m \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\text{ri }\mathcal{A})$ and $\Lambda^{\beta, \text{usc}}$ is positively 1-homogeneous. Also, define

(3.6)
$$\mathcal{U}_{m,\mathcal{A}}^{\beta} = \{\xi \in \mathcal{U}_{\mathcal{A}} : \Lambda_{\mathcal{A},\mathrm{res}}^{\beta,\mathrm{usc}}(\xi) = m \cdot \xi\}.$$

Theorem 3.14 implies, in particular, that this set is not empty (a direct convex analytic proof is also possible).

Definition 3.10. For $x, y \in \mathbb{Z}^d$ with $y - x \in \mathcal{G}$ the Green's function is given by

(3.7)
$$g(x,y) = \sum_{\ell=0}^{\infty} Z_{x,y,\ell}^{\beta} = \mathbb{1}_{\{y=x\}} + \sum_{\ell=1}^{\infty} \mathbb{E}_x \Big[e^{-\beta \sum_{j=0}^{\ell-1} V(T_{X_j}\omega, X_{j+1} - X_j)} \mathbb{1}_{\{X_\ell = y\}} \Big]$$
$$= \mathbb{E}_x \Big[\sum_{\ell=0}^{\infty} e^{-\beta \sum_{j=0}^{\ell-1} V(T_{X_j}\omega, X_{j+1} - X_j)} \mathbb{1}_{\{X_\ell = y\}} \Big],$$

where an empty sum is taken to be zero.

Let $\mathcal{D}_{\mathcal{A}}$ be a countable (possibly empty or finite) set of pairs (β, m) so that for each pair, $\beta \in (0, \infty]$, $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$ is deterministic and finite on \mathcal{A} , and there exists a $\xi \in (\text{ri }\mathcal{A}) \setminus \{\mathbf{0}\}$ such that $m \in W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta,\text{usc}}(\xi)$ and, if $\mathbf{0} \in \mathcal{U}_{\mathcal{A}}, \Lambda_{\mathcal{A}}^{\beta}(\xi) \neq 0$. An important case will be one where each m is an extreme point,

(3.8)
$$m \in \operatorname{ext}(W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta, \operatorname{usc}}(\xi)) \text{ for some } \xi \in (\operatorname{ri} \mathcal{A}) \setminus \{\mathbf{0}\}.$$

Also, define $\mathcal{D} = \{(\mathcal{A}, \beta, m) : \mathcal{A} \in \mathbb{A}, (\beta, m) \in \mathcal{D}_{\mathcal{A}}\}$ and $\mathcal{B} = \{\beta \in (0, \infty] : \exists m \in \mathbb{R}^d, \mathcal{A} \in \mathbb{A} \text{ with } (\mathcal{A}, \beta, m) \in \mathcal{D}\}.$

Remark 3.11. For maximum utility, one would take $\mathcal{D}_{\mathcal{A}}$ to consist of a countable dense set of $\beta \in (0, \infty]$ which includes ∞ and then for each such β , an appropriately dense subset of $m \in \bigcup_{\xi \in \mathrm{ri} \mathcal{A} \setminus \{\mathbf{0}\}} \{ \mathrm{ext}(W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta, \mathrm{usc}}(\xi)) \}$. One natural way to construct such a set is to take a dense set of β as above and then a dense subset of directions $\xi \in \mathrm{ri}(\mathcal{A}) \setminus \{\mathbf{0}\}$ (for which $\Lambda_{\mathcal{A}}^{\beta, \mathrm{usc}}(\xi) \neq 0$ if $\mathbf{0} \in \mathcal{U}_{\mathcal{A}}$) which are directions of differentiability of $\Lambda_{\mathcal{A}}^{\beta}$. For such ξ , $\partial \Lambda_{\mathcal{A}}^{\beta}(\xi)$ consists of a single point. Existence of such a set follows from [62, Theorem 25.4].

Remark 3.12. It can happen that (β, m) can be in $\mathcal{D}_{\mathcal{A}}$ for multiple faces \mathcal{A} of \mathcal{C} . A simple, albeit degenerate, example is one where d = 3, $\mathcal{R} = \{e_1, e_2, e_3\}$, and $\Lambda^{\beta}(\xi) = \xi \cdot e_1$. Then for $\mathcal{A} \in \{\mathcal{C}, \mathcal{C}(e_1, e_2), \mathcal{C}(e_1, e_3), \mathcal{C}(e_1)\}$, and $\xi \in \operatorname{ri} \mathcal{A}, \, \partial \Lambda^{\beta}_{\mathcal{A}}(\xi) = e_1$. For all other faces $\mathcal{A}, \, \Lambda^{\beta}_{\mathcal{A}} \equiv 0$ and $\partial \Lambda^{\beta}_{\mathcal{A}} = \mathbf{0}$.

Remark 3.13 (RWRE on \mathbb{Z}). We illustrate the conditions on the membership in $\mathcal{D}_{\mathcal{A}}$ in the case of the RWRE models from Figure 2.1, which were defined in Example 2.9. In this model, $\mathcal{C} = \mathbb{R}$ is the only face and $W_{\mathcal{C}} = \mathbb{R}$. Recall the description of Λ^1 from Remark 2.11 and Figures 2.1. In the recurrent case, we have $\partial \Lambda^1(t) = \{0\}$ for all $t \in \mathbb{R}$. In this case, there are no points that can belong to $\mathcal{D}_{\mathcal{C}}$ with $\beta = 1$. In all the other cases, $\partial \Lambda^1(t) = \{0\}$ for t > 0, $\partial \Lambda^1(t) = \{-\Lambda^1(-1)\}$ for t < 0, and $\partial \Lambda^1(0) = [-\Lambda^1(-1), 0]$ (with the extreme slopes in the last case being $-\Lambda^1(-1)$ and 0). Thus, the only point that can belong to $\mathcal{D}_{\mathcal{C}}$ with $\beta = 1$ is $(1, -\Lambda^1(-1))$.

If, on the other hand, we consider the restricted-length version of the RWRE model and rewrite it as an unrestricted-length model, following Remark 2.2, then the paths become directed and the restriction on the choice of ξ in the definition of $\mathcal{D}_{\overline{\mathcal{C}}}$ disappears. See Remark 4.7 for further details.

With the previous discussion in mind, the following is our main result. It produces for each linear facet of $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$ (which can degenerate into a singleton) a probability measure on semi-infinite paths that is consistent with both the restricted-length and the unrestricted-length point-to-point quenched polymer measures and under which the polymer is directed into the given facet. Appendix B discusses the Gibbs consistency properties of the restricted- and unrestricted-length measures, from which one can connect to the familiar

setting of Dobrushin-Lanford-Ruelle equations. Before stating the result, note that because C is finitely generated, it has finitely many faces [62, Theorem 19.1].

Theorem 3.14. For each face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself) where $\mathcal{D}_{\mathcal{A}} \neq \emptyset$, assume that V satisfies Conditions 3.2 and 3.5 on \mathcal{A} and that (3.8) holds for each $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$. For each $x \in \mathbb{Z}^d$, there exists a family $\{Q_x^{\mathcal{A},\beta,m,\omega} : \mathcal{A} \in \mathbb{A}, (\beta,m) \in \mathcal{D}_{\mathcal{A}}\}$ of random probability measures on semi-infinite paths $x_{0:\infty} \in \mathbb{X}_x(\mathcal{R}_{\mathcal{A}})$, such that the following is satisfied. There exists an event $\Omega_{dir} \subset \Omega$ with $\mathbb{P}(\Omega_{dir}) = 1$, on which the following properties hold:

(a) If $\beta < \infty$, then for all non-negative integers $j \leq k \leq n$, points $u, x, y \in \mathbb{Z}^d$ with x - u and y - x in $\mathcal{G}^+_{\mathcal{A}}, x_{0:k} \in \overline{\mathbb{X}}_{u,y}$ with $j = \min\{0 \leq i \leq k : x_i = x\}$, and $x_{k:n} \in \mathbb{X}^{n-k}_y(\mathcal{R}_{\mathcal{A}})$,

$$\begin{aligned} \mathbf{Q}_{u}^{\mathcal{A},\beta,m,\omega}(X_{0:\tau_{y}+n-k} = x_{0:n} \,|\, \tau_{x} \leqslant \tau_{y} < \infty, X_{0:\tau_{x}} = x_{0:j}, X_{\tau_{y}:\tau_{y}+n-k} = x_{k:n}) \\ &= \mathbf{Q}_{u}^{\mathcal{A},\beta,m,\omega}(X_{\tau_{x}:\tau_{y}} = x_{j:k} \,|\, \tau_{x} \leqslant \tau_{y} < \infty) = \mathbf{Q}_{x,y}^{\beta,\omega}(X_{0:\tau_{y}} = x_{j:k}). \end{aligned}$$

(b) If $\beta < \infty$, then for each distinct $x, y \in \mathbb{Z}^d$ such that $y - x \in \mathcal{G}_4^+$,

$$\mathbf{E}^{\mathbf{Q}^{\mathcal{A},\beta,m,\omega}_{x}}\left[\sum_{n=0}^{\infty}\mathbbm{1}_{\{X_{n}=y\}}\Big|\tau_{y}<\infty\right]=\frac{g(x,y)}{Z^{\beta,\omega}_{x,y}}$$

(c) If $\beta < \infty$, then for all non-negative integers $j \leq k \leq n$, points $u, x, y \in \mathbb{Z}^d$ such that $x - u \in D_j(\mathcal{R}_{\mathcal{A}})$ and $y - x \in D_{k-j}(\mathcal{R}_{\mathcal{A}})$, $x_{0:j} \in \mathbb{X}_{u,x}^{0,j}$, $x_{j:k} \in \mathbb{X}_{x,y}^{j,k}$, and $x_{k:n} \in \mathbb{X}_y^{n-k}(\mathcal{R}_{\mathcal{A}})$,

$$Q_{u}^{\mathcal{A},\beta,m,\omega}(X_{0:n} = x_{0:n} | X_{0:j} = x_{0:j}, X_{k:n} = x_{k:n}) = Q_{u}^{\mathcal{A},\beta,m,\omega}(X_{j:k} = x_{j:k} | X_{j} = x, X_{k} = y)$$
$$= Q_{x,y,k-j}^{\beta,\omega}(X_{0:k-j} = x_{j:k}).$$

- (d) If $\beta = \infty$, then for each $x \in \mathbb{Z}^d$, $Q_x^{\mathcal{A},\infty,m,\omega}$ is supported on a set of semi-infinite geodesics rooted at x and having increments in $\mathcal{R}_{\mathcal{A}}$.
- (e) If either $\beta = \infty$, or $\beta < \infty$ while $\mathcal{A} \neq \mathcal{A}_{\mathbf{0}}$, or $\beta < \infty$, $\mathcal{A} = \mathcal{A}_{\mathbf{0}}$, and Condition 3.8 holds, then for all $x \in \mathbb{Z}^d$, $\mathbb{Q}_x^{\mathcal{A},\beta,m,\omega}\{|X_n|_1 \to \infty\} = 1$ and

(3.9)
$$Q_x^{\mathcal{A},\beta,m,\omega}(X_{0:\infty} \text{ is directed into } \mathcal{F}_{m,\mathcal{A}}^{\beta}) = 1.$$

(f) For each $x \in \mathbb{Z}^d$, $Q_x^{\mathcal{A},\beta,m,\omega}(all\ limit\ points\ of\ X_n/n\ are\ contained\ in\ \mathcal{U}_{m,\mathcal{A}}^\beta) = 1$.

Remark 3.15 (Previous work). This result extends several works in the literature, including [23, Theorem 1.1] and the extension of that result to \mathbb{Z}^d described in [3, Section 5.4] (though with slightly more restrictive moment hypotheses), [39, Theorem 3.2], and [27, Theorem 2.1 (i) and (ii)]. The proof of Theorem 3.14 relies primarily on Theorem 4.5, which provides another construction of the queueing fixed points in [51, Theorem 5.1] as well as the harmonic functions for RWREs considered in [69, Theorem 4] and [71, Theorem 3.1] (for directions where the quenched and averaged rate functions coincide), [70, Lemma 1.6] (the one-dimensional RWRE case), and [26, Example 5.6] (the L^2 weak disorder regime). It can also be applied to exactly solvable models such as those in [9, 29] to construct infinite-volume Gibbs measures, infinite geodesics, and generalized Busemann functions. In [26, 69, 71], stronger almost-sure limits are proved using martingales available in these settings. Similarly, almost-sure limits are obtained in [9, 24, 27, 29, 39, 54, 70] by exploiting the geometric constraints on paths inherent to lower-dimensional cases.

Remark 3.16. If $\mathcal{F}^{\beta}_{m,\mathcal{A}}$ consists of a single point, e.g. when $\Lambda^{\beta,\text{usc}}_{\mathcal{A}}$ is strictly concave, then (3.9) says that the polymer has an asymptotic direction, $Q^{\mathcal{A},\beta,m,\omega}_x$ -almost surely. Similarly, when $\mathcal{U}^{\beta}_{m,\mathcal{A}}$ is a singleton, (f) says that the polymer has an asymptotic velocity, $Q^{\mathcal{A},\beta,m,\omega}_x$ -almost surely.

Remark 3.17. A common way to parametrize the measures in the above theorem is by directions $\xi \in (\mathrm{ri}\,\mathcal{A}) \setminus \{\mathbf{0}\}, \ \mathcal{A} \in \mathbb{A}$ instead of by vectors m in the superdifferential $\partial \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(\mathrm{ri}\,\mathcal{A}), \ \mathcal{A} \in \mathbb{A}$. This can be done by applying Theorem 3.14 with $m \in \mathrm{ext}(W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(\xi))$. Such an extreme point must exist, as explained earlier in this section. Then by homogeneity, $(\xi/|\xi|_1) \cdot m = \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(\xi/|\xi|_1)$, and $\xi/|\xi|_1 \in \mathcal{F}_{m,\mathcal{A}}^{\beta}$. As a result, \mathbb{P} -almost surely, under $Q_x^{\mathcal{A},\beta,m,\omega}, X_{0:\infty}$ is almost surely directed into $\mathcal{F}_{m,\mathcal{A}}^{\beta}$, which is a linear facet of $\Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}$ that contains ξ .

Remark 3.18 (FPP with a forbidden step). In Example 2.14, applying the previous result to $\mathcal{A} = \partial \mathcal{C} = \{(x, y, 0) : x, y \in \mathbb{R}\}$ with $\mathcal{D}_{\mathcal{A}}$ given by a countable dense set of directions of differentiability, we recover the construction of semi-infinite geodesics in planar Exponential FPP which can be obtained from the results in [23]. Because here we have $\mathbf{0} \in \mathcal{U}_{\mathcal{A}}$, our conditions defining $\mathcal{D}_{\mathcal{A}}$ require us to know that such directions satisfy $\Lambda_{\mathcal{A}}^{\infty}(\xi) \neq 0$. This follows from [44, Theorems 3.1 and 6.1]. Applying the result to $\mathcal{A} = \mathcal{C} = \{(x, y, z) \in \mathbb{R}^3 : z \geq 0\}$ with $\mathcal{D}_{\mathcal{A}}$ given by a countable dense set of interior directions of differentiability in \mathcal{C} , we similarly obtain the existence of semi-infinite geodesics corresponding to a countable collection of faces of the limit shape defined by sub-level sets of $-\Lambda_{\mathcal{A}}^{\infty}$ as illustrated in Figure 2.2A. These have asymptotic directions contained in those faces by part (e) of the theorem. Note that by a coupling with standard three-dimensional exponential FPP, it also follows from [44, Theorems 3.1 and 6.1] that $\Lambda_{\mathcal{A}}^{\infty}(\xi) \neq 0$ for $\xi \in \mathcal{C} \setminus \{\mathbf{0}\}$ here.

Remark 3.19. (Restricted-length models as unrestricted-length models) Remark 2.2 explained how one can rewrite a restricted-length model as an unrestricted-length one. We explain here how certain quantities for the restricted-length model transfer to ones for the unrestricted-length model. Consider a face \mathcal{U}' of \mathcal{U} . Then the cone \mathcal{A} generated by \mathcal{U}' is a face of the cone \mathcal{C} and $\mathcal{R}_{\mathcal{A}} = \mathcal{R} \cap \mathcal{U}'$. Continuing to use bars to denote the quantities in the unrestricted-length model, let $\overline{\mathcal{R}}' = \{\langle z, 1 \rangle : z \in \mathcal{R}_{\mathcal{A}}\}$ and let $\overline{\mathcal{A}}$ be the cone generated by this set. Then $\overline{\mathcal{R}}_{\overline{\mathcal{A}}} = \overline{\mathcal{R}}'$, and $\overline{\mathcal{A}} = \{\langle \zeta, t \rangle : \zeta \in \mathcal{A}, t > 0, \zeta/t \in \mathcal{U}'\} \cup \{\langle \mathbf{0}, 0 \rangle\}$. Since, by definition, $\overline{T}_{\langle z, 1 \rangle} = T_z$, for all $z \in \mathcal{R}$, we have $\overline{\mathcal{R}}_{\overline{\mathcal{A}}}^{\mathrm{id}} = \{\langle z, 1 \rangle : z \in \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}\}$. In particular, $\mathcal{R}_{\mathcal{A}} \neq \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}$ implies $\overline{\mathcal{R}}_{\overline{\mathcal{A}}} \neq \overline{\mathcal{R}}_{\overline{\mathcal{A}}}^{\mathrm{id}}$ and $\overline{\mathbb{A}}$ is the set of faces $\overline{\mathcal{A}}$ such that $\mathcal{A} \in \mathbb{A}$.

If V^+ satisfies the conditions of Theorem 2.17, then \overline{V}^+ satisfies the conditions of Theorem 2.15. Similarly, Condition 3.3 transfers to Condition 3.2. If V^+ satisfies Condition 3.5, then \overline{V}^+ also satisfies that condition.

Since for all $k \in \mathbb{Z}_{>0}$ and $x \in D_k$, $\overline{F}^{\beta}_{\langle \mathbf{0}, 0 \rangle, \langle x, k \rangle} = F^{\beta}_{\mathbf{0}, x, k}$ and $\overline{\mathcal{I}}_{\langle x, k \rangle} = \mathcal{I}_x$,

(3.10)
$$\mathbb{E}\left[\sup_{n\geq 1}n^{-1}\mathbb{E}[\overline{F}^{\beta}_{\langle \mathbf{0},0\rangle,n\langle x,k\rangle} \,|\, \overline{\mathcal{I}}_{\langle x,k\rangle}]\right] = \mathbb{E}\left[\sup_{n\geq 1}n^{-1}\mathbb{E}[F^{\beta}_{\mathbf{0},nx,nk} \,|\, \mathcal{I}_x]\right]$$

Therefore, if the right-hand side is finite (as assumed, e.g., in Theorem 2.17), then so is the left-hand side. In this case, Theorem 2.15 implies that with P-probability one, $\overline{\Lambda}^{\beta}_{\overline{\mathcal{A}}}$ is finite on $\overline{\mathcal{A}}$. Then, from (2.4) and (2.6), we get that, P-almost surely, $\overline{\Lambda}^{\beta}_{\overline{\mathcal{A}}}(\langle \zeta, t \rangle) = t \Lambda^{\beta}_{\mathcal{U}', \text{res}}(\zeta/t)$ for all $\zeta \in \mathcal{A}$ and t > 0 with $\zeta/t \in \mathcal{U}'$. This implies that, with P-probability one, $\Lambda^{\beta}_{\overline{\mathcal{A}}', \text{res}}$ is finite on \mathcal{U}' , and as mentioned above Remark 2.7, this implies that, P-almost surely, $\Lambda^{\beta, \text{usc}}_{\mathcal{U}', \text{res}}$ is the unique continuous extension of $\Lambda^{\beta}_{\mathcal{U}', \text{res}}$ to \mathcal{U}' . By Theorem 2.15, P-almost surely, $\overline{\Lambda}^{\beta, \text{usc}}_{\overline{\mathcal{A}}}$ is the unique continuous extension of $\overline{\Lambda}^{\beta}_{\overline{\mathcal{A}}}$ to $\overline{\mathcal{A}}$. Therefore, with P-probability one, $\overline{\Lambda}^{\beta, \text{usc}}_{\overline{\mathcal{A}}}(\langle \zeta, t \rangle) = t \Lambda^{\beta, \text{usc}}_{\mathcal{U}', \text{res}}(\zeta/t)$ for all $\zeta \in \mathcal{A}$ and t > 0 with $\zeta/t \in \mathcal{U}'$.

Remark 3.20 (Extension to $|\mathcal{R}| = \infty$). It is natural to wonder to what extent our methods can be expected to generalize to the setting of an unbounded number of steps. Some extension in this direction is undoubtedly possible under some hypotheses on the reference walk and the potential. At this point, however, it is not clear to us what the correct conditions for such an extension are in general. The place to start any such program is the existence of the free energy and the shape theorem and then development of a variational representation of the free energy in terms of cocycles similar to [37, Theorem 2.14]. With these inputs in hand, our methods should mostly extend (as related methods have already done in the fully continuous setting of the KPZ equation and its associated continuum directed polymer [41]), though one expects additional technical complications.

The proof of Theorem 3.14 is given in Section 6. Sections 4 and 5 develop a more general theory from which Theorem 3.14 follows.

4. Recovering cocycles

In this section, we construct the generalized Busemann functions mentioned in the introduction. We then use them to construct the semi-infinite polymer measures in Theorem 3.14.

Definition 4.1. Given a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself), consider a Borel-measurable function

$$B: \{(x,y): x, y \in \mathbb{Z}^d, y - x \in \mathcal{G}_{\mathcal{A}}\} \times \Omega \to \mathbb{R}.$$

- (a) *B* is said to be a cocycle, if for \mathbb{P} -almost every ω , $B(v, x, \omega) + B(x, y, \omega) = B(v, y, \omega)$ for all $v, x, y \in \mathbb{Z}^d$ with $x v, y x \in \mathcal{G}_A$.
- (b) B is T-covariant if for \mathbb{P} -almost every ω , $B(x+v, y+v, \omega) = B(x, y, T_v \omega)$ for all $v, x, y \in \mathbb{Z}^d$ with $y-x \in \mathcal{G}_A$.
- (c) For $\beta \in (0, \infty]$, we say that B β -recovers, if for \mathbb{P} -almost every ω , we have for all $x \in \mathbb{Z}^d$,

(4.1)
$$\sum_{z \in \mathcal{R}_A} p(z) \exp\left\{-\beta B(x, x+z, \omega) - \beta V(T_x \omega, z)\right\} = 1, \text{ for } \beta < \infty \qquad and$$

- (4.2) $\min_{z \in \mathcal{R}_A} \left\{ B(x, x + z, \omega) + V(T_x \omega, z) \right\} = 0, \text{ for } \beta = \infty.$
 - (d) A cocycle B on \mathcal{A} is said to be $L^1(\mathbb{P})$ if $\mathbb{E}[|B(\mathbf{0}, z)|] < \infty$ for all $z \in \mathcal{R}_{\mathcal{A}}$. If also $\mathbb{E}[B(\mathbf{0}, z)] = 0$ for all $z \in \mathcal{R}_{\mathcal{A}}$, then B is centered.
 - (e) An $L^1(\mathbb{P})$ T-covariant β -recovering cocyle is called a generalized Busemann function (in inverse temperature β).

For a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself), recall that $W_{\mathcal{A}}$ is the linear span of $\mathcal{R}_{\mathcal{A}}$. Let $\mathcal{I}_{\mathcal{A}}$ be the σ -algebra generated by the events that are invariant under T_z for all $z \in \mathcal{G}_{\mathcal{A}}$. For a *T*-covariant $L^1(\mathbb{P})$ cocycle B on \mathcal{A} , the invariance of \mathbb{P} under the shifts T_z implies that $\mathbb{E}[B(x, y) | \mathcal{I}_{\mathcal{A}}] = \mathbb{E}[B(\mathbf{0}, y - x) | \mathcal{I}_{\mathcal{A}}]$ for all $x, y \in \mathcal{G}_{\mathcal{A}}$. Consequently, there exists a unique (possibly random) vector $\mathbf{m}(B) \in W_{\mathcal{A}}$ such that

(4.3)
$$\mathbb{E}[B(x,y) \mid \mathcal{I}_{\mathcal{A}}] = \mathbf{m}(B) \cdot (y-x).$$

Remark 4.2 (Space-time cocycles). If we write a restricted-length model as an unrestricted-length model, as in Remark 2.2, then a cocycle in the unrestricted-length version corresponds to a *space-time cocycle* in the restricted-length version. Precisely, let $\overline{\mathcal{G}}_{\mathcal{A}}$ be the group generated by $\{\langle z, 1 \rangle : z \in \mathcal{R}_{\mathcal{A}}\}$. Lemma A.3 gives a characterization of this group. Consider a Borel-measurable function

$$B: \{(x, j, y, k) \in \mathbb{Z}^d \times \mathbb{Z} \times \mathbb{Z}^d \times \mathbb{Z}, \langle y - x, k - j \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}\} \times \Omega \longrightarrow \mathbb{R}.$$

(a) B satisfies the space-time cocycle property if for \mathbb{P} -almost every ω , all $x, y, z \in \mathbb{Z}^d$, and all $j, k, \ell \in \mathbb{Z}$ with $\langle y - x, k - j \rangle, \langle z - y, \ell - k \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$,

(4.4)
$$B(x, j, y, k, \omega) + B(y, k, z, \ell, \omega) = B(x, j, z, \ell, \omega).$$

- (b) B is T-covariant if for P-almost every ω, B(x + v, j + ℓ, y + v, k + ℓ, ω) = B(x, j, y, k, T_vω) for all j, k, ℓ ∈ Z and x, y, v ∈ Z^d such that ⟨y x, k j⟩ ∈ G_A. In particular, B(x, j + ℓ, y, k + ℓ, ω) = B(x, j, y, k, ω), i.e., the cocycle depends on the time coordinates only through the size of the time increment. Consequently, in the case of a space-time directed polymer, described in Example 2.5(2), one can drop the time coordinates and write B(x, y), since (y x) · û determines the time increment.
- (c) For $\beta \in (0, \infty]$, the β -recovery property is that for \mathbb{P} -almost every ω , we have for all $x \in \mathbb{Z}^d$ and $j \in \mathbb{Z}$,

(4.5)
$$\sum_{z \in \mathcal{R}_{\mathcal{A}}} p(z) \exp\left\{-\beta B(x, j, x+z, j+1, \omega) - \beta V(T_x \omega, z)\right\} = 1, \text{ if } \beta < \infty, \text{ and}$$

(4.6)
$$\min_{z \in \mathcal{R}_{\mathcal{A}}} \left\{ B(x, j, x+z, j+1, \omega) + V(T_x \omega, z) \right\} = 0, \text{ if } \beta = \infty.$$

- (d) A space-time cocycle is said to be $L^1(\mathbb{P})$ if $\mathbb{E}[|B(\mathbf{0}, 0, z, 1)|] < \infty$ for all $z \in \mathcal{R}_A$. If in addition $\mathbb{E}[B(\mathbf{0}, 0, z, 1)] = 0$ for all $z \in \mathcal{R}_A$, then B is centered.
- (e) For a shift-covariant $L^1(\mathbb{P})$ space-time cocycle B, the random vector $\mathbf{m}(B) \in W_{\mathcal{A}} \times \mathbb{R}$ is the unique vector such that $\mathbb{E}[B(x, j, y, k) | \mathcal{I}_{\mathcal{A}}] = \mathbf{m}(B) \cdot \langle y x, k j \rangle$ for all $x, y \in \mathbb{Z}^d$ and $j, k \in \mathbb{Z}$ such that $\langle x, j \rangle, \langle y, k \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$.

We next note a property of space-time cocycles which explains their relationship to the cocycles one sees when working with unrestricted-length models and which will allow us to connect our results to some of the previous literature. In words, it says that a space-time cocycle can be decomposed into a cocycle in the sense of Definition 4.1 plus a shift-invariant linear function of the time increment.

Lemma 4.3. Fix a face $A \in \mathbb{A}$ (possibly C itself). Let B be a space-time T-covariant cocycle as in Remark 4.2. Then there exists an \mathcal{I}_A -measurable random variable c and a cocycle \overline{B} (as in Definition 4.1) such that \mathbb{P} -almost surely, $B(x, j, y, k) = \overline{B}(x, y) + c(k-j)$, for all $(x, j, y, k) \in \mathbb{Z}^d \times \mathbb{Z} \times \mathbb{Z}^d \times \mathbb{Z}$ with $\langle y - x, k - j \rangle \in \overline{\mathcal{G}}_A$.

Proof. Let A denote the set of $m \in \mathbb{Z}$ such that $\langle \mathbf{0}, m \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$. Note that A is an additive subgroup of Z. Fix $m \in A$. Take any $z \in \mathcal{R}_{\mathcal{A}}$. Then $\langle z, 1 \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$ and $\langle z, m+1 \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$ and so \mathbb{P} -almost surely,

$$B(\mathbf{0}, 0, \mathbf{0}, m, \omega) + B(\mathbf{0}, 0, z, 1, \omega) = B(\mathbf{0}, 0, z, m + 1, \omega) = B(\mathbf{0}, 0, z, 1, \omega) + B(z, 0, z, m, \omega)$$

= $B(\mathbf{0}, 0, z, 1, \omega) + B(\mathbf{0}, 0, \mathbf{0}, m, T_z\omega).$

This implies that \mathbb{P} -almost surely, $B(\mathbf{0}, 0, \mathbf{0}, m, \omega) = B(\mathbf{0}, 0, \mathbf{0}, m, T_z \omega)$ for all $z \in \mathcal{R}_A$. Consequently, for any $m \in A$, $B(\mathbf{0}, 0, \mathbf{0}, m, \omega)$ is $\mathcal{I}_{\mathcal{A}}$ -measurable. Call $c_m = \mathbb{E}[B(\mathbf{0}, 0, \mathbf{0}, m) | \mathcal{I}_{\mathcal{A}}]$, which satisfies $c_m =$ $B(\mathbf{0}, 0, \mathbf{0}, m, \omega)$, \mathbb{P} -almost surely.

If $m, n \in A$, then $n + m \in A$ and the cocycle property of B implies $c_{m+n} = c_m + c_n$. This implies that there exists an $\mathcal{I}_{\mathcal{A}}$ -measurable random variable c such that $c_m = cm$ for all $m \in \mathcal{A}$. Then we have \mathbb{P} -almost surely, for any $m \in A$, $B(\mathbf{0}, 0, \mathbf{0}, m) = cm$.

By the covariance of B, we have

$$B(x,k,x,j,\omega) = B(\mathbf{0},0,\mathbf{0},j-k,T_x\omega) = c(j-k),$$

for \mathbb{P} -almost every ω and any $x \in \mathbb{Z}^d$ and $(j,k) \in \mathbb{Z} \times \mathbb{Z}$ with $k - j \in A$.

Now, take $(x, y, j, k, m, n) \in \mathbb{Z}^d \times \mathbb{Z}^d \times \mathbb{Z} \times \mathbb{Z} \times \mathbb{Z} \times \mathbb{Z}$ with $\langle y - x, k - j \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$ and $\langle y - x, n - m \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$. Then

B(x, j, y, k) - B(x, m, y, n) = B(x, 0, y, k - j) - B(x, 0, y, n - m) = B(y, n - m, y, k - j) = c((k - j) - (n - m)).Thus, if $x, y \in \mathbb{Z}^d \times \mathbb{Z}^d$ are such that $y - x \in \mathcal{G}_A$, then

$$\overline{B}(x,y) = B(x,j,y,k) - c(k-j) \text{ for any } j,k \in \mathbb{Z} \times \mathbb{Z} \text{ such that } \langle y-x,k-j \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$$

is well defined and we have

$$B(x, j, y, k) = B(x, y) + c(k - j)$$

as desired.

We document next a property of the random vector $\mathbf{m}(B)$, which will be of use to us in what follows.

Lemma 4.4. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Let B be an $L^1(\mathbb{P})$ T-covariant β -recovering cocycle on $\mathcal{A}. \quad If \mathbb{E}[\mathbf{m}(B)] \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi) \text{ for some } \xi \in \text{ri }\mathcal{A}, \text{ then } \mathbf{m}(B) \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi), \mathbb{P}\text{-almost surely.} \quad If additionally \\ \mathbb{E}[\mathbf{m}(B)] \in \text{ext } \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi), \text{ then } \mathbf{m}(B) = \mathbb{E}[\mathbf{m}(B)], \mathbb{P}\text{-almost surely.}$

Proof. Let $\xi \in \operatorname{ri} \mathcal{A}$ be such that $\mathbb{E}[\mathbf{m}(B)] \in \partial \Lambda_{\mathcal{A}}^{\beta, \operatorname{usc}}(\xi)$. By Lemma A.1, this implies that $\Lambda^{\beta}(\xi) = \Lambda_{\mathcal{A}}^{\beta, \operatorname{usc}}(\xi) = \mathbb{E}[\mathbf{m}(B)] \cdot \xi$. By the variational formula [37, Theorem 2.14], we have \mathbb{P} -almost surely, $\Lambda^{\beta}(\zeta) \leq \mathbf{m}(B) \cdot \zeta$ for all $\zeta \in \mathcal{A}$. Thus, it must be that, \mathbb{P} -almost surely,

$$0 = \Lambda^{\beta}(\xi) - \mathbf{m}(B) \cdot \xi \leq \sup_{\zeta \in \mathcal{A}} (\Lambda^{\beta}(\zeta) - \mathbf{m}(B) \cdot \zeta) \leq 0.$$

Then applying [59, Theorem 4.21], we obtain that $\mathbf{m}(B) \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi)$. If additionally $\mathbb{E}[\mathbf{m}(B)]$ is an extreme point of $\partial \Lambda_{A}^{\beta,\text{usc}}(\xi)$, then $\mathbf{m}(B) = \mathbb{E}[\mathbf{m}(B)]$ P-almost surely by the definition of an extreme point.

For a face \mathcal{A} of \mathcal{C} and a non-empty set $I \subset \mathbb{Z}^d$, define the set of points reachable from some point of I taking steps in $\mathcal{R}_{\mathcal{A}}$,

(4.7)
$$I_{\mathcal{A}}^{\geq} = \{ y \in \mathbb{Z}^d : y - x \in \mathcal{G}_{\mathcal{A}}^+ \text{ for some } x \in I \}.$$

Also, define the set of points unreachable from all points in I with steps in $\mathcal{R}_{\mathcal{A}}$,

(4.8)
$$I_{\mathcal{A}}^{<} = \mathbb{Z}^{d} \setminus I_{\mathcal{A}}^{\geq} = \{ y \in \mathbb{Z}^{d} : y - x \notin \mathcal{G}_{\mathcal{A}}^{+} \text{ for all } x \in I \}.$$

If the model is undirected on \mathcal{A} , i.e. $\mathbf{0} \in \mathrm{ri}\mathcal{U}_{\mathcal{A}}$, then $\mathcal{G}_{\mathcal{A}}^+ = \mathcal{G}_{\mathcal{A}}$ so that $I_{\mathcal{A}}^{\geq} = \bigcup_{x \in I} (x + \mathcal{G}_{\mathcal{A}})$. Recall the countable (possibly finite or empty) set of inverse temperatures and extremal supergradients $\mathcal{D}_{\mathcal{A}}$ defined in the paragraph containing (3.8). For each $\mathcal{A} \in \mathbb{A}$, let $\mathcal{E}_{\mathcal{A}} = \{(\beta, m, x, z) : (\beta, m) \in \mathcal{D}_{\mathcal{A}}, x \in \mathbb{A}\}$ $\mathbb{Z}^d, z \in \mathcal{R}_{\mathcal{A}}$ and let $\hat{\Omega} = \Omega \times \prod_{\mathcal{A} \in \mathbb{A}} \mathbb{R}^{\mathcal{E}_{\mathcal{A}}}$ be equipped with the product topology and its Borel σ -algebra, $\widehat{\mathfrak{S}}$. For $\widehat{\omega} \in \widehat{\Omega}$, let $\pi_{\Omega}(\widehat{\omega})$ be the projection to its Ω coordinate ω . We will sometimes write this as $\omega(\widehat{\omega})$. For each $\mathcal{A} \in \mathbb{A}$ and $(\beta, m, x, z) \in \mathcal{E}_{\mathcal{A}}$ let $B^{\mathcal{A},\beta,m}(x, x + z, \hat{\omega})$ be the $(\mathcal{A}, \beta, m, x, z)$ -coordinate of $\hat{\omega}$. Let $\widehat{T} = \{\widehat{T}_v : v \in \mathbb{Z}^d\}$ be the $\widehat{\mathfrak{S}}$ -measurable group of transformations that map

$$(\omega, \{b_{\mathcal{A},\beta,m,x,z} : \mathcal{A} \in \mathbb{A}, (\beta,m,x,z) \in \mathcal{E}_{\mathcal{A}}\})$$

 to

$$(T_v\omega, \{b_{\mathcal{A},\beta,m,x+v,z} : \mathcal{A} \in \mathbb{A}, (\beta,m,x,z) \in \mathcal{E}_{\mathcal{A}}\}).$$

As such, we have $\pi_{\Omega}(\hat{T}_v\hat{\omega}) = T_v\pi_{\Omega}(\hat{\omega})$ and for all $\mathcal{A} \in \mathbb{A}$ and $(\beta, m, x, z) \in \mathcal{E}_{\mathcal{A}}, B^{\mathcal{A},\beta,m}(x, x+z, \hat{T}_v\hat{\omega}) = B^{\mathcal{A},\beta,m}(x+v, x+v+z, \hat{\omega})$. Note that $\hat{\Omega}$ satisfies the hypotheses on Ω in Section 2.2.

Theorem 4.5. For each face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself) where $\mathcal{D}_{\mathcal{A}} \neq \emptyset$, assume that V satisfies Conditions 3.2 and 3.5 on \mathcal{A} . There exists a \hat{T} -invariant probability measure $\hat{\mathbb{P}}$ on $(\hat{\Omega}, \hat{\mathfrak{S}})$ and a real-valued measurable function $B^{\mathcal{A},\beta,m}(x,y,\hat{\omega})$ of $\{(\mathcal{A},\beta,m,x,y,\hat{\omega}) : \mathcal{A} \in \mathbb{A}, (\beta,m) \in \mathcal{D}_{\mathcal{A}}, x \in \mathbb{Z}^d, y \in \mathbb{Z}^d, y - x \in \mathcal{G}_{\mathcal{A}}, \hat{\omega} \in \hat{\Omega}\}$ such that the following hold.

- (a) For any event $A \in \mathfrak{S}$, $\widehat{\mathbb{P}}(\pi_{\Omega}(\widehat{\omega}) \in A) = \mathbb{P}(A)$.
- (b) There exists a \hat{T} -invariant event $\hat{\Omega}_{coc}$ with $\hat{\mathbb{P}}(\hat{\Omega}_{coc}) = 1$ such that for each face $\mathcal{A} \in \mathbb{A}$, $\hat{\omega} \in \hat{\Omega}_{coc}$, $u, v, x, y \in \mathbb{Z}^d$ such that $x u \in \mathcal{G}_{\mathcal{A}}$, $y x \in \mathcal{G}_{\mathcal{A}}$, and $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$,

(4.9)
$$B^{\mathcal{A},\beta,m}(x+v,y+v,\widehat{\omega}) = B^{\mathcal{A},\beta,m}(x,y,\widehat{T}_v\widehat{\omega})$$

(4.10) $B^{\mathcal{A},\beta,m}(u,x,\widehat{\omega}) + B^{\mathcal{A},\beta,m}(x,y,\widehat{\omega}) = B^{\mathcal{A},\beta,m}(u,y,\widehat{\omega})$

(4.11)
$$\sum_{z \in \mathcal{R}_{\mathcal{A}}} p(z) \exp\left\{-\beta B^{\mathcal{A},\beta,m}(x,x+z,\widehat{\omega}) - \beta V(\omega(\widehat{T}_x\widehat{\omega}),z)\right\} = 1, \ if \ \beta < \infty$$

(4.12)
$$\min_{z \in \mathcal{R}_{\mathcal{A}}} \left\{ B^{\mathcal{A},\beta,m}(x,x+z,\widehat{\omega}) + V(\omega(\widehat{T}_x\widehat{\omega}),z) \right\} = 0, \text{ if } \beta = \infty.$$

(c) For any face $\mathcal{A} \in \mathbb{A}$, $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$, and $x, y \in \mathbb{Z}^d$ such that $y - x \in \mathcal{G}_{\mathcal{A}}$, $B^{\mathcal{A},\beta,m}(x,y)$ is integrable under $\widehat{\mathbb{P}}$,

(4.13)
$$\widehat{\mathbb{E}}[B^{\mathcal{A},\beta,m}(x,y)] = m \cdot (y-x),$$

and $\widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})] = m.$

(d) Fix a face $\mathcal{A} \in \mathbb{A}$. If the random variables $\{(V(T_v\omega, z))_{z \in \mathcal{R}_{\mathcal{A}}} : v \in \mathbb{Z}^d\}$ are independent under \mathbb{P} , then for any non-empty set $I \subset \mathbb{Z}^d$, the variables

(4.14)
$$\left\{B^{\mathcal{A},\beta,m}(x,y,\widehat{\omega}): x \in I, y - x \in \mathcal{G}^+_{\mathcal{A}}, (\beta,m) \in \mathcal{D}_{\mathcal{A}}\right\}$$

are independent, under $\widehat{\mathbb{P}}$, of the variables $\{V(T_v\omega, z) : z \in \mathcal{R}_A, v \in I_A^<\}$.

Remark 4.6 (Restricted-length space-time cocycles). The above can be applied to restricted-length polymers as follows. Let \mathcal{U}' be a face of \mathcal{U} with $\mathcal{R}_{\mathcal{A}} \neq \mathcal{R}^{id}_{\mathcal{A}}$, where \mathcal{A} is the cone generated by \mathcal{U}' . Let $\beta \in (0, \infty]$. Write the restricted-length model in terms of an unrestricted-length model as described in Remark 2.2. If V satisfies Conditions 3.3 and 3.5, it satisfies the conditions of Theorem 2.17, and \overline{V} satisfies Conditions 3.2 and 3.5 on $\overline{\mathcal{A}}$, and hence it satisfies the conditions of Theorem 2.15. As explained in Remark 3.19, with \mathbb{P} -probability one, $\Lambda^{\beta,\text{usc}}_{\mathcal{U}',\text{res}}$ is finite on $\mathcal{U}', \overline{\Lambda}^{\beta,\text{usc}}_{\overline{\mathcal{A}}}$ is finite on $\overline{\mathcal{A}}$, and $\overline{\Lambda}^{\beta,\text{usc}}_{\overline{\mathcal{A}}}(\langle \zeta, t \rangle) = t\Lambda^{\beta,\text{usc}}_{\mathcal{U}',\text{res}}(\zeta/t)$ for all $\zeta \in \mathcal{A}$ and t > 0 with $\zeta/t \in \mathcal{U}'$. Therefore, if $\Lambda^{\beta,\text{usc}}_{\mathcal{U}',\text{res}}$ is \mathbb{P} -almost surely deterministic on \mathcal{U}' , then $\overline{\Lambda}^{\beta,\text{usc}}_{\overline{\mathcal{A}}}$ is \mathbb{P} -almost surely deterministic on $\overline{\mathcal{A}}$. Observe next that if $\xi \in \text{ri}\mathcal{U}'$, then $\overline{\xi} = \langle \xi, 1 \rangle \in (\text{ri}\overline{\mathcal{A}}) \backslash \{\langle 0, 0 \rangle\}$. Take $\overline{m} \in \overline{W}_{\overline{\mathcal{A}}} \cap \partial \overline{\Lambda}^{\beta,\text{usc}}_{\overline{\mathcal{A}}}(\langle \xi, 1 \rangle)$ for some $\xi \in \text{ri}\mathcal{U}$. (This set is non-empty, bounded, and has at least one extreme point, as explained at the beginning of Section 3.) Now all the hypotheses of Theorem 4.5 are satisfied for the unrestricted-length model, and the theorem produces the cocycles $\overline{B}^{\overline{\mathcal{A}},\beta,\overline{m}}(\langle x,k \rangle, \langle y,\ell \rangle, \hat{\omega}),$ $\langle y - x, k - \ell \rangle \in \overline{\mathcal{G}}_{\overline{\mathcal{A}}}$. Then, for the original restricted-length model, we define the space-time cocycles (see Remark 4.2) by setting

$$B^{\mathcal{U}',\beta,\overline{m}}(x,k,y,\ell,\widehat{\omega}) = \overline{B}^{\overline{\mathcal{A}},\beta,\overline{m}}(\langle x,k\rangle,\langle y,\ell\rangle,\widehat{\omega}).$$

In particular, if we write $\overline{m} = \langle m, c \rangle$, then

$$\widehat{\mathbb{E}}[B^{\mathcal{U}',\beta,\overline{m}}(x,k,y,\ell)] = \widehat{\mathbb{E}}[\overline{B}^{\mathcal{A},\beta,\overline{m}}(\langle x,k\rangle,\langle y,\ell\rangle)] = \overline{m}\cdot\langle y-x,\ell-k\rangle = m\cdot(y-x) + c(\ell-k)\cdot c(\ell-k)$$

A direct computation given in Appendix A shows that

(4.15)
$$c = \Lambda_{\mathcal{U}', \text{res}}^{\beta, \text{usc}}(\xi) - m \cdot \xi;$$

moreover, m is in the superdifferential at ξ of the concave function that is equal to $\Lambda_{\mathcal{A}}^{\beta,\text{usc}}$ on \mathcal{U}' and is set to $-\infty$ outside \mathcal{U}' . When $\mathcal{I}_{\mathcal{A}}$ is trivial, this is also the value of c from Lemma 4.3.

Remark 4.7 (RWRE on \mathbb{Z}). Consider the models in Figure 2.1. As noted in Remark 3.13, in the recurrent case, there are no points in $\mathcal{D}_{\mathcal{C}}$ with $\beta = 1$ that can serve as input for Theorem 4.5. In contrast, in the transient cases, there exists a unique point (1, m), where $m = -\Lambda^1(-1)$, that can belong to $\mathcal{D}_{\mathcal{C}}$. Using this as input in Theorem 4.5 yields a cocycle $B^{\mathcal{C},1,m}$, which we will denote more concisely as \overline{B}^m .

Next, consider the restricted-length version of the RWRE, reformulated as an unrestricted-length model per Remark 4.6. Here, the set of admissible steps becomes $\overline{\mathcal{R}} = \{\langle 1, 1 \rangle, \langle -1, 1 \rangle\}$, and $\langle 0, 0 \rangle \notin \overline{\mathcal{U}}$. Consequently, if we take the full cone $\overline{\mathcal{C}} \subset \mathbb{R}^2$ generated by $\overline{\mathcal{R}}$, there are no restrictions on the choice of ξ in defining $\mathcal{D}_{\overline{\mathcal{C}}}$.

Remark 4.6 explains how the superdifferential of the limiting free energy for this unrestricted-length model connects back to the superdifferential of Λ_{res}^1 , which is either the single point $\frac{d}{dt}\Lambda_{\text{res}}^1(t)$ for $t \in (-1,1) \setminus \{0\}$, or an interval $[0, \frac{d}{dt}\Lambda_{\text{res}}^1(0-)]$ when t = 0. See Figure 2.1.

Using this setup, Theorem 4.5 produces space-time cocycles (as described in Remark 4.6) for a countable collection of $m \in \partial \Lambda_{res}^1(t)$ with $t \in (-1, 1)$. By Lemma 4.3, these cocycles take the form

$$B^{m}(x, j, y, k) = \overline{B}^{m}(x, y) + c(k - j),$$

where \overline{B}^m is a cocycle and c is a deterministic constant.

The corresponding Gibbs measure, constructed via Theorem 3.14, is the Doob *h*-transform of the RWRE, with the one-step transition probability kernel

$$\pi_z^m = \pi_z e^{\overline{B}^m(\mathbf{0},z) + c}$$

These measures were previously constructed and studied in [70] as the minimizers of the level-3 to level-1 contraction principle for quenched large deviations in a more general one-dimensional RWRE model. See, in particular, Lemma 1.6, Definition 5.6, and Theorem 5.17 of that paper. Our constant c corresponds to the parameter r in their notation.

 B^m is a genuine cocycle if and only if c = 0, in which case $B^m = \overline{B}^m$. In this case, (5.15) and (5.17) in [70] imply that this cocycle coincides with the one obtained in the first paragraph of this remark, when applying Theorem 4.5 to the unrestricted-length RWRE. Furthermore, by (4.15), c = 0 implies $\Lambda^1_{\text{res}}(t) = mt$ and, by the observation following (4.15), m is also in the superdifferential of Λ^1 . This forces $m \in \{\Lambda^1(-1), 0\}$. Thus, in the transient nestling cases, the points $t \in (-1, 0)$ for which c = 0 correspond precisely to the points in the interval left of 0 where Λ^1_{res} is linear (see Figures 2.1B and 2.1C). In the non-nestling case, there exists a unique such t, which is the sole point in (-1, 0) where Λ^1_{res} and Λ^1 intersect. See Figure 2.1D.

We conclude this remark with a high-level explanation of the significance of the unique intersection point mentioned at the end of the previous paragraph. Let $t_0 \in (-1,0)$ denote this point. By the maximum entropy principle (see [59, Section 5.3] for a similar application), conditioning the RWRE on $\{\tau_{-n} < \infty\}$ and taking $n \to \infty$ causes the path measure to converge almost surely to the Doob *h*-transformed RWRE tilted by the cocycle \overline{B}^m , with $m = -\Lambda^1(-1)$. But since $\overline{B}^m = B^m$, Lemma 1.6 and Theorem 5.17 in [70] along with another application of the maximum entropy principle imply that this tilted RWRE corresponds exactly to conditioning the original RWRE on having the asymptotic velocity t_0 . Succinctly, if we condition the walk on the (zero probability) event $\{X_n \to -\infty\}$, it will proceed ballistically with velocity t_0 .

Remark 4.8. Our construction shows that the claim in Theorem 4.5(d) continues to hold if independence is replaced by mixing. That is, if $\sigma\{(V(T_v\omega, z))_{z\in\mathcal{R}_A} : v\in A\}$ and $\sigma\{(V(T_v\omega, z))_{z\in\mathcal{R}_A} : v\in A'\}$ mix under \mathbb{P} at rate r_k , when $A, A' \subset \mathbb{Z}^d$ are separated by distance k, then $\sigma\{B^{\mathcal{A},\beta,m}(x,y,\hat{\omega}) : x\in I, y-x\in\mathcal{G}^+_{\mathcal{A}}, (\beta,m)\in\mathcal{D}_{\mathcal{A}}\}$ and $\sigma\{V(T_v\omega, z) : z\in\mathcal{R}_{\mathcal{A}}, v\in I'\}$ mix at the same rate r_k if $I \subset \mathbb{Z}^d$ and $I' \subset I^{\mathcal{A}}_{\mathcal{A}}$ are separated by distance k. See [12] for a definition and basic properties of the strong mixing rate between σ -algebras. \bigtriangleup

We prove the theorem towards the end of the section. For a high-level summary of the ideas, see Section 1.1 in the introduction. We begin by defining a few objects and proving a few bounds. Assume the hypotheses of the above theorem for the rest of this section.

For a face $\mathcal{A} \in \mathbb{A}$ and a pair $(\beta, m) \in \mathcal{D}_{\mathcal{A}}, m \in W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi)$ for some $\xi \in \text{ri } \mathcal{A} \setminus \{\mathbf{0}\}$. Define $\hat{u}, h \in \mathbb{R}^d$ in the following way:

If either Condition 3.2(b) or 3.2(c) holds, then $\mathbf{0} \notin \mathcal{U}_{\mathcal{A}}$. [37, Lemma A.1] implies that there exists a vector u such that $u \cdot z > 0$ for all $z \in \mathcal{R}_{\mathcal{A}}$. Then $u \cdot \xi > 0$. Define $\hat{u} = u/(u \cdot \xi)$ and $h = \hat{u} - m$.

If instead Condition 3.2(a) holds, then $\mathbf{0} \in \mathcal{U}_{\mathcal{A}}$ and by the definition of $\mathcal{D}_{\mathcal{A}}$ (above (3.8)), we have that $\Lambda^{\beta}(\xi) \neq 0$ and hence $\Lambda^{\beta}(\xi) < 0$. Also by Remark 3.6, $\Lambda^{\beta}(\xi) > -\infty$. Define $\hat{u} = \frac{m}{\Lambda^{\beta}(\xi)}$ and $h = \frac{1-\Lambda^{\beta}(\xi)}{\Lambda^{\beta}(\xi)}m$. Note that $\hat{u} \cdot \xi = \frac{m}{\Lambda^{\beta}(\xi)} \cdot \xi = 1$, since $m \cdot \xi = \Lambda^{\beta, \text{usc}}_{\mathcal{A}}(\xi) = \Lambda^{\beta}(\xi)$ by Lemma A.1.

With the above definitions, we can now consider that throughout the proof, each $\mathcal{A} \in \mathbb{A}$ and $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$ are accompanied by ξ , \hat{u} , and h that satisfy these properties:

(4.16)
$$m = \hat{u} - h,$$
$$\hat{u} \cdot \xi = 1,$$
$$\hat{u} \cdot z > 0 \text{ for all } z \in \mathcal{R}_{\mathcal{A}}, \text{ if either Condition 3.2(b) or 3.2(c) holds, and}$$
$$h = (1 - \Lambda^{\beta}(\xi))\hat{u}, \text{ if Condition 3.2(a) holds.}$$

Let

(4.17)
$$R_0 = \sum_{z \in \mathcal{R}} |z \cdot \hat{u}|$$

and take $R > R_0$. For $x \in \mathbb{Z}^d$ and $t \in \mathbb{R}$, we introduce the following slab at \hat{u} -level t of width R,

$$L_{\mathcal{A},x,t} = L_{\mathcal{A},x,t}(R,\hat{u}) = \{ v \in \mathbb{Z}^d : v - x \in \mathcal{G}_{\mathcal{A}} \text{ and } t \leq v \cdot \hat{u} < t + R \}.$$

Note that if $y \in \mathbb{Z}^d$ with $y - x \in \mathcal{G}_A$, then $L_{A,x,t} = L_{A,y,t}$. We abbreviate $L_{A,0,t} = L_{A,t}$. For $\beta < \infty$, $x \in \mathbb{Z}^d$, $t > x \cdot \hat{u}$, and $\varepsilon \ge 0$ define the (regularized) tilted point-to-level partition function

$$Z^{\mathcal{A},\beta,\varepsilon}_{x,(t)}(\omega) = \sum_{v \in L_{\mathcal{A},x,t}} Z^{\beta}_{x,v}(\omega) e^{\beta h \cdot (v-x) - \varepsilon |v|_1}.$$

Above, the regularization by introducing $\varepsilon |v|_1$ in the exponent is there solely to ensure finiteness.

Recall the convention that $Z_{x,v}^{\beta} = 0$ if there are no admissible paths from x to v. Hence, the above sum is only over sites v that are accessible from x, i.e., $v - x \in \mathcal{G}_{\mathcal{A}}^+$. Since $\xi \cdot \hat{u} = 1$ there exists a $z \in \mathcal{R}_{\mathcal{A}}$ such that $z \cdot \hat{u} > 0$. Since $z \cdot \hat{u} < R$ and $x \cdot \hat{u} < t$, there is at least one $v \in L_{\mathcal{A},x,t}$ which is reachable from x. The corresponding free energy is

$$F_{x,(t)}^{\mathcal{A},\beta,\varepsilon}(\omega) = \frac{1}{\beta} \log Z_{x,(t)}^{\mathcal{A},\beta,\varepsilon}(\omega).$$

At zero temperature, define

$$F_{x,(t)}^{\mathcal{A},\infty,0}(\omega) = \sup_{v \in L_{\mathcal{A},x,t}} (F_{x,v}^{\infty}(\omega) + h \cdot (v - x)).$$

Recall the convention that $F_{x,v}^{\infty} = -\infty$ if there are no admissible paths from x to v, meaning that v is not considered in the supremum.

Condition 3.2 guarantees that $F_{x,(t)}^{\mathcal{A},\beta,\varepsilon}(\omega)$ is finite and integrable, which we now show.

Lemma 4.9. If $\beta = \infty$ or Condition 3.2(b) or (c) is satisfied, take $\varepsilon = 0$. If $\beta < \infty$ and Condition 3.2(a) holds, take $\varepsilon > 0$. Then, for each $\mathcal{A} \in \mathbb{A}$, $x \in \mathbb{Z}^d$, $\beta \in (0, \infty]$, and $t > x \cdot \hat{u}$,

$$\mathbb{E}[|F_{x,(t)}^{\mathcal{A},\beta,\varepsilon}|] < \infty.$$

Proof. Let $\mathcal{A}, x, \beta, \varepsilon$, and t be as in the statement. Choose y such that $y \in L_{\mathcal{A},x,t}$ and $y - x \in \mathcal{G}^+_{\mathcal{A}}$. The hypothesis on t and the definition of R ensure that such a y exists. For such y, let $x_{0:n}$ be an admissible path with $x_0 = x$ and $x_n = y$. Then for $\beta \in (0, \infty)$, we have

$$F_{x,(t)}^{\mathcal{A},\beta,\varepsilon} = \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} Z_{x,v}^{\beta} e^{\beta h \cdot (v-x) - \varepsilon |v|_1} \ge \frac{1}{\beta} \log \left(Z_{x,y}^{\beta} e^{\beta h \cdot (y-x) - \varepsilon |y|_1} \right), \text{ and}$$

$$F_{x,(t)}^{\mathcal{A},\infty,0} \ge F_{x,y}^{\infty} + h \cdot (y-x).$$

Recall that $F_{x,y}^{\infty} \geq \frac{1}{\beta} \log Z_{x,y}^{\beta}$ and then bound

$$\mathbb{E}\left[\frac{1}{\beta}\log Z_{x,y}^{\beta}\right] \ge \sum_{i=0}^{n-1} \left[\log p(x_{i+1} - x_i) - \mathbb{E}\left[V(T_{x_i}\omega, x_{i+1} - x_i)\right]\right] > -\infty$$

For the upper bound, first consider the case where Condition 3.2(a) is satisfied so that $V(\omega, z) \ge 0$ for all $z \in \mathcal{R}_{\mathcal{A}}$. Then for all v such that $v - x \in \mathcal{G}_{\mathcal{A}}^+$, $Z_{x,v}^{\beta} = \mathbb{E}_x \left[e^{-\beta \sum_{i=0}^{\tau_v - 1} V(T_{X_i}\omega, X_{i+1} - X_i)} \mathbb{1}_{\{\tau_v < \infty\}} \right] \le 1$. Recall that $\Lambda^{\beta}(\xi) < 0$ and $h = (1 - \Lambda^{\beta}(\xi))\hat{u}$. Then, with the convention that C is a chameleon constant that may change line-to-line,

$$\begin{aligned} F_{x,(t)}^{\mathcal{A},\beta,\varepsilon} &= \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} Z_{x,v}^{\beta} e^{\beta h \cdot (v-x) - \varepsilon |v|_{1}} \leqslant \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} e^{\beta (1-\Lambda^{\beta}(\xi)) \hat{u} \cdot v - \beta h \cdot x - \varepsilon |v|_{1}} \\ &\leqslant \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} e^{\beta (1-\Lambda^{\beta}(\xi))(t+R) - \beta h \cdot x - \varepsilon |v|_{1}} = (1-\Lambda^{\beta}(\xi))(t+R) - h \cdot x + \frac{1}{\beta} \log \sum_{v \in \mathbb{Z}^{d}} e^{-\varepsilon |v|_{1}} \\ &\leqslant (1-\Lambda^{\beta}(\xi))(t+R) - h \cdot x + \frac{1}{\beta} \log \left(C \sum_{k=0}^{\infty} k^{d} e^{-\varepsilon k}\right) \\ &\leqslant (1-\Lambda^{\beta}(\xi))(t+R) - h \cdot x + \frac{1}{\beta} \log \left(C \sum_{k=0}^{\infty} e^{-\varepsilon k/2}\right) \\ &\leqslant (1-\Lambda^{\beta}(\xi))(t+R) - h \cdot x + \beta^{-1} \log(C\varepsilon^{-1}), \end{aligned}$$

where the last bound applies for ε sufficiently small. At zero temperature, the equivalent bound is

(4.19)
$$F_{x,(t)}^{\mathcal{A},\infty,0} = \sup_{v \in L_{\mathcal{A},x,t}} (F_{x,v}^{\infty}(\omega) + h \cdot (v-x)) \leqslant \sup_{v \in L_{\mathcal{A},x,t}} ((1 - \Lambda^{\beta}(\xi))v \cdot \hat{u} - h \cdot x)$$
$$\leqslant (1 - \Lambda^{\beta}(\xi))(t+R) - h \cdot x.$$

Next, assume that either Condition 3.2(b) or (c) holds. Then $\mathbf{0} \notin \mathcal{U}_{\mathcal{A}}$ and there exists a $\delta > 0$ such that for all $z \in \mathcal{R}_{\mathcal{A}}$, $\hat{u} \cdot z \ge \delta > 0$. For $v \in L_{\mathcal{A},x,t}$, all paths from x to v have length at least $(t - x \cdot \hat{u})R^{-1}$ since $R > \sum_{z \in \mathcal{R}_{\mathcal{A}}} |z \cdot \hat{u}|$. Similarly all paths have length at most $(t + R - x \cdot \hat{u})\delta^{-1}$ since $z \cdot \hat{u} \ge \delta$ for all $z \in \mathcal{R}_{\mathcal{A}}$. Let $\ell = \ell(t) = \lfloor (t + R - x \cdot \hat{u})\delta^{-1} \rfloor$ be the maximum possible length. Let $\tilde{L}_{\mathcal{A},x,t}$ be the finite set of $v \in L_{\mathcal{A},x,t}$ which are reachable from x. The cardinality of $\tilde{L}_{\mathcal{A},x,t}$ is bounded above by $C\ell^d$ for some fixed constant C > 0 depending only on the dimension d and $\mathcal{R}_{\mathcal{A}}$.

Suppose Condition 3.2(b) holds so that $V(\omega, z) \ge c$ for some $c \in \mathbb{R}$ and all $z \in \mathcal{R}_{\mathcal{A}}$. Then,

$$\begin{aligned} F_{x,(t)}^{\mathcal{A},\beta,0} &= \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} Z_{x,v}^{\beta} e^{\beta h \cdot (v-x)} = \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} \mathbf{E}_{x} \Big[e^{-\beta \sum_{i=0}^{\tau_{v}-1} V(T_{X_{i}}\omega, X_{i+1}-X_{i})} \mathbb{1}_{\{\tau_{v}<\infty\}} \Big] e^{\beta h \cdot (v-x)} \\ &\leq \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} \mathbf{E}_{x} \Big[e^{\beta \ell |c|} \mathbb{1}_{\{\tau_{v}<\infty\}} \Big] e^{\beta h \cdot (v-x)} \leq \frac{1}{\beta} \log \sum_{v \in \tilde{L}_{\mathcal{A},x,t}} e^{\beta \ell |c|+\beta h \cdot (v-x)} \\ &\leq \ell |c| + \frac{1}{\beta} \log C \ell^{d} + \max_{v \in \tilde{L}_{\mathcal{A},x,t}} h \cdot (v-x) \\ (4.20) &\leq \ell |c| + \frac{1}{\beta} \log C \ell^{d} + \ell d \max_{1 \leq i \leq d} |h_{i}| \cdot \max_{\substack{z \in \mathcal{R}, i \\ 1 \leq i \leq d}} |z_{i}|. \end{aligned}$$

The last line follows since v is at most ℓ steps from x. At zero temperature, the equivalent bound is

(4.21)
$$F_{x,(t)}^{\mathcal{A},\infty,0} = \sup_{v \in L_{\mathcal{A},x,t}} (F_{x,v}^{\infty} + h \cdot (v - x)) \leq \sup_{v \in \tilde{L}_{\mathcal{A},x,t}} (\ell|c| + h \cdot (v - x))$$
$$\leq \ell|c| + \ell d \max_{1 \leq i \leq d} |h_i| \cdot \max_{\substack{z \in \mathcal{R}_{\mathcal{A}} \\ 1 \leq i \leq d}} |z_i|.$$

Lastly, suppose Condition 3.2(c) is satisfied. Then in the positive temperature case,

$$F_{x,(t)}^{\mathcal{A},\beta,0} = \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t}} Z_{x,v}^{\beta} e^{\beta h \cdot v} \leqslant \frac{1}{\beta} \log \left(C\ell^d \max_{v \in \tilde{L}_{\mathcal{A},x,t}} Z_{x,v}^{\beta} e^{\beta h \cdot v} \right) \leqslant \max_{v \in \tilde{L}_{\mathcal{A},x,t}} \frac{1}{\beta} \log e^{\beta F_{x,v}^{\infty} + \beta h \cdot v} + \frac{1}{\beta} \log C\ell^d$$

$$(4.22) \qquad \leqslant \max_{(t-x\cdot\hat{u})R^{-1} \leqslant k \leqslant (t+R-x\cdot\hat{u})\delta^{-1}} \max_{v \in \tilde{L}_{\mathcal{A},x,t}} \left(F_{x,v,k}^{\infty} + h \cdot v \right) + \frac{1}{\beta} \log C\ell^d.$$

Similarly, in the zero temperature case,

(4.23)
$$F_{x,(t)}^{\mathcal{A},\infty,0} \leq \max_{(t-x\cdot\hat{u})R^{-1} \leq k \leq (t+R-x\cdot\hat{u})\delta^{-1}} \max_{v \in \tilde{L}_{\mathcal{A},x,t}} (F_{x,v,k}^{\infty} + h \cdot v)$$

The bound now follows since $\mathbb{E}[F_{x,v,k}^{\infty}] < \infty$ for each of the finitely many k and v because $V^{-}(\omega, z) \in L^{1}(\mathbb{P})$ for each $z \in \mathcal{R}_{\mathcal{A}}$.

For each $t > \max\{x \cdot \hat{u}, (x + y) \cdot \hat{u}\}$ and $\varepsilon \ge 0$, the free energy satisfies an approximate shift-covariance property, which is exact if $\varepsilon = 0$. For $\beta \in (0, \infty)$, start by writing

$$F_{x+y,(t)}^{\mathcal{A},\beta,\varepsilon}(\omega) = \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x+y,t}} Z_{x+y,v}^{\beta}(\omega) e^{\beta h \cdot (v-x-y) - \varepsilon |v|_{1}} = \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x+y,t}} Z_{x,v-y}^{\beta}(T_{y}\omega) e^{\beta h \cdot (v-x-y) - \varepsilon |v|_{1}}$$

$$(4.24) \qquad = \frac{1}{\beta} \log \sum_{v \in L_{\mathcal{A},x,t-y,\tilde{u}}} Z_{x,v}^{\beta}(T_{y}\omega) e^{\beta h \cdot (v-x) - \varepsilon |v+y|_{1}}.$$

Using the bounds $-\varepsilon |v|_1 - \varepsilon |y|_1 \leq -\varepsilon |v + y|_1 \leq -\varepsilon |v|_1 + \varepsilon |y|_1$, we see that

(4.25)
$$\left| F_{x+y,(t)}^{\mathcal{A},\beta,\varepsilon}(\omega) - F_{x,(t-(y\cdot\hat{u}))}^{\mathcal{A},\beta,\varepsilon}(T_y\omega) \right| \leq \frac{\varepsilon |y|_1}{\beta}$$

If $\beta = \infty$ and $\varepsilon = 0$, this shift-covariance is exact:

(4.26)
$$F_{x+y,(t)}^{\mathcal{A},\infty,0}(\omega) = \sup_{v \in L_{\mathcal{A},x+y,t}} (F_{x,v-y}^{\infty}(T_{y}\omega) + h \cdot (v-x-y))$$
$$= \sup_{v \in L_{\mathcal{A},x,t-(v,\hat{u})}} (F_{x,v}^{\infty}(T_{y}\omega) + h \cdot (v-x)) = F_{x,(t-(y,\hat{u}))}^{\mathcal{A},\infty,0}(T_{y}\omega).$$

For $\mathcal{A} \in \mathbb{A}$, $(\beta, m) \in \mathcal{D}_{\mathcal{A}}, \varepsilon \ge 0, t \in \mathbb{R}, x \in \mathbb{Z}^d$, and $z \in \mathcal{R}_{\mathcal{A}}$, define

(4.27)
$$B_{t,\varepsilon}^{\mathcal{A},\beta,m}(x,x+z) = F_{x,(t)}^{\mathcal{A},\beta,\varepsilon} - F_{x+z,(t)}^{\mathcal{A},\beta,\varepsilon} - h \cdot z,$$

 $\text{if }t>\max\big\{x\cdot \widehat{u},(x+z)\cdot \widehat{u}:z\in \mathcal{R}_{\mathcal{A}}\big\}\text{ and }B_{t,\varepsilon}^{\mathcal{A},\beta,m}(x,x+z)=0\text{ otherwise}.$

Lemma 4.10. Let $\mathcal{A} \in \mathbb{A}$, $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$, and $\varepsilon \ge 0$. Then \mathbb{P} -almost surely for all $x \in \mathbb{Z}^d$ and $t > \max\{x \cdot \hat{u}, (x+z) \cdot \hat{u} : z \in \mathcal{R}_{\mathcal{A}}\},\$

(4.28)
$$\sum_{z \in \mathcal{R}_{\mathcal{A}}} p(z) \exp\left\{-\beta B_{t,\varepsilon}^{\mathcal{A},\beta,m}(x,x+z) - \beta V(T_x\omega,z)\right\} = 1 \quad if \ \beta < \infty \ and$$

(4.29)
$$\min_{z \in \mathcal{R}_{\mathcal{A}}} \left\{ B_{t,\varepsilon}^{\mathcal{A},\infty,m}(x,x+z) + V(T_x\omega,z) \right\} = 0 \quad if \ \beta = \infty.$$

Proof. The desired equations essentially come from the one-step decomposition of $Z_{x,(t)}^{\mathcal{A},\beta,\varepsilon}$. Take $\mathcal{A},\beta,m,\varepsilon,x$, and t as in the statement. Recall that $L_{\mathcal{A},x,t} = L_{\mathcal{A},x+z,t}$ if $z \in \mathcal{R}_{\mathcal{A}}$. Then

$$= (Z_{x,(t)}^{\mathcal{A},\beta,\varepsilon})^{-1} \sum_{v \in L_{\mathcal{A},x,t}} e^{\beta h \cdot (v-x) - \varepsilon |v|_1} \operatorname{E}_x \left[\exp\left\{ -\beta \sum_{i=0}^{\tau_v - 1} V(T_{X_i}\omega, X_{i+1} - X_i) \right\} \mathbb{1}_{\{\tau_v < \infty\}} \right]$$
$$= (Z_{x,(t)}^{\mathcal{A},\beta,\varepsilon})^{-1} Z_{x,(t)}^{\mathcal{A},\beta,\varepsilon} = 1.$$

The fifth equality above comes from the Markov property. The recovery property (4.28) is proved. For the zero temperature case, write

$$\begin{split} & \min_{z \in \mathcal{R}_{\mathcal{A}}} \left\{ B_{t,\varepsilon}^{\mathcal{A},\infty,m}(x,x+z) + V(T_{x}\omega,z) \right\} \\ &= \min_{z \in \mathcal{R}_{\mathcal{A}}} \left\{ F_{x,(t)}^{\mathcal{A},\infty,0} - F_{x+z,(t)}^{\mathcal{A},\infty,0} - h \cdot z + V(T_{x}\omega,z) \right\} \\ &= \min_{z \in \mathcal{R}_{\mathcal{A}}} \left\{ F_{x,(t)}^{\mathcal{A},\infty,0} - \sup_{v \in L_{\mathcal{A},x+z,t}} \sup_{k \ge 1} \sup_{x_{0:k} \in \overline{\mathbb{X}}_{x,v}^{k}} \left\{ -\sum_{i=0}^{k-1} V(T_{x_{i}}\omega,x_{i+1} - x_{i}) + h \cdot (v - x - z) \right\} - h \cdot z + V(T_{x}\omega,z) \right\} \\ &= F_{x,(t)}^{\mathcal{A},\infty,0} - \max_{z \in \mathcal{R}_{\mathcal{A}}} \sup_{v \in L_{\mathcal{A},x,t}} \sup_{k \ge 2} \sup_{x_{0:k} \in \overline{\mathbb{X}}_{x,v}^{k}} \left\{ -\sum_{i=0}^{k-1} V(T_{x_{i}}\omega,x_{i+1} - x_{i}) + h \cdot (v - x) \right\} \\ &= F_{x,(t)}^{\mathcal{A},\infty,0} - \sup_{v \in L_{\mathcal{A},x,t}} \sup_{k \ge 1} \sup_{x_{0:k} \in \overline{\mathbb{X}}_{x,v}^{k}} \left\{ -\sum_{i=0}^{k-1} V(T_{x_{i}}\omega,x_{i+1} - x_{i}) + h \cdot (v - x) \right\} \\ &= F_{x,(t)}^{\mathcal{A},\infty,0} - F_{x,(t)}^{\mathcal{A},\infty,0} = 0. \end{split}$$

The recovery property (4.29) is proved.

The next lemma gives some information about where the minimum in (4.29) is attained.

Lemma 4.11. Let $\mathcal{A} \in \mathbb{A}$ and let m be such that $(\infty, m) \in \mathcal{D}_{\mathcal{A}}$. Then \mathbb{P} -almost surely, for any finite set $A \subset \mathbb{Z}^d$ and any $t > \max\{x \cdot \hat{u}, (x+z) \cdot \hat{u} : x \in A, z \in \mathcal{R}_{\mathcal{A}}\}$, there exists $y \in A$ and $z \in \mathcal{R}_{\mathcal{A}}$ such that $y + z \notin A$ and $V(T_y\omega, z) + B_{t,0}^{\mathcal{A},\infty,m}(y, y+z) = 0$.

Proof. Take A and t as in the statement, $x \in A$ and $\varepsilon > 0$. Let $v \in L_{\mathcal{A},x,t}$ be such that

$$F_{x,(t)}^{\mathcal{A},\infty,0}(\omega) - \varepsilon \leqslant F_{x,v}^{\infty}(\omega) + h \cdot (v-x) \leqslant F_{x,(t)}^{\mathcal{A},\infty,0}(\omega).$$

Take a path $x_{0:n} = x_{0:n}(\varepsilon) \in \mathbb{X}^n_{x,v}$ such that

$$F_{x,v}^{\infty}(\omega) - \varepsilon \leqslant -\sum_{i=0}^{n-1} V(T_{x_i}\omega, x_{i+1} - x_i) \leqslant F_{x,v}^{\infty}(\omega).$$

Note $v \notin A$ by the definition of t. Let $k = k(\varepsilon) = \min\{i \in [0, n-1] : x_{i+1} \notin A\}, y = y(\varepsilon) = x_k$, and $z = z(\varepsilon) = x_{k+1} - x_k$. Since $F_{x,y}^{\infty}(\omega) + F_{y,v}^{\infty}(\omega) \leqslant F_{x,v}^{\infty}(\omega)$ for all $v \in L_{\mathcal{A},x,t}$, we have $F_{x,y}^{\infty}(\omega) + h \cdot (y - x) + F_{y,(t)}^{\mathcal{A},\infty,0}(\omega) \leqslant F_{x,(t)}^{\mathcal{A},\infty,0}(\omega)$. Thus,

$$F_{x,y}^{\infty}(\omega) + h \cdot (y - x) + F_{y,(t)}^{\mathcal{A},\infty,0}(\omega) - 2\varepsilon \leqslant F_{x,(t)}^{\mathcal{A},\infty,0}(\omega) - 2\varepsilon \leqslant -\sum_{i=0}^{n-1} V(T_{x_i}\omega, x_{i+1} - x_i) + h \cdot (v - x) \\ \leqslant F_{x,y}^{\infty}(\omega) + h \cdot (y - x) - \sum_{i=k}^{n-1} V(T_{x_i}\omega, x_{i+1} - x_i) + h \cdot (v - y),$$

which implies

$$F_{y,(t)}^{\mathcal{A},\infty,0}(\omega) - 2\varepsilon \leqslant -\sum_{i=k}^{n-1} V(T_{x_i}\omega, x_{i+1} - x_i) + h \cdot (v - y)$$

Thus.

$$\begin{split} V(T_{y}\omega,z) + B_{t,0}^{\mathcal{A},\infty,m}(y,y+z) &= V(T_{y}\omega,z) + F_{y,(t)}^{\mathcal{A},\infty,0} - F_{y+z,(t)}^{\mathcal{A},\infty,0} - h \cdot z \\ &\leq V(T_{y}\omega,z) + F_{y,(t)}^{\mathcal{A},\infty,0} - F_{y+z,v}^{\infty}(\omega) - h \cdot (v-y-z) - h \cdot z \\ &\leq F_{y,(t)}^{\mathcal{A},\infty,0} + \sum_{i=k}^{n-1} V(T_{x_{i}}\omega,x_{i+1}-x_{i}) - h \cdot (v-y) \leq 2\varepsilon. \end{split}$$

Since A and $\mathcal{R}_{\mathcal{A}}$ are finite, we can find a subsequence $\varepsilon_j \to 0$ and $y \in A, z \in \mathcal{R}_{\mathcal{A}}$, such that $y(\varepsilon_j) = 0$ $y \in A$ and $z(\varepsilon_j) = z \in \mathcal{R}_A$ for all j. Applying the above with $\varepsilon = \varepsilon_j$ and taking $j \to \infty$ shows that $V(T_y\omega, z) + B_{t,0}^{\mathcal{A},\infty,m}(y, y+z) \leq 0$. The claim follows from this and (4.29). \square

Next, we develop bounds on the limiting free energy. Recall the definition of R_0 in (4.17). If necessary, enlarge the slab width to satisfy $R > 2R_0$.

Lemma 4.12. If $\beta = \infty$ or Condition 3.2(b) or (c) is satisfied, take $\varepsilon = 0$. If $\beta < \infty$ and Condition 3.2(a) holds, take $\varepsilon > 0$. Then, for any a < b,

(4.30)
$$1 - \frac{|\xi|_{1}\varepsilon}{\beta} \leq \lim_{n \to \infty} \frac{1}{n} \mathbb{E}\left[\inf_{a \leq s \leq b} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon}\right] \leq \lim_{n \to \infty} \frac{1}{n} \mathbb{E}\left[\sup_{a \leq s \leq b} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon}\right] \leq 1$$

and, \mathbb{P} -almost surely,

(4.31)
$$1 - \frac{|\xi|_{1}\varepsilon}{\beta} \leq \lim_{n \to \infty} \frac{1}{n} \inf_{a \leq s \leq b} F^{\mathcal{A},\beta,\varepsilon}_{\mathbf{0},(n+s)} \leq \lim_{n \to \infty} \frac{1}{n} \sup_{a \leq s \leq b} F^{\mathcal{A},\beta,\varepsilon}_{\mathbf{0},(n+s)} \leq 1.$$

Proof. It will be convenient in this and the following paragraph to modify our notation to include the width of the slab, writing $F_{\mathbf{0},(n+a,R)}^{\mathcal{A},\beta,\varepsilon}$ when this width is R. With this notation, we see that if R' > R, then $s \leq t \leq t+R \leq s+R'$ and $F_{\mathbf{0},(t,R)}^{\mathcal{A},\beta,\varepsilon} \leq F_{\mathbf{0},(s,R')}^{\mathcal{A},\beta,\varepsilon}$. Thus, for any $s \in [a,b]$, $n+a \leq n+s \leq n+s+R \leq n+a+(R+b-a)$ and, therefore, $\sup_{a \leq s \leq b} F_{\mathbf{0},(n+s,R)}^{\mathcal{A},\beta,\varepsilon} \leq F_{\mathbf{0},(n+a,R+b-a)}^{\mathcal{A},\beta,\varepsilon}$. Similarly, for any $s \in [a,b]$, take $j = \lfloor 2R^{-1}(s-a) \rfloor$, which satisfies $n+s \leq n+a+jR/2 \leq n+a+(j+1)R/2 \leq 1 \leq n+a+(j+1)R/2 < n+a+(j+$

n + s + R and $0 \le i \le 2R^{-1}(b - a) + 1$. Therefore,

$$\inf_{a\leqslant s\leqslant b} F_{\mathbf{0},(n+s,R)}^{\mathcal{A},\beta,\varepsilon} \ge \min_{j\in [0,2R^{-1}(b-a)+1]\cap \mathbb{Z}} F_{\mathbf{0},(n+a+jR/2,R/2)}^{\mathcal{A},\beta,\varepsilon}.$$

The upshot is that, modulo increasing or decreasing the size of R, it is enough to prove the claims of the lemma for a fixed $s \in \mathbb{R}$, without taking suprema or infima over $s \in [a, b]$. In the rest, we will go back to omitting the width of the slab from the free energy notation.

Fix $s \in \mathbb{R}$. We will show that for each $s \in \mathbb{R}$, we have, \mathbb{P} -almost surely,

(4.32)
$$1 - \frac{|\xi|_{1}\varepsilon}{\beta} \leq \lim_{n \to \infty} \frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon} \leq \lim_{n \to \infty} \frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon} \leq 1.$$

This then proves (4.31), as explained in the above two paragraphs.

We start with the lower bound. Write $\xi = \sum_{z \in \mathcal{R}_{\mathcal{A}}} b_z z$ with coefficients $b_z \ge 0$. For $n \in \mathbb{Z}_{\ge 0}$ let $b_{n,z} =$ $[(n+s)b_z]$ if $z \cdot \hat{u} \ge 0$ and $b_{n,z} = \lfloor (n+s)b_z \rfloor$ otherwise. Let $v_n = \sum_{z \in \mathcal{R}_A} b_{n,z} z$. Note that $v_n/n \to \xi$ as $n \to \infty$. Since $\xi \cdot \hat{u} = 1$,

$$n+s = (n+s)\sum_{z \in \mathcal{R}_{\mathcal{A}}} b_z z \cdot \hat{u} \leqslant v_n \cdot \hat{u} \leqslant (n+s)\sum_{z \in \mathcal{R}_{\mathcal{A}}} b_z z \cdot \hat{u} + \sum_{z \in \mathcal{R}_{\mathcal{A}}} |z \cdot \hat{u}| < n+s+R.$$

Thus, $v_n \in L_{\mathcal{A},n+s}$ for all n and

$$\frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon} = \frac{1}{n\beta} \log \sum_{v \in L_{\mathcal{A},n+s}} Z_{\mathbf{0},v}^{\beta} e^{\beta h \cdot v - \varepsilon |v|_1} \ge \frac{1}{n\beta} \log Z_{\mathbf{0},v_n}^{\beta} + \frac{\beta h \cdot v_n - |v_n|_1 \varepsilon}{n\beta}.$$

Since $\hat{u} - h \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi)$ and $\xi \cdot \hat{u} = 1$, by Lemma A.1, $\Lambda^{\beta}(\xi) = \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi) = (\hat{u} - h) \cdot \xi = 1 - h \cdot \xi$. Taking $n \to \infty$ and applying (2.8), we obtain, \mathbb{P} -almost surely,

$$\lim_{n \to \infty} \frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon} \ge \Lambda^{\beta}(\xi) + h \cdot \xi - \frac{|\xi|_{1}\varepsilon}{\beta} = 1 - \frac{|\xi|_{1}\varepsilon}{\beta}.$$

The equivalent result at zero temperature is

$$\lim_{n \to \infty} \frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\infty,0} \ge \lim_{n \to \infty} \frac{1}{n} (F_{\mathbf{0},v_n}^{\infty} + h \cdot v_n) = \Lambda^{\infty}(\xi) + h \cdot \xi = 1.$$

Next, we derive the almost sure upper bound. Recall that we are now working with a fixed s. The upper bound (2.7) implies that, with \mathbb{P} -probability one, there exists $n_0 = n_0(\omega, \varepsilon)$ such that whenever $|v|_1 \cdot |\hat{u}|_{\infty} \ge n_0$,

$$\frac{1}{\beta} \log Z_{\mathbf{0},v}^{\beta} \leqslant \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(v) + \frac{\varepsilon}{2\beta} |v|_{1}.$$

Also note that since $\hat{u} - h \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi)$, for $v \in L_{\mathcal{A}, n+s}$

$$\Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(v) \leq \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(\xi) + (v-\xi) \cdot (\widehat{u}-h) = v \cdot (\widehat{u}-h) \leq n+s+R-v \cdot h.$$

For $n \ge n_0$,

$$\begin{split} \frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon} &= \frac{1}{n\beta} \log \sum_{v \in L_{\mathcal{A},n+s}} Z_{\mathbf{0},v}^{\beta} e^{\beta h \cdot v - \varepsilon |v|_{1}} \\ &\leqslant \frac{1}{n\beta} \log \sum_{v \in L_{\mathcal{A},n+s}} e^{\beta \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(v) + \beta h \cdot v - \varepsilon |v|_{1}/2} \\ &\leqslant \frac{1}{n\beta} \log \sum_{v \in L_{\mathcal{A},n+s}} e^{\beta(n+s+R-v \cdot h) + \beta h \cdot v - \varepsilon |v|_{1}/2} \\ &\leqslant \frac{n+s+R}{n} + \frac{1}{n\beta} \log \sum_{v \in \mathbb{Z}^{d}} e^{-\varepsilon |v|_{1}/2} \\ &\leqslant \frac{n+s+R}{n} + \frac{1}{n\beta} \log \left(C \sum_{k=0}^{\infty} k^{d} e^{-\varepsilon k/2}\right) \xrightarrow[n \to \infty]{} 1. \end{split}$$

The equivalent bound at zero temperature is

$$\frac{1}{n}F_{\mathbf{0},(n+s)}^{\mathcal{A},\infty,0} = \frac{1}{n}\sup_{v\in L_{\mathcal{A},n+s}}(F_{\mathbf{0},v}^{\infty} + h \cdot v) \leqslant \frac{1}{n}\sup_{v\in L_{\mathcal{A},n+s}}(\Lambda_{\mathcal{A}}^{\infty,\mathrm{usc}}(v) + 1 + h \cdot v) \leqslant \frac{1}{n}(n+s+R+1)\xrightarrow[n\to\infty]{}1.$$

Now we turn to proving (4.30). As explained at the beginning of the proof, it is enough to prove that for each $s \in \mathbb{R}$,

(4.33)
$$1 - \frac{|\xi|_{1}\varepsilon}{\beta} \leqslant \lim_{n \to \infty} \frac{1}{n} \mathbb{E}[F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon}] \leqslant \lim_{n \to \infty} \frac{1}{n} \mathbb{E}[F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon}] \leqslant 1.$$

We first prove the upper bound. For this, we consider the three cases in Condition 3.2.

If Condition 3.2(a) or (b) is satisfied, then the upper bound in (4.33) follows from the bounds (4.18-4.21), the upper bound in (4.32), and Fatou's Lemma.

Suppose now that Condition 3.2(c) holds. Without any loss of generality, we can assume $r_0 > R$. Let $f(\omega) = \max_{z \in \mathcal{R}_A} V^-(\omega, z) + \max_{z \in \mathcal{R}_A} |h \cdot z|$. Then, with the notation $\ell = \ell(n + s)$ as introduced above (4.19), the bounds in (4.22) and (4.23) give

$$(4.34) \qquad \frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,0} \leqslant \frac{1}{n} \max_{(n+s)R^{-1} \leqslant k \leqslant (n+s+R)\delta^{-1}} \max_{v \in \tilde{L}_{\mathcal{A},\mathbf{0},n+s}} (F_{\mathbf{0},v,k}^{\infty} + h \cdot v) + \frac{1}{n\beta} \log C\ell^{d}$$

$$(4.34) \qquad \leqslant \frac{1}{n} \max_{x_{0:\ell} \in \mathbb{X}_{\mathbf{0}}^{\ell}(\mathcal{R}_{\mathcal{A}})} \sum_{k=0}^{\ell-1} f(T_{x_{k}}\omega) + \frac{1}{n\beta} \log C\ell^{d},$$

where it is understood that in the zero temperature case, the last term on the right-hand side is 0. In the second inequality, we used the fact that f is non-negative, and hence the sum of shifts of f does not get smaller if we add more terms, and the maximum does not get smaller if we drop the requirement that the endpoint $v \in L_{A,n+s}$. We will bound the first term on the right-hand side using lattice animal bounds.

A subset S of \mathbb{Z}^d is said to be connected if, for any two points in S, there exists a nearest-neighbor path in S that connects them. Given $n \in \mathbb{Z}_{>0}$, a connected subset of \mathbb{Z}^d that has cardinality n is called a lattice animal of size n. Let S_n denote the set of lattice animals of size n. Let $B = \{0, 1, 2, \ldots, r_0 - 1\}^d$. Then $\{r_0y + B : y \in \mathbb{Z}^d\}$ is a disjoint tiling of \mathbb{Z}^d . Consider a path $x_{0:\ell} \in \mathbb{X}_0^{\ell}(\mathcal{R}_A)$. Let $y_0, \ldots, y_\ell \in \mathbb{Z}^d$ and $u_0, \ldots, u_\ell \in B$ be (the unique points) such that $x_k = r_0 y_k + u_k$ for each $0 \leq k \leq \ell$. It must be that $|y_{k+1} - y_k|_{\infty} \leq 1$, since $r_0 > R$. Although y_k and y_{k+1} might not be nearest neighbors, we can fill in with at most d-1 intermediate points to obtain a nearest-neighbor sequence. In total, we start with $\ell + 1$ points then add at most $\ell(d-1)$ more. The resulting sequence may have repetitions and fewer than $\ell d + 1$ points, but it will be contained in a lattice animal $S \in \mathcal{S}_{\ell d+1}$. Since the path $x_{0:\ell}$ does not repeat points, the pairs (y_k, u_k) are distinct. Therefore,

$$\sum_{k=0}^{\ell-1} f(T_{x_k}\omega) = \sum_{k=0}^{\ell-1} f(T_{r_0y_k+u_k}\omega) \leqslant \sum_{u\in B} \max_{S\in\mathcal{S}_{\ell d+1}} \sum_{y\in S} f(T_{r_0y+u}\omega).$$

We have thus shown that

$$\frac{1}{n} \max_{x_{0:\ell} \in \mathbb{X}_{0}^{\ell}(\mathcal{R}_{\mathcal{A}})} \sum_{k=0}^{\ell-1} f(T_{x_{k}}\omega) \leqslant \frac{\ell}{n} \cdot \frac{1}{\ell} \sum_{u \in B} \max_{S \in \mathcal{S}_{\ell d+1}} \sum_{y \in S} f(T_{r_{0}y+u}\omega).$$

Since $\{f(T_{r_0y+u}\omega) : y \in \mathbb{Z}^d\}$ are i.i.d. and $f \in L^q(\mathbb{P})$ for some q > d, [52, Theorem 1.1] implies that the right-hand side converges both almost-surely and in $L^1(\mathbb{P})$. This, the upper bound (4.34), the upper bound in (4.32), and Fatou's Lemma imply the upper bound in (4.33) and, therefore, the upper bound in (4.30) is proved.

We finish the proof by arguing for the lower bound in (4.33). We work out the case of a positive temperature, with the case of zero temperature being similar. Observe that for any $n \in \mathbb{Z}_{>0}$ and $u, v \in \mathbb{Z}^d$ such that $v - u \in \mathcal{G}_A \cap D_n$,

$$Z_{u,v,n}^{\beta}(\omega) = \mathbf{E}_{u} \bigg[e^{-\beta \sum_{k=0}^{n-1} V(T_{X_{k}}\omega, X_{k+1} - X_{k})} \mathbb{1}_{\{X_{n} = v\}} \bigg] \leq \mathbf{E}_{u} \bigg[e^{-\beta \sum_{k=0}^{\tau_{v} - 1} V(T_{X_{k}}\omega, X_{k+1} - X_{k})} \mathbb{1}_{\{\tau_{v} < \infty\}} \bigg]$$
$$\leq \mathbf{E}_{u} \bigg[e^{-\beta \sum_{k=0}^{\tau_{v} - 1} V(T_{X_{k}}\omega, X_{k+1} - X_{k})} \mathbb{1}_{\{\tau_{v} < \infty\}} \bigg] = Z_{u,v}^{\beta}(\omega).$$

To justify the first inequality note that if $\mathbf{0} \notin \mathcal{U}$, then $X_n = v$ implies $\tau_v = n$ while if $\mathbf{0} \in \mathcal{U}$, then Condition 3.2(a) must hold, in which case $X_n = v$ implies $V(T_{X_k}\omega, X_{k+1} - X_k) \ge 0$ for all $k \le n$ and, therefore, $e^{-\beta \sum_{k=\tau_v}^{n-1} V(T_{X_k}\omega, X_{k+1} - X_k)} \le 1$.

Take any t > 0 such that $\xi/t \in \mathcal{U}_{\mathcal{A}}$. Let $\hat{x}_n(\xi/t)$ be the path defined in [58, (2.1)]. Note that this path has increments in $\mathcal{R}_{\mathcal{A}}$ and satisfies $\hat{x}_n(\xi/t) \in \mathcal{G}_{\mathcal{A}} \cap D_n$ for all $n \in \mathbb{Z}_{>0}$ and $\hat{x}_n(\xi/t)/n \to \xi/t$ as $n \to \infty$. This and $\xi \cdot \hat{u} = 1$ imply that there exists a subsequence m_n such that $\hat{x}_{m_n}(\xi/t) \in L_{\mathcal{A},n+s}$ for all n large enough. Thus, for n large enough,

$$\begin{aligned} \frac{1}{n} F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon} &= \frac{1}{n\beta} \log \sum_{v \in L_{\mathcal{A},n+s}} Z_{\mathbf{0},v}^{\beta} e^{\beta h \cdot v - \varepsilon |v|_1} \geqslant \frac{1}{n\beta} \log Z_{\mathbf{0},\hat{x}_{m_n}(\xi/t)}^{\beta} + \frac{\beta h \cdot \hat{x}_{m_n}(\xi/t) - |\hat{x}_{m_n}(\xi/t)|_1 \varepsilon}{n\beta} \\ &\geqslant \frac{1}{n\beta} \log Z_{\mathbf{0},\hat{x}_{m_n}(\xi/t),m_n}^{\beta} + \frac{\beta h \cdot \hat{x}_{m_n}(\xi/t) - |\hat{x}_{m_n}(\xi/t)|_1 \varepsilon}{n\beta}. \end{aligned}$$

By [58, Theorem 2.2] (and [26, Theorem 2.4] for the zero temperature case), $m^{-1}F^{\beta}_{\mathbf{0},\hat{x}_m(\xi/t),m}$ converges both almost surely and in $L^1(\mathbb{P})$, as $m \to \infty$. Furthermore, dividing

$$n + s \leq \hat{x}_{m_n}(\xi/t) \cdot \hat{u} < n + s + R$$

by m_n and taking $n \to \infty$ implies that $m_n/n \to t$ as $n \to \infty$. Thus, $(n\beta)^{-1} F^{\beta}_{\mathbf{0},\hat{x}_{m_n}(\xi/t),m_n}$ converges both almost surely and in $L^1(\mathbb{P})$, as $n \to \infty$. This allows us to apply Fatou's Lemma and deduce the lower bound in (4.33) from the one in (4.32), completing the proof of the lemma.

For each $n \in \mathbb{Z}_{>0}$, let $(U_n)_{n \in \mathbb{Z}_{>0}}$ be a sequence of independent random variables, independent of everything else, and such that for each n, U_n is uniformly distributed on [0, n]. Denote the distribution of this sequence by \mathbb{U} and let $\overline{\mathbb{P}} = \mathbb{P} \otimes \mathbb{U}$ with expectation $\overline{\mathbb{E}}$. Define $\mathbf{P}_{n,\varepsilon}$ to be the distribution of

$$\left(\omega, \left\{B_{U_n,\varepsilon}^{\mathcal{A},\beta,m}(x,x+z) : \mathcal{A} \in \mathbb{A}, (\beta,m) \in \mathcal{D}_{\mathcal{A}}, x \in \mathbb{Z}^d, z \in \mathcal{R}_{\mathcal{A}}\right\}\right)$$

induced by $\overline{\mathbb{P}}$ on $(\widehat{\Omega}, \widehat{\mathfrak{S}})$.

Lemma 4.13. If $\beta = \infty$ or Condition 3.2(b) or (c) is satisfied, take $\varepsilon = 0$. If $\beta < \infty$ and Condition 3.2(a) holds, take $\varepsilon > 0$. Then, the family $\{\mathbf{P}_{n,\varepsilon} : n \in \mathbb{Z}_{>0}\}$ is tight.

Proof. Take $x \in \mathbb{Z}^d$ and $z \in \mathcal{R}_A$. Let $t_0 = \max\{0, x \cdot \hat{u}, (x+z) \cdot \hat{u} : z \in \mathcal{R}_A\}$. Recall that $B_{t,\varepsilon}^{\mathcal{A},\beta,m}(x, x+z) = 0$ whenever $0 < t \leq t_0$. Then for $n > t_0$, use the approximate shift-covariance properties (4.25) and (4.26) to obtain

$$\begin{split} \overline{\mathbb{E}} \Big[B_{U_n,\varepsilon}^{\mathcal{A},\beta,m}(x,x+z) \Big] \\ &= \frac{1}{n} \int_{t_0}^n \mathbb{E} \Big[B_{s,\varepsilon}^{\mathcal{A},\beta,m}(x,x+z) \Big] \, ds \\ &= \frac{1}{n} \int_{t_0}^n \mathbb{E} \Big[F_{x,(s)}^{\mathcal{A},\beta,\varepsilon} - F_{x+z,(s)}^{\mathcal{A},\beta,\varepsilon} - h \cdot z \Big] \, ds \\ &\leq \frac{1}{n} \int_{t_0}^n \mathbb{E} \Big[F_{\mathbf{0},(s-x\cdot\hat{u})}^{\mathcal{A},\beta,\varepsilon} - F_{\mathbf{0},(s-(x+z)\cdot\hat{u})}^{\mathcal{A},\beta,\varepsilon} \Big] \, ds - \frac{(n-t_0)h \cdot z}{n} + \frac{\varepsilon}{\beta} (|x|_1 + |x+z|_1) \\ &= \frac{1}{n} \int_{t_0}^n \mathbb{E} \Big[F_{\mathbf{0},(s-x\cdot\hat{u})}^{\mathcal{A},\beta,\varepsilon} \Big] \, ds - \frac{1}{n} \int_{t_0-z\cdot\hat{u}}^{n-z\cdot\hat{u}} \mathbb{E} \Big[F_{\mathbf{0},(s-x\cdot\hat{u})}^{\mathcal{A},\beta,\varepsilon} \Big] \, ds - \frac{(n-t_0)h \cdot z}{n} + \frac{\varepsilon}{\beta} (|x|_1 + |x+z|_1) \\ &= \frac{1}{n} \int_{n-z\cdot\hat{u}}^n \mathbb{E} \Big[F_{\mathbf{0},(s-x\cdot\hat{u})}^{\mathcal{A},\beta,\varepsilon} \Big] \, ds - \frac{1}{n} \int_{t_0-z\cdot\hat{u}}^{t_0} \mathbb{E} \Big[F_{\mathbf{0},(s-x\cdot\hat{u})}^{\mathcal{A},\beta,\varepsilon} \Big] \, ds - \frac{(n-t_0)h \cdot z}{n} + \frac{\varepsilon}{\beta} (|x|_1 + |x+z|_1) \\ &= \frac{1}{n} \int_{-(x+z)\cdot\hat{u}}^{-x\cdot\hat{u}} \mathbb{E} \Big[F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon} \Big] \, ds - \frac{1}{n} \int_{t_0-(x+z)\cdot\hat{u}}^{t_0-x\cdot\hat{u}} \mathbb{E} \Big[F_{\mathbf{0},(s)}^{\mathcal{A},\beta,\varepsilon} \Big] \, ds - \frac{(n-t_0)h \cdot z}{n} + \frac{\varepsilon}{\beta} (|x|_1 + |x+z|_1) . \end{split}$$

In the above computation, we use the convention that if b < a, then $\int_a^b = -\int_b^a$. When $\beta = \infty$, we take $\varepsilon = 0$, and the last term is understood to be zero.

On the event $\{U_n > t_0\}$, (4.28) and (4.29) imply that

$$(4.36) \qquad B_{U_n,\varepsilon}^{\mathcal{A},\beta,m}(x,x+z) \ge -V(T_x\omega,z) + \beta^{-1}\log p(z) \ge -V^+(T_x\omega,z) + \beta^{-1}\log p(z).$$

Again when $\beta = \infty$, the last term is understood to be 0. Then,

$$(4.37) \qquad \begin{split} \overline{\mathbb{E}}\left[\left|B_{U_{n},\varepsilon}^{\mathcal{A},\beta,m}(x,x+z)\right|\right] &= \overline{\mathbb{E}}\left[\left|B_{U_{n},\varepsilon}^{\mathcal{A},\beta,m}(x,x+z)\right|\mathbb{1}\left\{U_{n} > t_{0}\right\}\right] \\ &= \overline{\mathbb{E}}\left[B_{U_{n},\varepsilon}^{\mathcal{A},\beta,m}(x,x+z)\mathbb{1}\left\{U_{n} > t_{0}\right\}\right] - 2\overline{\mathbb{E}}\left[\min\left\{0, B_{U_{n},\varepsilon}^{\mathcal{A},\beta,m}(x,x+z)\right\}\mathbb{1}\left\{U_{n} > t_{0}\right\}\right] \\ &= \overline{\mathbb{E}}\left[B_{U_{n},\varepsilon}^{\mathcal{A},\beta,m}(x,x+z)\right] - 2\overline{\mathbb{E}}\left[\min\left\{0, B_{U_{n},\varepsilon}^{\mathcal{A},\beta,m}(x,x+z)\right\}\mathbb{1}\left\{U_{n} > t_{0}\right\}\right] \\ &\leq \frac{1}{n} \int_{-(x+z)\cdot\hat{u}}^{-x\cdot\hat{u}} \mathbb{E}\left[F_{\mathbf{0},(n+s)}^{\mathcal{A},\beta,\varepsilon}\right] ds - \frac{1}{n} \int_{t_{0}-(x+z)\cdot\hat{u}}^{t_{0}-x\cdot\hat{u}} \mathbb{E}\left[F_{\mathbf{0},(s)}^{\mathcal{A},\beta,\varepsilon}\right] ds \\ &- \frac{(n-t_{0})h\cdot z}{n} + \frac{\varepsilon}{\beta}(|x|_{1}+|x+z|_{1}) + 2\mathbb{E}[V^{+}(\omega,z)] - 2\beta^{-1}\log p(z). \end{split}$$

All the terms on the right-hand side except the first are trivially uniformly bounded in n. The first term on the right-hand side is also bounded in n by (4.33). This proves the claimed tightness.

If $\beta = \infty$ or Condition 3.2(b) or (c) is satisfied, then let \mathbf{P}_0 (with expectation \mathbf{E}_0) be any weak subsequential limit point of $\mathbf{P}_{n,0}$. Otherwise, for each $0 < \varepsilon \leq 1$, let \mathbf{P}_{ε} (with expectation \mathbf{E}_{ε}) denote any weak subsequential limit point in n of $\mathbf{P}_{n,\varepsilon}$. Recall that $B^{\mathcal{A},\beta,m}(x,x+z)$ is the $(\mathcal{A},\beta,m,x,z)$ -coordinate of $\hat{\omega} \in \hat{\Omega}$. By (4.36) we have that $\mathbf{P}_{n,\varepsilon}$ -almost surely, for all $\mathcal{A} \in \mathbb{A}$, $(\beta,m) \in \mathcal{D}_{\mathcal{A}}$, $x \in \mathbb{Z}^d$, and $z \in \mathcal{R}_{\mathcal{A}}$, $B^{\mathcal{A},\beta,m}(x,x+z) \geq -|V(T_x\omega,z)| + \beta^{-1} \log p(z)$.

Use (4.37), Fatou's Lemma, and the upper bound in (4.30) to obtain

$$\mathbf{E}_{\varepsilon} \left[\left| B^{\mathcal{A},\beta,m}(x,x+z) \right| \right] \leq \lim_{n \to \infty} \frac{1}{n} \left(\int_{-(x+z)\cdot \hat{u}}^{-x\cdot \hat{u}} \mathbb{E} \left[F^{\mathcal{A},\beta,\varepsilon}_{\mathbf{0},(n+s)} \right] ds - \frac{1}{n} \int_{t_0 - (x+z)\cdot \hat{u}}^{t_0 - x\cdot \hat{u}} \mathbb{E} \left[F^{\mathcal{A},\beta,\varepsilon}_{\mathbf{0},(s)} \right] ds - \frac{(n-t_0)h\cdot z}{n} \right) \\ + \frac{\varepsilon}{\beta} (|x|_1 + |x+z|_1) + 2\mathbb{E} [|V(\omega,z)|] - 2\beta^{-1} \log p(z)$$

$$\leq m \cdot z + \frac{1}{\beta} (|x|_1 + |x+z|_1) + 2\mathbb{E} [|V(\omega,z)|] - 2\beta^{-1} \log p(z).$$

$$(4.38) \qquad \leq m \cdot z + \frac{1}{\beta} (|x|_1 + |x+z|_1) + 2\mathbb{E} [|V(\omega,z)|] - 2\beta^{-1} \log p(z).$$

The family $\{\mathbf{P}_{\varepsilon} : 0 < \varepsilon \leq 1\}$ is therefore tight. Similarly, using Fatou's Lemma with (4.35) and (4.30) gives

(4.39)
$$\mathbf{E}_{\varepsilon} \Big[B^{\mathcal{A},\beta,m}(x,x+z) \Big] \leqslant \widehat{u} \cdot z - h \cdot z + \frac{\varepsilon}{\beta} (|x|_1 + |x+z|_1) = m \cdot z + \frac{\varepsilon}{\beta} (|x|_1 + |x+z|_1).$$

Let $\widehat{\mathbb{P}}$ denote any weak subsequential limit point of \mathbf{P}_{ε} as $\varepsilon \to 0$ in (0,1]. If $\beta = \infty$ or Condition 3.2(b) or (c) is satisfied, then take $\widehat{\mathbb{P}} = \mathbf{P}_0$.

Lemma 4.14. $\widehat{\mathbb{P}}$ is \widehat{T} -invariant.

Proof. Let $x, y \in \mathbb{Z}^d$, and $z \in \mathcal{R}_A$. Using (4.25) we get

(4.40)
$$\left| B_{t,\varepsilon}^{\mathcal{A},\beta,m}(x+y,x+y+z,T_{-y}\omega) - B_{t-y\cdot\hat{u},\varepsilon}^{\mathcal{A},\beta,m}(x,x+z,\omega) \right| \leq \frac{2\varepsilon|y|_1}{\beta}$$

Fix $y \in \mathbb{Z}^d$. Let $\alpha \ge 0$, $\ell \in \mathbb{Z}_{>0}$, and take f to be a bounded, α -Lipschitz continuous function of any ℓ coordinates of $\hat{\omega}$. These coordinates can be faces \mathcal{A} , vectors m, inverse temperatures β , vertices x, or increments z. Let t_0 be large enough so that $t_0 > \max\{y \cdot \hat{u}, x \cdot \hat{u}, (x+z) \cdot \hat{u}, (x+y) \cdot \hat{u}, (x+y+z) \cdot \hat{u} : z \in \mathcal{R}_{\mathcal{A}}\}$, where the maximum goes over all of the at most ℓ values of the $x \in \mathbb{Z}^d$ coordinates on which f depends. Let $(n_{\varepsilon,k})_{k \in \mathbb{Z}_{>0}}$ be the subsequence along which $\mathbf{P}_{n,\varepsilon}$ converges weakly to \mathbf{P}_{ε} . Let $\varepsilon_j \to 0$ be the sequence along which $\mathbf{P}_{\varepsilon_i}$ converges weakly to $\hat{\mathcal{P}}$. Then using continuity of \hat{T}_y , we have

$$\begin{split} \widehat{\mathbb{E}}[f \circ \widehat{T}_{y}] &= \lim_{j \to \infty} \mathbf{E}_{\varepsilon_{j}}[f \circ \widehat{T}_{y}] = \lim_{j \to \infty} \lim_{k \to \infty} \mathbf{E}_{n_{k},\varepsilon_{j}}[f \circ \widehat{T}_{y}] \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \mathbf{E}_{n_{k},\varepsilon_{j}}[f(T_{y}\omega, \{B^{\mathcal{A},\beta,m}(x+y,x+y+z)\})] \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \left(\mathcal{O}(n_{k}^{-1}) + \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(T_{y}\omega, \{B^{\mathcal{A},\beta,m}(x+y,x+y+z,\omega)\})] \, ds\right) \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x+y,x+y+z,T_{-y}\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \left\{ \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s-y\cdot\widehat{u},\varepsilon_{j}}(x,x+z,\omega)\})] + \alpha \ell \mathcal{O}(\beta^{-1}\varepsilon_{j}|y|_{1}) \right\} \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}-y\cdot\widehat{u}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega, \{B^{\mathcal{A},\beta,m}_{s,\varepsilon_{j}}(x,x+z,\omega)\})] \, ds \\ &= \lim_{j \to \infty} \lim_{k \to \infty} \frac{1}{n_{k}} \int_{t_{0}}^{n_{k}} \mathbb{E}[f(\omega,$$

On the fourth line, we used the *T*-invariance of \mathbb{P} . On the fifth line, we used (4.40) and that f is an α -Lipschitz continuous function of ℓ coordinates of $\hat{\omega}$. The first equality on the last line follows because the integrals differ by at most $2|y \cdot \hat{u}| \sup |f|$.

We are now ready to prove the main theorem of this section.

Proof of Theorem 4.5. For each $n \in \mathbb{Z}_{>0}$, $\varepsilon \ge 0$, and $A \in \mathfrak{S}$, $\mathbf{P}_{n,\varepsilon}(\pi_{\Omega}(\widehat{\omega}) \in A) = \mathbb{P}(\omega \in A)$. Since $\pi_{\Omega}(\widehat{\omega})$ is continuous, taking $n \to \infty$ and $\varepsilon \to 0$ shows that $\widehat{\mathbb{P}}$ too has this property, for all closed sets A and hence also for all $A \in \mathfrak{S}$. Part (a) is proved.

Take $x, y \in \mathbb{Z}^d$ with $y - x \in \mathcal{G}^+_{\mathcal{A}}$. Consider a path $x_{0:k} \in \mathbb{X}_{x,y}$ of some length $k \in \mathbb{Z}_{\geq 0}$. For any t large enough that $t > x_i \cdot \hat{u}$ for all $0 \leq i \leq k$,

$$\sum_{i=0}^{k-1} B_{t,\varepsilon}^{\mathcal{A},\beta,m}(x_i,x_{i+1}) = \sum_{i=0}^{k-1} \left(F_{x_i,(t)}^{\mathcal{A},\beta,\varepsilon} - F_{x_{i+1},(t)}^{\mathcal{A},\beta,\varepsilon} - h \cdot (x_{i+1} - x_i) \right) = F_{x,(t)}^{\mathcal{A},\beta,\varepsilon} - F_{y,(t)}^{\mathcal{A},\beta,\varepsilon} - h \cdot (y - x).$$

Thus, for any pair of paths $x_{0:k'} \in \mathbb{X}_{x,y}$, $\overline{\mathbb{P}}$ -almost surely, on the event

$$\left\{U_n > c'' = \max\{x_i \cdot \hat{u}, x'_j \cdot \hat{u} : 0 \le i \le k, 0 \le j \le k'\}\right\},\$$

we have

$$\sum_{i=0}^{k-1} B_{U_n,\varepsilon}^{\mathcal{A},\beta,m}(x_i, x_{i+1}) = \sum_{i=0}^{k'-1} B_{U_n,\varepsilon}^{\mathcal{A},\beta,m}(x'_i, x'_{i+1}).$$

From this, we get

$$\mathbf{P}_{n,\varepsilon}\left\{\sum_{i=0}^{k-1} B^{\mathcal{A},\beta,m}(x_i,x_{i+1}) = \sum_{i=0}^{k'-1} B^{\mathcal{A},\beta,m}(x'_i,x'_{i+1})\right\} \ge \overline{\mathbb{P}}\{U_n > c''\} \underset{n \to \infty}{\longrightarrow} 1.$$

Since the event on the left-hand side is closed and $\widehat{\mathbb{P}}$ is a weak limit point of $\mathbf{P}_{n,\varepsilon}$, we conclude that the event has $\widehat{\mathbb{P}}$ -probability one. Similarly, for any $n \in \mathbb{Z}_{>0}$ and $\varepsilon \ge 0$, $B_{U_n,\varepsilon}^{\mathcal{A},\beta,m}(x,x) = 0$, $\overline{\mathbb{P}}$ -almost surely and hence $B^{\mathcal{A},\beta,m}(x,x) = 0$, $\widehat{\mathbb{P}}$ -almost surely, for any $x \in \mathbb{Z}^d$. Consequently, there is a \widehat{T} -invariant event $\widehat{\Omega}_0$ with $\widehat{\mathbb{P}}(\widehat{\Omega}_0) = 1$ such that for all $\widehat{\omega} \in \widehat{\Omega}_0$, $\mathcal{A} \in \mathbb{A}$, $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$, and $x, y \in \mathbb{Z}^d$ with $y - x \in \mathcal{G}_{\mathcal{A}}^+$, $B^{\mathcal{A},\beta,m}(x, x, \widehat{\omega}) = 0$ and

(4.41)
$$B^{\mathcal{A},\beta,m}(x,y,\widehat{\omega}) = \sum_{i=0}^{k-1} B^{\mathcal{A},\beta,m}(x_i,x_{i+1},\widehat{\omega})$$

has the same value for all paths $x_{0:k} \in \mathbb{X}_{x,y}$. For each $z \in \mathcal{R}_{\mathcal{A}}$ and $x \in \mathbb{Z}^d$, set $B^{\mathcal{A},\beta,m}(x,x-z,\hat{\omega}) = -B^{\mathcal{A},\beta,m}(x-z,x,\hat{\omega})$. This definition is consistent with the one above when it also happens to be that $-z \in \mathcal{G}_{\mathcal{A}}^+$. With this definition we have that the right-hand side of (4.41) has the same value for all paths $x_{0:k}$ with $k \in \mathbb{Z}_{\geq 0}$, $x_0 = x$, $x_k = y$, and $x_{i+1} - x_i \in \mathcal{R}_A \cup (-\mathcal{R}_A)$ for $0 \leq i \leq k-1$. Therefore, one can use the right-hand side of (4.41) to define $B^{\mathcal{A},\beta,m}(x,y,\hat{\omega})$ for all $x, y \in \mathbb{Z}^d$ with $y - x \in \mathcal{G}_A$ and $\hat{\omega} \in \hat{\Omega}_0$. With this definition, we have the cocycle property (4.10), for all $\widehat{\omega} \in \widehat{\Omega}_0, (\beta, m) \in \mathcal{D}_{\mathcal{A}}, \text{ and } u, x, y \in \mathbb{Z}^d \text{ such that } x - u, y - x \in \mathcal{G}_{\mathcal{A}}.$

The covariance property (4.9) holds, for all $\hat{\omega} \in \hat{\Omega}$ and $v, x, y \in \mathbb{Z}^d$ with $y - x \in \mathcal{R}_A$ simply by the definition of the shift \hat{T}_z . Then the definition (4.41) ensures that this holds for all $\hat{\omega} \in \hat{\Omega}_0$ and $v, x, y \in \mathbb{Z}^d$ such that $y - x \in \mathcal{G}_{\mathcal{A}}$, without the restriction $y - x \in \mathcal{R}_{\mathcal{A}}$.

Using a similar argument as above one deduces from (4.28) and (4.29) and passing $n \to \infty$ and $\varepsilon \to 0$ that there exists a \hat{T} -invariant event $\hat{\Omega}_{coc} \subset \hat{\Omega}_0$ with $\hat{\mathbb{P}}(\hat{\Omega}_{coc}) = 1$ such that (4.11) and (4.12) hold for all $\hat{\omega} \in \widehat{\Omega}_{coc}, x \in \mathbb{Z}^d$, and $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$. Part (b) is proved.

Next, we prove part (c). By (4.36) we have that $\overline{\mathbb{P}}$ -almost surely, for all $\mathcal{A} \in \mathbb{A}$, $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$, $x \in \mathbb{Z}^d$, and $z \in \mathcal{R}_{\mathcal{A}}, B_{\varepsilon}^{\mathcal{A},\beta,m}(x,x+z) \ge -|V(T_x\omega,z)| + \beta^{-1}\log p(z).$

Recall the vector $\mathbf{m}(B^{\mathcal{A},\beta,m})$ from (4.3). Using (4.11-4.12) and applying Fatou's Lemma to (4.38-4.39) we get that $B^{\mathcal{A},\beta,m}(x,x+z)$ is integrable under $\widehat{\mathbb{P}}$ and that

(4.42)
$$-\mathbb{E}[|V(T_x\omega, z)|] + \beta^{-1}\log p(z) \leq \widehat{\mathbb{E}}[B^{\mathcal{A},\beta,m}(x, x+z)] \leq m \cdot z.$$

This implies that $\widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})\cdot\zeta] \leq m\cdot\zeta$ for all $\zeta \in \mathcal{A}$. In particular, $\widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})\cdot\zeta] \leq m\cdot\xi = \Lambda^{\beta}(\xi)$.

By the variational formula [37, (2.8)], $-\Lambda^{\beta}(\xi) \ge -\mathbf{m}(B^{\mathcal{A},\beta,m}) \cdot \xi$, $\widehat{\mathbb{P}}$ -almost surely. Hence $\mathbf{m}(B^{\mathcal{A},\beta,m}) \cdot \xi =$ $\Lambda^{\beta}(\xi)$, $\widehat{\mathbb{P}}$ -almost surely. Since $\xi \in \mathrm{ri}\,\mathcal{A}$, we can scale ξ by a positive constant to obtain a direction in ri $\mathcal{U}_{\mathcal{A}}$, apply [62, Theorem 6.9], and rescale back to ξ to obtain constants $a_z > 0$ for $z \in \mathcal{R}_{\mathcal{A}}$, such that $\xi = \sum_{z \in \mathcal{R}_A} a_z z$. Then \mathbb{P} -almost surely,

$$\sum_{z \in \mathcal{R}_{\mathcal{A}}} a_z z \cdot \mathbf{m}(B^{\mathcal{A},\beta,m}) = \mathbf{m}(B^{\mathcal{A},\beta,m}) \cdot \xi = \Lambda^{\beta}(\xi) = m \cdot \xi = \sum_{z \in \mathcal{R}_{\mathcal{A}}} a_z z \cdot m.$$

Taking the expectation and rearranging, we obtain

$$\sum_{z \in \mathcal{R}_{\mathcal{A}}} a_z(m \cdot z - \widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})] \cdot z) = 0.$$

By (4.42), $\widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})] \cdot z \leq m \cdot z$ for each $z \in \mathcal{R}_{\mathcal{A}}$, so each term in the sum is non-negative. Therefore each term is exactly 0. Since $a_z > 0$, it must be that $\widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})] \cdot z = m \cdot z$ for each $z \in \mathcal{R}_{\mathcal{A}}$. Since m and $\widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})]$ are vectors of the linear subspace generated by the steps $z \in \mathcal{R}_{\mathcal{A}}$, this implies that $m = \widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})], \text{ and part (c) follows.}$

It remains to prove part (d) of the theorem. Recall the definitions of $I_{\mathcal{A}}^{\geq}$ and $I_{\mathcal{A}}^{\leq}$ in (4.7) and (4.8). Observe that the distribution of

$$\left(\{V(T_v\omega, z): z \in \mathcal{R}_{\mathcal{A}}, v \in I_{\mathcal{A}}^{<}\}, \{B^{\mathcal{A},\beta,m}(x, x+z, \widehat{\omega}): x \in I, z \in \mathcal{R}_{\mathcal{A}}, (\beta, m) \in \mathcal{D}_{\mathcal{A}}\}\right)$$

under $\widehat{\mathbb{P}}$ comes from taking subsequential limits of the distribution of

(4.43)
$$\left(\{V(T_v\omega, z) : z \in \mathcal{R}_{\mathcal{A}}, v \in I_{\mathcal{A}}^<\}, \{B_{U_n,\varepsilon}^{\mathcal{A},\beta,m}(x, x+z, \omega) : x \in I, z \in \mathcal{R}_{\mathcal{A}}, (\beta, m) \in \mathcal{D}_{\mathcal{A}}\}\right)$$

under \mathbb{P} , first taking $n \to \infty$ then $\varepsilon \to 0$. Note that the hypotheses of part (d) of the theorem imply that the two collections in (4.43) are independent. The independence claim comes then from the fact that if $\{(V(T_v\omega, z))_{z \in \mathcal{R}_A} : v \in \mathbb{Z}^d\}$ are independent under \mathbb{P} , then the two families in the above display are independent under $\overline{\mathbb{P}}$. The proof of Theorem 4.5 is complete.

The same argument that gives (4.12) from (4.29) proves the following lemma using Lemma 4.11.

Lemma 4.15. Suppose the assumptions of Theorem 4.5 hold. There exists a \hat{T} -invariant event $\hat{\Omega}_0$ with $\hat{\mathbb{P}}(\hat{\Omega}_0) = 1$ such that for all $\hat{\omega} \in \hat{\Omega}_0$, each face $\mathcal{A} \in \mathbb{A}$, m such that $(\infty, m) \in \mathcal{D}_{\mathcal{A}}$, and any finite set $\mathcal{A} \subset \mathbb{Z}^d$, there exist a $y \in \mathcal{A}$ and a $z \in \mathcal{R}_{\mathcal{A}}$ such that $y + z \notin \mathcal{A}$ and $V(T_y\omega, z) + B^{\mathcal{A},\infty,m}(y, y+z) = 0$.

We close this section with a result showing a version of weak continuity of covariant cocycles. For $\mathcal{A} \in \mathbb{A}$ let $\widetilde{\Omega}_{\mathcal{A}} = \Omega \times \mathbb{R}^{\mathbb{Z}^d \times \mathcal{R}_{\mathcal{A}}}$, equipped with the product topology and its Borel σ -algebra, $\widetilde{\mathfrak{S}}_{\mathcal{A}}$. For $\widetilde{\omega} \in \widetilde{\Omega}_{\mathcal{A}}$, let $\omega(\widetilde{\omega})$ be the projection to its Ω coordinate ω and let $B^{\mathcal{A}}(x, x + z, \widetilde{\omega})$ be the (x, z)-coordinate of $\widetilde{\omega}$.

Lemma 4.16. Fix a face $\mathcal{A} \in \mathbb{A}$. Suppose that $V^{-}(\omega, z) \in L^{1}(\mathbb{P})$ for each $z \in \mathcal{R}_{\mathcal{A}}$. Let (β, m) be such that $\beta \in (0, \infty]$, $\Lambda_{\mathcal{A}}^{\beta, \text{usc}}$ is deterministic and finite on \mathcal{A} , and there exists a $\xi \in (\text{ri }\mathcal{A}) \setminus \{\mathbf{0}\}$ such that $m \in W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi)$. Let $\beta_n \in (0, \infty]$ and $m_n \in \partial \Lambda_{\mathcal{A}}^{\beta_n, \text{usc}}(\text{ri }\mathcal{A})$ be such that $\beta_n \to \beta$ and $m_n \to m$. For each n, let B^{β_n, m_n} be an $L^1(\mathbb{P})$ shift-covariant β_n -recovering cocycle on \mathcal{A} which satisfies the conclusions of Theorem 4.5(b) and (c). Then there is a subsequence $\{n_k\}_{k\geq 1}$ such that $B^{\beta_{n_k}, m_{n_k}}$ converges weakly to a random variable $B^{\beta, m}$ on $(\widetilde{\Omega}, \widetilde{\mathfrak{S}})$, which also satisfies the conclusions of Theorem 4.5(b) and (c) on \mathcal{A} .

Proof. Let $\beta_n \to \beta$, $m_n \to m$, and B^{β_n,m_n} be as in the statement. Recovery implies that for $x \in \mathbb{Z}^d$ and $z \in \mathcal{A}$, $B^{\beta_n,m_n}(x,x+z) \ge V(\omega(\hat{T}_x\hat{\omega}),z) - \beta_n^{-1}\log p(z)$; combining this with convergence of the mean vectors $m_n \to m$, we see that the distributions of $(\omega, B^{\beta_n,m_n}) \in \tilde{\Omega}$ under \mathbb{P} are tight. Let $\tilde{\mathbb{P}}$ be a weak limit along a subsequence $\{n_k\}_{k\ge 1}$. Then, we can repeat the proof of Theorem 4.5(c) to show that, under $\tilde{\mathbb{P}}$, the coordinate projection $B^{\mathcal{A}}(x, x+z)$ is integrable with mean $m \cdot z$. The covariance property (4.9) and cocycle property (4.10) hold for each B^{β_n,m_n} and are inherited in the limit since coordinate projections and shifts are continuous in the product topology.

Define the function $f: (0, \infty] \times \mathbb{R}^{\mathbb{Z}^d \times \mathcal{R}_{\mathcal{A}}} \to \mathbb{R}$ by

$$f(\beta, B) = \begin{cases} \frac{1}{\beta} \log \sum_{z \in \mathcal{R}_{\mathcal{A}}} p(z) \exp\{-\beta B(x, x+z) - \beta V(\omega(\widetilde{T}_{x}\widetilde{\omega}), z)\} & \text{if } \beta \in (0, \infty) \\ f(\infty, B) = -\min_{z \in \mathcal{R}_{\mathcal{A}}} (B(x, x+z) + V(\omega(\widetilde{T}_{x}\widetilde{\omega}), z)). \end{cases}$$

Note that f is continuous on $(0, \infty] \times \mathbb{R}^{\mathbb{Z}^d \times \mathcal{R}_A}$, including at $\beta = \infty$.

Thus, the recovery properties (4.11) and (4.12) that $B^{\beta_{n_k},m_{n_k}}$ satisfy for each k can be transferred to $B^{\mathcal{A}}$ by applying Skorohod's representation theorem.

5. Semi-infinite path measures

We begin by defining semi-infinite polymer measures from a shift-covariant recovering cocycle. Throughout this section, $(\Omega, \mathfrak{S}, \mathbb{P}, \{T_x : x \in \mathbb{Z}^d\})$ satisfies the general stationary setting described in the first paragraph of Section 2.2. We note that the extended space $(\widehat{\Omega}, \widehat{\mathfrak{S}}, \widehat{\mathbb{P}}, \{\widehat{T}_x : x \in \mathbb{Z}^d\})$ from Theorem 4.5 satisfies these hypotheses. We switch to this case in Section 6 where we prove Theorem 3.14.

Definition 5.1. Let B be an $L^1(\mathbb{P})$ T-covariant β -recovering cocycle on a face $\mathcal{A} \in \mathbb{A}$ (possibly C itself). For $x_0 \in \mathbb{Z}^d$, $0 < \beta < \infty$, and $\omega \in \Omega$ such that (4.1) is satisfied for all $x \in \mathbb{Z}^d$, define $Q_{x_0}^{\beta,B,\omega}$ to be the probability measure on $\mathbb{X}_{x_0}(\mathcal{R}_{\mathcal{A}})$ that is the distribution of the Markov process $X_{0:\infty}$ with transition probability from $x \in \mathbb{Z}^d$ to x + z, $z \in \mathcal{R}_{\mathcal{A}}$, given by

$$p(z)e^{-\beta B(x,x+z,\omega)-\beta V(T_x\omega,z)}$$

For a subset $A \subset \mathbb{Z}^d$, let \mathcal{P}_A be the set of all probability measures on A, with the convention that $\mathcal{P}_{\varnothing} = \varnothing$.

Definition 5.2. A tie-breaker is an element $\mathfrak{t} \in \left(\prod_{A \subset \mathcal{R}} \mathcal{P}_A\right)^{\mathbb{Z}^d}$. A covariant tie-breaker is a measurable function $\mathfrak{t} : \Omega \to \left(\prod_{A \subset \mathcal{R}} \mathcal{P}_A\right)^{\mathbb{Z}^d}$ that is covariant: for any $x, z \in \mathbb{Z}^d$ and $A \subset \mathcal{R}$, $\mathfrak{t}_{x,A}^{T_z \omega} = \mathfrak{t}_{x+z,A}^{\omega}$.

Definition 5.3. Let B be an $L^1(\mathbb{P})$ T-covariant ∞ -recovering cocycle on a face $\mathcal{A} \in \mathbb{A}$ (possibly C itself). For $x_0 \in \mathbb{Z}^d$, $\omega \in \Omega$ such that (4.2) is satisfied for all $x \in \mathbb{Z}^d$, and a tie-breaker \mathfrak{t} define $Q_{x_0}^{\infty,B,\mathfrak{t},\omega}$ to be the probability measure on $\mathbb{X}_{x_0}(\mathcal{R}_{\mathcal{A}})$ that is the distribution of the Markov process $X_{0:\infty}$ with transition probability from $x \in \mathbb{Z}^d$ to x + z, $z \in \mathcal{R}_{\mathcal{A}}$, given by $\mathfrak{t}_{x,A}(z)$, where

$$A = \{ z \in \mathcal{R}_{\mathcal{A}} : B(x, x + z, \omega) + V(T_x \omega, z) = 0 \}.$$

Example 5.4. Let S denote the set of bijections from \mathcal{R} to $\{1, \ldots, |\mathcal{R}|\}$. One natural way to define a covariant tie-breaker is by taking a covariant measurable function $\mathfrak{z} : \Omega \to S^{\mathbb{Z}^d}$ and setting $\mathfrak{t}_{x,A}^{\omega}$ to be a Dirac mass at the step in A that has the lowest \mathfrak{z}_x^{ω} -value. In this case, the probability measure $Q_{x_0}^{\infty,B,\mathfrak{t},\omega}$ is a Dirac mass on a single semi-infinite path in $\mathbb{X}_{x_0}(\mathcal{R}_A)$, which is the path that out of x follows the increment that satisfies $B(x, x + z, \omega) + V(T_x\omega, z) = 0$ and uses the ranking given by \mathfrak{z}_x to break the tie if there are multiple such increments.

In the next theorem, we will use a condition to guarantee the transience of semi-infinite geodesic paths at zero temperature.

Condition 5.5. Given a face $A \in A$ (possibly C itself) and an ∞ -recovering cocycle B on the face A, assume that \mathbb{P} -almost surely, for any finite set $A \subset \mathbb{Z}^d$, there exist a $y \in A$ and a $z \in \mathcal{R}_A$ such that $y + z \notin A$ and $V(T_y\omega, z) + B(y, y + z) = 0$.

Remark 5.6. This condition is necessary for transience at zero temperature since a set A violating the condition acts as a trap from which the semi-infinite path cannot exit, $Q_u^{\infty,B,t,\omega}$ -almost surely. We will show in Lemma 5.15 below that the converse is also true. Hence, this condition is, in fact, equivalent to having transient paths.

Remark 5.7. Lemma 5.17 below shows that ∞ -recovering cocycles that violate the above condition may exist. Nevertheless, by Lemma 4.15, Condition 5.5 is always satisfied for the cocycles $B^{\mathcal{A},\beta,m}$ that we construct in Theorem 4.5.

The following theorem shows that the measures $Q_u^{\beta,B,\omega}$ are consistent with both the restricted-length and unrestricted-length point-to-point measures. It also relates these measures to the Green's function (3.7). As the next result is a precursor to Theorem 3.14, it may be helpful to consult the statement and discussion around that theorem to contextualize the notation below.

Theorem 5.8. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Assume V satisfies Conditions 3.2 and 3.5 on \mathcal{A} . Let B be an $L^1(\mathbb{P})$ T-covariant β -recovering cocycle on the face \mathcal{A} . There exists a T-invariant event $\Omega_B \subset \Omega$ with $\mathbb{P}(\Omega_B) = 1$ such that for all $\omega \in \Omega_B$, the following hold.

(a) (Consistency with $Q_{x,y}^{\beta,\omega}$) If $\beta < \infty$, for all non-negative integers $j \leq k \leq n$, points $u, x, y \in \mathbb{Z}^d$ with x - u and y - x in $\mathcal{G}^+_{\mathcal{A}}$, $x_{0:k} \in \overline{\mathbb{X}}_{u,y}$ with $j = \min\{0 \leq i \leq k : x_i = x\}$, and $x_{k:n} \in \mathbb{X}_y^{n-k}(\mathcal{R}_{\mathcal{A}})$, then

(5.1)
$$Q_{u}^{\beta,B,\omega}(X_{0:\tau_{y}+n-k} = x_{0:n} | \tau_{x} \leqslant \tau_{y} < \infty, X_{0:\tau_{x}} = x_{0:j}, X_{\tau_{y}:\tau_{y}+n-k} = x_{k:n}) = Q_{u}^{\beta,B,\omega}(X_{\tau_{x}:\tau_{y}} = x_{j:k} | \tau_{x} \leqslant \tau_{y} < \infty) = Q_{x,y}^{\beta,\omega}(X_{0:\tau_{y}} = x_{j:k}).$$

(b) (Consistency with g(x, y).) If $\beta < \infty$, then for each $x, y \in \mathbb{Z}^d$ such that $y - x \in \mathcal{G}_A^+$,

(5.2)
$$\mathrm{E}^{\mathrm{Q}^{\beta,B,\omega}_{x}}\left[\sum_{n=0}^{\infty}\mathbb{1}_{\{X_{n}=y\}}\right] = g(x,y)e^{-\beta B(x,y)}$$

Hence, if $x \neq y$,

(5.3)
$$\mathbb{E}^{\mathbb{Q}_x^{\beta,B,\omega}} \left[\sum_{n=0}^{\infty} \mathbb{1}_{\{X_n=y\}} \middle| \tau_y < \infty \right] = \frac{g(x,y)}{Z_{x,y}^{\beta,\omega}}$$

(c) (Consistency with $Q_{x,y,n}^{\beta,\omega}$.) If $\beta < \infty$, for all non-negative integers $j \leq k \leq n$, points $u, x, y \in \mathbb{Z}^d$ such that $x - u \in D_j(\mathcal{R}_A)$ and $y - x \in D_{k-j}(\mathcal{R}_A)$, $x_{0:j} \in \mathbb{X}_{u,x}^{0,j}$, $x_{j:k} \in \mathbb{X}_{x,y}^{j,k}$, and $x_{k:n} \in \mathbb{X}_y^{n-k}(\mathcal{R}_A)$,

(5.4)
$$Q_{u}^{\beta,B,\omega}(X_{0:n} = x_{0:n} | X_{0:j} = x_{0:j}, X_{k:n} = x_{k:n}) = Q_{u}^{\beta,B,\omega}(X_{j:k} = x_{j:k} | X_{j} = x, X_{k} = y) \\ = Q_{x,y,k-j}^{\beta,\omega}(X_{0:k-j} = x_{j:k}).$$

(d) If $\beta = \infty$, for each $x \in \mathbb{Z}^d$ and tie-breaker \mathfrak{t} , $\mathbf{Q}_x^{\infty,B,\mathfrak{t},\omega}$ is supported on a set of semi-infinite geodesics rooted at x and having increments in \mathcal{R}_A .

(e) If $\beta < \infty$, then for each $x \in \mathbb{Z}^d$,

{

 $Q_x^{\beta,B,\omega}(all\ limit\ points\ of\ X_n/n\ are\ contained\ in\ \mathcal{U}_{\mathbf{m}(B),\mathcal{A}}^{\beta}) = 1.$

Similarly, if $\beta = \infty$, then for each $x \in \mathbb{Z}^d$ and tie-breaker \mathfrak{t} ,

 $Q_x^{\beta,B,\mathfrak{t},\omega}(all\ limit\ points\ of\ X_n/n\ are\ contained\ in\ \mathcal{U}^{\beta}_{\mathbf{m}(B),\mathcal{A}})=1.$

Furthermore, we have the following.

- (f) If $\beta < \infty$, then for any (and hence all) $x \in \mathbb{Z}^d$, $Q_x^{\beta,B,\omega}\{|X_n|_1 \to \infty\} = 1$ \mathbb{P} -almost surely if and only if either $\mathcal{A} \neq \mathcal{A}_0$ or both $\mathcal{A} = \mathcal{A}_0$ and Condition 3.8 holds.
- (g) If $\beta = \infty$ and Condition 5.5 is satisfied, then there exists a covariant tie-breaker t such that for \mathbb{P} -almost every ω and for every $x \in \mathbb{Z}^d$, $Q_x^{\infty,B,\mathfrak{t},\omega}\{|X_n|_1 \to \infty\} = 1$.
- (h) For $\beta \in (0, \infty)$, if $Q_{\mathbf{0}}^{\beta, B, \omega}(|X_n|_1 \to \infty) = 1$, \mathbb{P} -almost surely, then for \mathbb{P} -almost every ω , for all $x \in \mathbb{Z}^d$, $Q_x^{\beta, B, \omega}(X_{0:\infty} \text{ is directed into } \mathcal{F}^{\beta}_{\mathbf{m}(B), \mathcal{A}}) = 1.$

Similarly, for $\beta = \infty$ and a covariant tie-breaker \mathfrak{t} , if $\mathbf{Q}_{\mathbf{0}}^{\beta,B,\mathfrak{t},\omega}(|X_n|_1 \to \infty) = 1$, \mathbb{P} -almost surely, then for \mathbb{P} -almost every ω , for all $x \in \mathbb{Z}^d$,

$$Q_x^{\beta,B,\mathfrak{t},\omega}(X_{0:\infty} \text{ is directed into } \mathcal{F}_{\mathbf{m}(B),\mathcal{A}}^{\beta}) = 1.$$

Remark 5.9. If we consider a restricted-length model with V satisfying Conditions 3.3 and 3.5, then we can use the space-time cocycles from Remark 4.6 to define semi-infinite polymer measures that are consistent with the restricted-length point-to-point polymer measures. Unlike the measures in the above theorem, these polymer measures are not necessarily consistent with the unrestricted-length point-to-point polymer measures. \triangle

We will prove the various claims in the above theorem as separate lemmas, beginning with the consistency with the finite-path measures.

Lemma 5.10. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Let $0 < \beta < \infty$ and B be an $L^1(\mathbb{P})$ T-covariant β -recovering cocycle on \mathcal{A} . Take $\omega \in \Omega$ such that (4.1) holds for all $x \in \mathbb{Z}^d$. Then for all non-negative integers $j \leq k \leq n$, points $u, x, y \in \mathbb{Z}^d$ with $x - u \in \mathcal{G}^+_{\mathcal{A}}$ and $y - x \in \mathcal{G}^+_{\mathcal{A}}$, $x_{0:k} \in \overline{\mathbb{X}}_{u,y}$ with $j = \min\{0 \leq i \leq k : x_i = x\}$, and $x_{k:n} \in \mathbb{X}^{n-k}_y(\mathcal{R}_{\mathcal{A}})$, (5.1) holds.

Proof. Take β , B, ω , j, k, n, u, x, y, $x_{0:n}$ as in the statement. Let $\tau_{x,y} = \inf\{k \ge \tau_x : X_k = y\}$ where $\inf \emptyset = \infty$. Note that $\{\tau_x \le \tau_y\}$ is in the stopped σ -algebra corresponding to the stopping time τ_x : for any integer $t \ge 0$,

$$\tau_x = t\} \cap \{\tau_x \leqslant \tau_y\} = \{\tau_x = t\} \cap \{\tau_y \ge t\} \in \sigma(X_k : k \leqslant t)$$

In the second equality of the next computation, use the fact that on the event $\{\tau_x \leq \tau_y\}$, $\tau_y = \tau_{x,y}$. In the third equality, use the above measurability observation and the strong Markov property. In the second-to-last equality, use the cocycle property. We have

$$\begin{split} Q_{u}^{\beta,B,\omega}(X_{\tau_{x}:\tau_{y}} = x_{j:k} \mid \tau_{x} \leqslant \tau_{y} < \infty) \\ &= \frac{Q_{u}^{\beta,B,\omega}(\tau_{x} \leqslant \tau_{y} < \infty, X_{\tau_{x}:\tau_{y}} = x_{j:k})}{Q_{u}^{\beta,B,\omega}(\tau_{x} \leqslant \tau_{y} < \infty)} \\ &= \frac{Q_{u}^{\beta,B,\omega}(\tau_{x} \leqslant \tau_{y}, \tau_{x,y} < \infty, X_{\tau_{x}:\tau_{x,y}} = x_{j:k})}{Q_{u}^{\beta,B,\omega}(\tau_{x} \leqslant \tau_{y}, \tau_{x,y} < \infty)} \\ &= \frac{Q_{u}^{\beta,B,\omega}(\tau_{x} \leqslant \tau_{y})Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty, X_{0:\tau_{y}} = x_{j:k})}{Q_{u}^{\beta,B,\omega}(\tau_{x} \leqslant \tau_{y})Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty)} \\ &= \frac{\prod_{i=j}^{k-1} p(x_{i+1} - x_{i})e^{-\beta B(x_{i},x_{i+1},\omega) -\beta V(T_{x_{i}}\omega,x_{i+1} - x_{i})}}{\sum_{k=0}^{\infty} \sum_{\pi_{0:k} \in \overline{\mathbb{X}}_{x,y}^{k}} \prod_{i=0}^{k-1} p(\pi_{i+1} - \pi_{i})e^{-\beta B(\pi_{i},\pi_{i+1},\omega) -\beta V(T_{\pi_{i}}\omega,\pi_{i+1} - \pi_{i})}} \\ &= \frac{e^{-\beta B(x,y,\omega)} \prod_{i=j}^{k-1} p(x_{i+1} - x_{i})e^{-\beta V(T_{x_{i}}\omega,x_{i+1} - x_{i})}}{e^{-\beta B(x,y,\omega)}Z_{x,y}^{\beta,\omega}} \\ &= Q_{x,y}^{\beta,\omega}(X_{0:\tau_{y}} = x_{j:k}). \end{split}$$

Similarly,

$$\begin{aligned} Q_{u}^{\beta,B,\omega}(X_{0:\tau_{y}+n-k} = x_{0:n} \mid \tau_{x} \leqslant \tau_{y} < \infty, X_{0:\tau_{x}} = x_{0:j}, X_{\tau_{y}:\tau_{y}+n-k} = x_{k:n}) \\ &= \frac{Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty, X_{0:\tau_{y}+n-k} = x_{j:n})}{Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty, X_{0:\tau_{y}} = x_{j:k})Q_{y}^{\beta,B,\omega}(X_{0:n-k} = x_{k:n})} \\ &= \frac{Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty, X_{0:\tau_{y}} = x_{j:k})Q_{y}^{\beta,B,\omega}(X_{0:n-k} = x_{k:n})}{Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty)Q_{y}^{\beta,B,\omega}(X_{0:n-k} = x_{k:n})} \\ &= \frac{Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty, X_{0:\tau_{y}} = x_{j:k})}{Q_{x}^{\beta,B,\omega}(\tau_{y} < \infty)} \\ &= Q_{x,y}^{\beta,\omega}(X_{0:\tau_{y}} = x_{j:k}), \end{aligned}$$

where the last equality comes from the previous calculation.

Lemma 5.11. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Fix $0 < \beta < \infty$ and let B be an $L^1(\mathbb{P})$ T-covariant β -recovering cocycle on \mathcal{A} . For each $x, y \in \mathbb{Z}^d$ such that $y - x \in \mathcal{G}^+_{\mathcal{A}}$, (5.2) and (5.3) hold.

Proof. Take β , B, x, y as in the statement. First, use the cocycle property to obtain

$$Q_x^{\beta,B,\omega}(X_n = y) = \sum_{x_{0:n} \in \mathbb{X}_{x,y}^n} \prod_{i=0}^{n-1} p(x_{i+1} - x_i) e^{-\beta B(x_i, x_{i+1}) - \beta V(T_{x_i}, \omega, x_{i+1} - x_i)} = Z_{x,y,n}^{\beta,\omega} e^{-\beta B(x,y)}.$$

Sum over n to obtain (5.2):

$$\mathbf{E}^{\mathbf{Q}_{x}^{\beta,B,\omega}}\left[\sum_{n=0}^{\infty}\mathbb{1}_{\{X_{n}=y\}}\right] = \sum_{n=0}^{\infty}\mathbf{Q}_{x}^{\beta,B,\omega}(X_{n}=y) = \sum_{n=0}^{\infty}Z_{x,y,n}^{\beta,\omega}e^{-\beta B(x,y)} = g(x,y)e^{-\beta B(x,y)}.$$

Similarly, if $x \neq y$,

(5.5)
$$Q_x^{\beta,B,\omega}(\tau_y < \infty) = \sum_{n=0}^{\infty} \sum_{x_{0:n} \in \overline{\mathbb{X}}_{x,y}^n} \prod_{i=0}^{n-1} p(x_{i+1} - x_i) e^{-\beta B(x_i, x_{i+1}) - \beta V(T_{x_i}\omega, x_{i+1} - x_i)} = Z_{x,y}^{\beta,\omega} e^{-\beta B(x,y)}.$$

Combining the two previous displays, we obtain (5.3).

We next show consistency with the restricted-length finite quenched polymer measures.

Lemma 5.12. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Fix $0 < \beta < \infty$ and let B be an $L^1(\mathbb{P})$ T-covariant β -recovering cocycle on \mathcal{A} . Take $\omega \in \Omega$ such that (4.1) holds for all $x \in \mathbb{Z}^d$. For $j, k, n \in \mathbb{Z}_{\geq 0}$ with $j \leq k \leq n$, $u, x, y \in \mathbb{Z}^d$ such that $x - u \in D_j(\mathcal{R}_{\mathcal{A}})$ and $y - x \in D_{k-j}(\mathcal{R}_{\mathcal{A}})$, $x_{0:j} \in \mathbb{X}_{u,x}^{0,j}$, $x_{j:k} \in \mathbb{X}_{x,y}^{j,k}$, and $x_{k:n} \in \mathbb{X}_y^{n-k}(\mathcal{R}_{\mathcal{A}})$, (5.4) holds.

Proof. Take β , B, ω , j, k, n, u, x, y, and $x_{0:n}$ as in the statement. Using the Markov property and the cocycle property,

$$\begin{aligned} Q_{u}^{\beta,B,\omega}(X_{0:n} = x_{0:n} | X_{0:j} = x_{0:j}, X_{k:n} = x_{k:n}) \\ &= \frac{Q_{u}^{\beta,B,\omega}(X_{0:n} = x_{0:n})}{Q_{u}^{\beta,B,\omega}(X_{0:j} = x_{0:j})Q_{x}^{\beta,B,\omega}(X_{0:k-j} = x_{j:k})Q_{y}^{\beta,B,\omega}(X_{0:n-k} = x_{k:n})} \\ &= \frac{Q_{u}^{\beta,B,\omega}(X_{0:j} = x_{0:j})Q_{x}^{\beta,B,\omega}(X_{0:k-j} = x_{j:k})Q_{y}^{\beta,B,\omega}(X_{0:n-k} = x_{k:n})}{Q_{u}^{\beta,B,\omega}(X_{0:j} = x_{0:j})Q_{x}^{\beta,B,\omega}(X_{k-j} = y)Q_{y}^{\beta,B,\omega}(X_{0:n-k} = x_{k:n})} \\ &= \frac{\prod_{i=j}^{k-1} p(x_{i+1} - x_{i})e^{-\beta B(x_{i},x_{i+1},\omega) - \beta V(T_{x_{i}}\omega,x_{i+1} - x_{i})}}{\sum_{\pi_{0:k-j} \in \mathbb{X}_{x,y}^{k,j}}\prod_{i=0}^{k-j-1} p(\pi_{i+1} - \pi_{i})e^{-\beta B(\pi_{i},\pi_{i+1},\omega) - \beta V(T_{\pi_{i}}\omega,\pi_{i+1} - \pi_{i})}} \\ &= \frac{e^{-\beta B(x,y,\omega)}\prod_{i=j}^{k-1} p(x_{i+1} - x_{i})e^{-\beta V(T_{x_{i}}\omega,x_{i+1} - x_{i})}}{e^{-\beta B(x,y,\omega)}Z_{x,y,k-j}^{\beta,\omega}} \\ &= Q_{x,y,k-j}^{\beta,\omega}(X_{0:k-j} = x_{j:k}). \end{aligned}$$

Now we work out the consistency in the zero temperature case. Recall that the definition of the quenched measures, in this case, requires a tie-breaker \mathfrak{t} .

Lemma 5.13. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Let B be an $L^1(\mathbb{P})$ T-covariant ∞ -recovering cocycle on \mathcal{A} . Let $\omega \in \Omega$ be such that (4.2) is satisfied for all $x \in \mathbb{Z}^d$. Then, for each $x \in \mathbb{Z}^d$, for any tie-breaker \mathfrak{t} , $Q_x^{\infty,B,\mathfrak{t},\omega}$ is supported on a set of semi-infinite geodesics rooted at x and having increments in $\mathcal{R}_{\mathcal{A}}$.

Proof. The measure $Q_x^{\infty,B,\mathfrak{t},\omega}$ only allows steps $z \in \mathcal{R}_A$, so the paths stay within x + A. Next, observe that $Q_x^{\infty,B,\mathfrak{t},\omega}$ -almost surely, $B(X_i, X_{i+1}) = -V(T_{X_i}\omega, X_{i+1} - X_i)$ for all $i \in \mathbb{Z}_{\geq 0}$. We will show that this implies that $X_{0:\infty}$ is a geodesic.

Let $x_{0:\infty} \in \mathbb{X}_x(\mathcal{R}_{\mathcal{A}})$ be a semi-infinite path such that $B(x_i, x_{i+1}) = -V(T_{x_i}\omega, x_{i+1} - x_i)$ for each $i \in \mathbb{Z}_{\geq 0}$. Let $0 \leq j < k, n \in \mathbb{Z}_{>0}$, and $y_{0:n} \in \mathbb{X}_{x_i, x_k}^n$. Using the cocycle and recovery properties,

$$-\sum_{i=j}^{k-1} V(T_{x_i}\omega, x_{i+1} - x_i) = \sum_{i=j}^{k-1} B(x_i, x_{i+1}) = B(x_j, x_k)$$
$$= \sum_{i=0}^{n-1} B(y_i, y_{i+1}) \ge -\sum_{i=0}^{n-1} V(T_{y_i}\omega, y_{i+1} - y_i).$$

Since this holds for all lengths n and all paths $y_{0:n}$ from x_j to x_k , $x_{j:k}$ is a geodesic from x_j to x_k . Since this holds for all $0 \leq j < k$, $x_{0:\infty}$ is a semi-infinite geodesic rooted at x.

Now that the consistency properties have been proven, we turn to directedness. We first address the question of transience versus recurrence of the semi-infinite polymers. Recall that \mathcal{U}_0 is the unique face of \mathcal{U} that contains 0 in its relative interior and that $\mathcal{R}_0 = \mathcal{R} \cap \mathcal{U}_0$.

Lemma 5.14. Fix a face $A \in A$ (possibly C itself). Assume (3.4) holds and $\beta \in (0, \infty)$. Let B be an $L^1(\mathbb{P})$ *T*-covariant β -recovering cocycle on A. The following are equivalent:

- (i) Either $\mathcal{A} \neq \mathcal{A}_0$ or both $\mathcal{A} = \mathcal{A}_0$ and Condition 3.8 holds.
- (ii) There exists an $x \in \mathbb{Z}^d$ such that \mathbb{P} -almost surely $Q_x^{\beta,B,\omega}\{|X_n|_1 \to \infty\} = 1$.
- (iii) \mathbb{P} -almost surely, for all $x \in \mathbb{Z}^d$, $\mathbb{Q}_x^{\beta,B,\omega}\{|X_n|_1 \to \infty\} = 1$.

Proof. Note that (ii) and (iii) are equivalent due to the shift-invariance of \mathbb{P} and the shift-covariance of B. We, therefore, fix an $x \in \mathbb{Z}^d$ and prove that (i) is equivalent to (ii) with this choice of x.

From (5.2) and a standard fact about time-homogeneous Markov chains, we see that the Markov chain $Q_x^{\beta,B,\omega}$ is recurrent if and only if $g(x,x) = \infty$. Let σ_1 be the time of first return of X_n to the starting point: $\sigma_1 = \inf\{k \ge 1 : X_k = X_0\}$. Lemma 6.2 in [37] gives

(5.6)
$$g(x,x) = \frac{1}{1 - E_x \left[e^{-\beta \sum_{k=0}^{\sigma_1 - 1} V(T_{X_k}\omega, X_{k+1} - X_k)} \mathbb{1}\{\sigma_1 < \infty\} \right]} \leqslant \frac{1}{P_x(\sigma_1 = \infty)}$$

If the reference random walk P_0 is transient, then the above implies that $g(x, x) < \infty$ and, therefore, the Markov chain $Q_x^{\beta,B,\omega}$ is transient. This includes the case $\mathcal{A} \neq \mathcal{A}_0$ because, in that case, $\mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_0 \neq \emptyset$, and once the reference random walk takes a step in $\mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_0$, it will never return to its starting point.

Suppose that P_0 is recurrent and, in particular, $\mathcal{A} = \mathcal{A}_0$. Assume Condition 3.8 holds. We will prove that, \mathbb{P} -almost surely, $Q_x^{\beta,B,\omega}$ is again transient, for all $x \in \mathbb{Z}^d$.

Without loss of generality, let x = 0. By Condition 3.8, with \mathbb{P} -probability one, there exist $y \in \mathcal{G}_{\mathcal{A}_0}^+$, $z_0 \in \mathcal{R}_0$, and $\varepsilon > 0$ such that $V(T_y \omega, z_0) \ge \varepsilon$. [37, Lemma 3.4] shows that y can be written as $y = \sum_{z \in \mathcal{R}_0} \gamma_z z$ with $\gamma_z \in \mathbb{Z}_{\ge 0}$ and $\gamma_z \le C|y|_1$ for all $z \in \mathcal{R}_0$ and a finite positive constant C. Use this to obtain an admissible path $y_{0:n}$ from 0 to y of length $n = \sum_{z \in \mathcal{R}_0} \gamma_z \le C|\mathcal{R}_0| \cdot |y|_1$. Next, we will extend this path to obtain a loop from 0 to 0. Let $y_{n+1} = y + z_0$. Then, use the same argument to construct a path $y_{n+1:m}$ from $y + z_0$ to 0, with length $m - n - 1 \le C'|\mathcal{R}_0| \cdot |y|_1$. Then $y_{0:m}$ is a loop which starts at 0, travels to y, takes a step to $y + z_0$, and returns to 0. The length $m \le C_1|y|_1$, where $C_1 = (C + C')|\mathcal{R}_0| + 1$. Let $\kappa = \min_{z \in \mathcal{R}_0} p(z)$. Then, $\mathcal{P}_0(X_{0:m} = y_{0:m}) \ge \kappa^{C_1|y|_1}$.

For the loop constructed in the above paragraph,

$$\exp\left\{-\beta\sum_{i=0}^{m-1}V(T_{y_i}\omega, y_{i+1}-y_i)\right\} \leqslant \exp\{-\beta V(T_y\omega, z_0)\} \leqslant e^{-\beta\varepsilon}.$$

On the other hand, for any loop $y'_{0:\ell}$ from **0** to **0**,

$$\exp\left\{-\beta \sum_{i=0}^{\ell-1} V(T_{y'_i}\omega, y'_{i+1} - y'_i)\right\} \le 1$$

because $V(T_v\omega, z) \ge 0$ for all $v \in \mathcal{G}^+_{\mathcal{A}_0}$ and $z \in \mathcal{R}_0$. Therefore,

$$\mathbb{E}_{\mathbf{0}} \Big[e^{-\beta \sum_{i=0}^{\sigma_{1}-1} V(T_{X_{k}}\omega, X_{k+1}-X_{k})} \mathbb{1}_{\{\sigma_{1}<\infty\}} \Big] \leq \mathbb{P}_{\mathbf{0}}(X_{0:m} \neq y_{0:m}) + e^{-\beta\varepsilon} \mathbb{P}_{\mathbf{0}}(X_{0:m} = y_{0:m})$$

= 1 - (1 - e^{-\beta\varepsilon}) $\mathbb{P}_{\mathbf{0}}(X_{0:m} = y_{0:m}) \leq 1 - (1 - e^{-\beta\varepsilon}) \kappa^{C_{1}|y|_{1}}$

Using (5.6), we see that \mathbb{P} -almost surely,

$$g(\mathbf{0}, \mathbf{0}) \leq \kappa^{-C_1 |y|_1} (1 - e^{-\beta \varepsilon})^{-1}$$

and $\mathbf{Q}_{\mathbf{0}}^{\beta,B,\omega}$ is transient.

If, on the other hand, Condition 3.8 is violated, then with positive \mathbb{P} -probability, $V(T_y\omega, z) = 0$ for all $y \in \mathcal{G}_{\mathcal{A}_0}^+$ and $z \in \mathcal{R}_{\mathcal{A}}$. On this event, $Q_0^{\beta,B,\omega}$ -almost surely, $V(T_{X_k}\omega, X_{k+1} - X_k) = 0$ for all $k \in \mathbb{Z}_{\geq 0}$ and (5.6) gives $g(\mathbf{0}, \mathbf{0}) = (1 - P_{\mathbf{0}}(\sigma_1 < \infty))^{-1} = \infty$. The lemma is proved.

Lemma 5.15. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Assume (3.4) holds and $\beta = \infty$. Let B be an $L^1(\mathbb{P})$ T-covariant ∞ -recovering cocycle on \mathcal{A} . Assume B satisfies Condition 5.5. Then there exists a covariant tie-breaker \mathfrak{t} such that, \mathbb{P} -almost surely, for any $x \in \mathbb{Z}^d$, $\mathbb{Q}_x^{\infty,B,\mathfrak{t},\omega}\{|X_n|_1 \to \infty\} = 1$.

Before we prove the lemma, we need a definition and an intermediate result. Given an ∞ -recovering cocycle B on a face $\mathcal{A} \in \mathbb{A}$ we will say that x and y in \mathbb{Z}^d communicate, and will write $x \leftrightarrow y$, if there exist paths $x_{0:k} \in \mathbb{X}_{x,y}(\mathcal{R}_{\mathcal{A}})$ and $y_{0:\ell} \in \mathbb{X}_{y,x}(\mathcal{R}_{\mathcal{A}})$, $k, \ell \in \mathbb{Z}_{>0}$, such that $V(T_{x_i}\omega, x_{i+1} - x_i) = B(x_i, x_{i+1}) = 0$ and $V(T_{y_j}\omega, y_{j+1} - y_j) = B(y_j, y_{j+1}) = 0$ for all integers $i \in [0, k-1]$ and $j \in [0, \ell-1]$. Note that $\leftrightarrow i$ is symmetric and transitive, but it is not necessarily reflexive. Namely, $x \leftrightarrow x$ holds if and only if there is a non-empty admissible loop from x to x along which V and B vanish. In other words, x does not communicate with itself if and only if it does not communicate with any site $y \in \mathbb{Z}^d$. We will call such a site *non-essential* and the sites that do communicate with themselves *essential*. Hence, $\leftrightarrow i$ is an equivalence relation on the set of essential sites.

Lemma 5.16. Fix a face $A \in A$ (possibly C itself). Assume (3.4) holds and $\beta = \infty$. Let B be an ∞ -recovering cocycle on A. If there exists a path $x_{0:k} \in \mathbb{X}_{x,x}^k$ with $x \in \mathbb{Z}^d$ and $k \in \mathbb{Z}_{>0}$, such that

(5.7)
$$V(T_{x_i}\omega, x_{i+1} - x_i) + B(x_i, x_{i+1}) = 0 \text{ for all integers } i \in [0, k-1],$$

then x is essential.

Proof. Sum (5.7) over *i* and use the cocycle property and B(x, x) = 0 to get that $\sum_{i=0}^{k-1} V(T_{x_i}\omega, x_{i+1}-x_i) = 0$. Next, observe that the steps of the loop $x_{0:k}$ are all in \mathcal{R}_0 . By (3.4), we get that $V(T_{x_i}\omega, x_{i+1}-x_i) = 0$ along the loop, which, in turn, implies that $B(x_i, x_{i+1}) = 0$ along the loop. Consequently, $x \leftrightarrow x$, and the lemma is proved.

Proof of Lemma 5.15. We begin by defining the covariant tie-breaker. To this end, fix a bijection \mathfrak{z} from \mathcal{R} to $\{1, \ldots, |\mathcal{R}|\}$. For any non-essential point x and subset $A \subset \mathcal{R}$, we define $\mathfrak{t}_{x,A}$ as a Dirac mass at the step in A that has the minimum \mathfrak{z} value.

Fix another bijection \mathfrak{z}' from \mathbb{Z}^d to $\mathbb{Z}_{>0}$. If x is an essential point, then let A be its equivalence class. We treat first the case where A is finite. Consider the pairs (y, z) that appear in Condition 5.5 for this set A. Among these pairs, choose the ones where y has the smallest \mathfrak{z}' value. If there are multiple such pairs, select the unique pair with the lowest \mathfrak{z} value for z.

Next, for a set $C \subset \mathcal{R}$ containing z, we define $\mathfrak{t}_{y,C}$ as a Dirac mass at z. If $z \notin C$, let $\mathfrak{t}_{y,C}$ be a Dirac mass at the point in C with the lowest \mathfrak{z} value.

To define the tie-breaker at points in $A \setminus \{y\}$, we first construct a spanning tree of A as follows. We start by selecting the point $u \in A \setminus \{y\}$ with the smallest \mathfrak{z}' value among all such points. We then choose the first step $z_1 \in \mathcal{R}_0$ of an admissible path from u to y that remains entirely within A and avoids u after leaving it. If there are multiple choices for z_1 , we select the one with the smallest \mathfrak{z} value. We continue by selecting the first step $z_2 \in \mathcal{R}_0$ of an admissible path from $u + z_1$ to y that remains entirely within A, does not visit u, and avoids $u + z_1$ after leaving it. We iterate this process until we reach y, constructing a path from u to y that forms a branch of the spanning tree.

Next, we select a point v in A that is not on the just constructed path and that has the smallest \mathfrak{z}' value among all such points. We then construct a path from v to y, similarly to how we constructed the path from u to y, except that if this second path intersects with the first one, we stop its construction at the intersection point. We continue this process until we have exhausted all points in A, resulting in a spanning tree of A with all paths leading to the root y.

For a point $u \in A$, let $z_1 \in \mathcal{R}_0$ be the first step on the path from u to y in the spanning tree constructed above. If $C \subset \mathcal{R}$ and $z_1 \in C$, define $\mathfrak{t}_{u,C}$ to be a Dirac mass at z_1 . Otherwise, we define $\mathfrak{t}_{u,C}$ to be a Dirac mass at the element in C with the smallest \mathfrak{z} value. Note that this second case is irrelevant because we know that $V(T_u\omega, z_1) + B(u, z_1) = 0$.

Next, consider the case when A is infinite. In this scenario, we will construct a spanning forest that contains semi-infinite coalescing paths. We will provide a brief overview of the construction, as it is similar to the one described above for the case of a finite A.

We begin by selecting the point $u \in A$ with the smallest \mathfrak{z}' value. Since A is infinite, we can find a sequence of points $x_n \in A$ such that $|x_n|_1 \ge n$ for each n. For every n, there exists an admissible path from u to x_n that does not return to u after leaving it. By restricting to a subsequence n_j if necessary, we can ensure that all paths from u to x_{n_j} pass through the same step $z_1 \in \mathcal{R}_0$. If there are multiple options, we choose the z_1 with the smallest \mathfrak{z} value.

Continuing the construction, we obtain a semi-infinite path that starts at u, remains entirely within A, and contains no loops. Once this path is constructed, we select a point $v \in A$ that is not on the path and construct another semi-infinite path that remains within A. If this second path intersects with the first one, we stop its construction at that intersection point. We repeat this process until all points in A have been exhausted.

Using the spanning forest we constructed, we define the tie-breaker \mathfrak{t}_x for points $x \in A$ similarly to when A is finite.

The shift-covariance of t is immediately evident from the construction and the shift-covariance of B. Additionally, the construction guarantees the following properties hold with $Q_x^{\infty,B,t,\omega}$ -probability one:

- (i) If $X_{0:\infty}$ enters a finite equivalence class A, it will exit A without forming a loop inside it.
- (ii) Once $X_{0:\infty}$ exits an equivalence class A, it cannot return to A because doing so would create a loop that remains entirely within A.
- (iii) If $X_{0:\infty}$ enters an infinite equivalence class A, it remains within A and does not form any loops inside A.
- (iv) By Lemma 5.16, the path does not form loops that include non-essential points.

These properties imply that, almost surely, under $Q_x^{\infty,B,\mathfrak{t},\omega}$, $X_{0:\infty}$ does not revisit any previously visited points, and as a result, $|X_n|_1$ tends to infinity.

Lemma 5.17. Assume $\mathcal{A} \in \mathbb{A}$ is such that $\mathbf{0} \in \operatorname{ri} \mathcal{U}_{\mathcal{A}}$. Fix an $\omega \in \Omega$. Assume that $V(T_v \omega, z) \geq 0$ for all $v \in \mathcal{G}_{\mathcal{A}}$ and $z \in \mathcal{R}_{\mathcal{A}}$. Assume that there exist a $k \in \mathbb{Z}_{>0}$ and a loop $x_{0:k} \in \mathbb{X}^k_{\mathbf{0},\mathbf{0}}$ with $V(T_{x_i}\omega, x_{i+1} - x_i) = 0$ for each $i \in [0, k)$. Then $B(x, y) = F^{\infty}_{x,\mathbf{0}}(\omega) - F^{\infty}_{y,\mathbf{0}}(\omega)$, $x, y \in \mathcal{G}_{\mathcal{A}}$, is an ∞ -recovering cocycle. For any tie-breaker \mathfrak{t} ,

$$\mathbf{Q}_{\mathbf{0}}^{\infty,B,\mathfrak{t},\omega}\left\{V(T_{X_{i}}\omega,X_{i+1}-X_{i})=0 \text{ for all } i \in \mathbb{Z}_{\geq 0}\right\}=1.$$

Remark 5.18. This lemma shows that Condition 5.5 is violated when, with positive probability, a loop $x_{0:k}$ as in the statement of the lemma exists and there is a finite set of sites $y \in \mathcal{G}_{\mathcal{A}}$ for which there exist an $\ell \in \mathbb{Z}_{>0}$ and a path $y_{0:\ell} \in \mathbb{X}_{0,y}^{\ell}$, such that $V(T_{y_i}\omega, y_{i+1} - y_i) = 0$ for each $i \in [0, \ell)$. This situation can arise in various scenarios, including the standard first-passage percolation model described in Example 2.5(1), when the edge weights are i.i.d. and non-negative, and there exists a positive probability of encountering a zero edge weight. Nevertheless, Lemma 4.15 asserts that even in such cases, Condition 5.5 remains valid for the specific cocycles $B^{\mathcal{A},\beta,m}$ constructed in Theorem 4.5.

Proof of Lemma 5.17. The cocycle property is immediate from the definition of B. For the recovery property, write

$$\begin{split} \min_{z \in \mathcal{R}_{\mathcal{A}}} \{B(x, x+z) + V(T_x \omega, z)\} &= F_{x, \mathbf{0}}^{\infty}(\omega) - \max_{z \in \mathcal{R}_{\mathcal{A}}} \{F_{x+z, \mathbf{0}}^{\infty}(\omega) - V(T_x \omega, z)\} \\ &= F_{x, \mathbf{0}}^{\infty}(\omega) - \max_{z \in \mathcal{R}_{\mathcal{A}}} \sup_{k \in \mathbb{Z}_{\geq 0}} \sup_{x_{0:k} \in \mathbb{X}_{x+z, \mathbf{0}}^{k}} \left\{ \sum_{i=0}^{k-1} (-V(T_{x_i} \omega, x_{i+1} - x_i)) - V(T_x \omega, z) \right\} \\ &= F_{x, \mathbf{0}}^{\infty}(\omega) - \sup_{k \in \mathbb{Z}_{\geq 0}} \sup_{x_{0:k} \in \mathbb{X}_{x, \mathbf{0}}^{k}} \sum_{i=0}^{k-1} (-V(T_{x_i} \omega, x_{i+1} - x_i)). \end{split}$$

The right-hand side vanishes when $x \neq \mathbf{0}$ because, to go from x to $\mathbf{0}$, the path needs at least one step, and therefore, the supremum over k > 0 is the same as the supremum over $k \ge 0$, which gives $F_{x,\mathbf{0}}^{\infty}(\omega)$. When $x = \mathbf{0}$, both terms on the right-hand side vanish because of the assumptions that the potential is non-negative (and hence both terms are suprema of non-positive quantities) and that there exists an admissible loop from $\mathbf{0}$ to $\mathbf{0}$, of length at least one, which has a total weight of zero. Either way, the right-hand side vanishes, and B is an ∞ -recovering cocycle.

To see the second claim first note that Definition 5.3 implies that for any tie-breaker \mathfrak{t} ,

(5.8)
$$Q_{\mathbf{0}}^{\infty,B,\mathfrak{t},\omega} \left\{ B(X_i, X_{i+1}) + V(T_{X_i}\omega, X_{i+1} - X_i) = 0 \text{ for all } i \in \mathbb{Z}_{\geq 0} \right\} = 1.$$

Take $k \in \mathbb{Z}_{\geq 0}$ and $\varepsilon > 0$. There exists a path $Y_{k:\ell}^{\varepsilon} \in \mathbb{X}_{X_k,0}^{k,L}$ with $L \in \mathbb{Z}_{\geq k}$ and

$$\sum_{j=k}^{L-1} (-V(T_{Y_j^{\varepsilon}}\omega, Y_{j+1}^{\varepsilon} - Y_j^{\varepsilon})) \ge F_{X_k, \mathbf{0}}^{\infty}(\omega) - \varepsilon.$$

Rearranging and using the definition of B, we get

$$B(X_k, \mathbf{0}) + \sum_{j=k}^{L-1} V(T_{Y_j^{\varepsilon}}\omega, Y_{j+1}^{\varepsilon} - Y_j^{\varepsilon}) \leqslant \varepsilon.$$

Together with (5.8), the cocycle property, and $B(\mathbf{0},\mathbf{0}) = 0$, this gives that $Q_{\mathbf{0}}^{\infty,B,t,\omega}$ -almost surely,

$$\sum_{i=0}^{k-1} V(T_{X_i}\omega, X_{i+1} - X_i) + \sum_{j=k}^{L-1} V(T_{Y_j^{\varepsilon}}\omega, Y_{j+1}^{\varepsilon} - Y_j^{\varepsilon}) \leqslant \varepsilon.$$

Since the potential is non-negative, this implies

$$0 \leq \sum_{i=0}^{k-1} V(T_{X_i}\omega, X_{i+1} - X_i) \leq \varepsilon.$$

Taking $\varepsilon \to 0$ and using again the fact that the potential is non-negative gives that $V(T_{X_i}\omega, X_{i+1} - X_i) = 0$ for all $i \in [0, k)$. Since k was arbitrary, the lemma is proved.

Now, we address the directedness of the cocycle polymer measures. Recall the set $\mathcal{F}^{\beta}_{\mathbf{m}(B),\mathcal{A}}$ defined in (3.5).

Theorem 5.19. Assume the setting of Theorem 5.8. Let B be a β -recovering cocycle on the face \mathcal{A} with $\mathbf{m}(B)$ as in (4.3). If $\beta = \infty$, then let \mathfrak{t} be a covariant tie-breaker and, for the purpose of this theorem, abbreviate $\mathbf{Q}_{\mathbf{0}}^{\beta,B,\omega} = \mathbf{Q}_{\mathbf{0}}^{\beta,B,\mathfrak{t},\omega}$. Assume that $\mathbf{Q}_{\mathbf{0}}^{\beta,B,\omega}\{|X_n|_1 \to \infty\} = 1$, \mathbb{P} -almost surely. Then \mathbb{P} -almost surely for any $x \in \mathbb{Z}^d$,

$$\mathbf{Q}_{x}^{\beta,B,\omega}\left(X_{0:\infty} \text{ is directed into } \mathcal{F}_{\mathbf{m}(B),\mathcal{A}}^{\beta}\right) = 1.$$

Proof. By the *T*-covariance of *B* and the *T*-invariance of \mathbb{P} , we can take $x = \mathbf{0}$ without any loss of generality. Fix $\kappa > 0$. Let

$$\mathcal{F}^{\beta}_{\mathbf{m}(B),\mathcal{A},\kappa} = \{\zeta \in \mathcal{A} : |\zeta|_{1} = 1 \text{ and } \exists \xi \in \mathcal{F}^{\beta}_{\mathbf{m}(B),\mathcal{A}} \text{ with } |\zeta - \xi|_{1} < \kappa \}.$$

Note that \mathbb{P} -almost surely, for any $\zeta \in \mathcal{A}$, $\mathbf{m}(B) \cdot \zeta \ge \Lambda^{\beta}(\zeta)$ by [37, Theorem 2.14]. Since $\mathbf{m}(B) \cdot \zeta$ is continuous on all of \mathcal{A} , Λ^{β} is continuous on ri \mathcal{A} , and $\Lambda^{\beta,\text{usc}}_{\mathcal{A}}$ is the unique continuous extension of $\Lambda^{\beta}_{\mathcal{A}}$ to all of \mathcal{A} , we conclude that $\mathbf{m}(B) \cdot \zeta \ge \Lambda^{\beta,\text{usc}}_{\mathcal{A}}(\zeta)$. Therefore, with \mathbb{P} -probability one, for all $\zeta \in \mathcal{A}$ with $|\zeta|_1 = 1$,

 $\zeta \notin \mathcal{F}^{\beta}_{\mathbf{m}(B),\mathcal{A},\kappa}$ implies $\mathbf{m}(B) \cdot \zeta > \Lambda^{\beta,\mathrm{usc}}_{\mathcal{A}}(\zeta)$. Since $\{\zeta \in \mathcal{A} \setminus \mathcal{F}^{\beta}_{\mathbf{m}(B),\mathcal{A},\kappa} : |\zeta|_{1} = 1\}$ is a compact set and $\Lambda^{\beta,\mathrm{usc}}_{\mathcal{A}}$ is continuous on \mathcal{A} , we see that

$$\inf \left\{ \mathbf{m}(B) \cdot \zeta - \Lambda_{\mathcal{A}}^{\beta, \mathrm{usc}}(\zeta) : \zeta \in \mathcal{A} \backslash \mathcal{F}_{\mathbf{m}(B), \mathcal{A}, \kappa}^{\beta} \text{ with } |\zeta|_{1} = 1 \right\} > 0,$$

P-almost surely.

Fix any $\delta \in (0, 1)$. The above implies that there exists a deterministic $\varepsilon > 0$ such that

$$\mathbb{P}\Big\{\mathbf{m}(B)\cdot\zeta-\Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(\zeta)\geqslant 3\varepsilon\;\forall\zeta\in\mathcal{A}\backslash\mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta}\;\mathrm{with}\;|\zeta|_{1}=1\Big\}>1-\frac{\delta}{3}.$$

The shape theorem for B [38, Theorem B.3] implies that there exists an $L_0 > 0$ such that for any $L \ge L_0$ we have

$$\mathbb{P}\Big\{B(\mathbf{0}, y) \ge \mathbf{m}(B) \cdot y - \varepsilon |y|_1 \ \forall y \text{ with } |y|_1 \ge L\Big\} > 1 - \frac{\delta}{3}.$$

Applying (2.7) to $F_{0,y}^{\beta}$ we get that there exists an $L_1 \ge L_0$ such that

$$\mathbb{P}\Big\{F_{\mathbf{0},y}^{\beta} \leq \Lambda_{\mathcal{A}}^{\beta,\mathrm{usc}}(y) + \varepsilon |y|_1 \ \forall y \text{ with } |y|_1 \geq L_1\Big\} > 1 - \frac{\delta}{3}.$$

Putting all of the above together, we get

(5.9)
$$\mathbb{P}\left\{F_{\mathbf{0},y}^{\beta} - B(\mathbf{0},y) \leqslant -\varepsilon |y|_{1} \ \forall y \text{ with } |y|_{1} \geqslant L_{1} \text{ and } \frac{y}{|y|_{1}} \in \mathcal{A} \setminus \mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta}\right\} > 1 - \delta.$$

At zero temperature, Lemma 5.13 implies that, $Q_{\mathbf{0}}^{\infty,B,\omega}$ -almost surely, $X_{0:n}$ is an unrestricted-length geodesic, so $F_{\mathbf{0},X_n}^{\infty} = \sum_{i=0}^{n-1} V(T_{X_i}\omega, X_{i+1} - X_i) = \sum_{i=0}^{n-1} B(X_i, X_{i+1}) = B(\mathbf{0}, X_n)$. This and the assumption that $|X_n|_1 \to \infty$, $Q_{\mathbf{0}}^{\infty,B,\omega}$ -almost surely, imply that on the event in the above probability we have

(5.10)
$$Q_{\mathbf{0}}^{\infty,B,\omega} \Big\{ \exists n_0 : \frac{X_n}{|X_n|_1} \in \mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta} \; \forall n \ge n_0 \Big\} = 1.$$

To derive this at positive temperature, first observe that from (5.5),

$$Z_{\mathbf{0},y}^{\beta}e^{-\beta B(\mathbf{0},y)} = \mathbf{Q}_{\mathbf{0}}^{\beta,B,\omega}(\tau_y < \infty).$$

Thus, we have

$$\mathbb{P}\Big\{\mathbf{Q}_{\mathbf{0}}^{\beta,B,\omega}(\tau_{y}<\infty)\leqslant e^{-\beta\varepsilon|y|_{1}}\;\forall y \text{ with } |y|_{1}\geqslant L_{1} \text{ and } \frac{y}{|y|_{1}}\in\mathcal{A}\backslash\mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta}\Big\}>1-\delta.$$

Again, since $|X_n|_1 \to \infty$, $Q_x^{\beta,B,\omega}$ -almost surely, Borel-Cantelli implies that on the event in the above probability, (5.10) holds again.

Taking $\delta \to 0$ implies that (5.10) is a full P-probability event. This, in turn, implies that

$$\mathbb{P}\left\{\mathbf{Q}_{\mathbf{0}}^{\beta,B,\omega}\left(X_{0:\infty} \text{ is directed into } \overline{\mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta}}\right) = 1\right\} = 1,$$

where $\overline{\mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta}}$ is the closure of $\mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta}$. Since $\mathcal{F}_{\mathbf{m}(B),\mathcal{A}}^{\beta}$ is closed, $\bigcap_{\kappa>0} \overline{\mathcal{F}_{\mathbf{m}(B),\mathcal{A},\kappa}^{\beta}} = \mathcal{F}_{\mathbf{m}(B),\mathcal{A}}^{\beta}$. Taking $\kappa \to 0$ in the above display finishes the proof.

Next, we turn to the question of determining the possible limit points of X_n/n under the polymer measure. We achieve this by proving a large deviation principle. Fix a face $\mathcal{A} \in \mathbb{A}$. For each $\xi \in \mathcal{U}_{\mathcal{A}} \cap \mathbb{Q}^d$, let $\{y_n(\xi)\}_{n \in \mathbb{Z}_{\geq 0}}$ be a path with $y_0(\xi) = \mathbf{0}, y_n(\xi) - y_{n-1}(\xi) \in \mathcal{R}_{\mathcal{A}}$ for all $n \in \mathbb{Z}_{>0}$, and $y_{kj} = kj\xi$ for all $k \in \mathbb{Z}_{\geq 0}$ and some $j = j(\xi) \in \mathbb{Z}_{>0}$.

Lemma 5.20. Fix $\beta \in (0, \infty]$ and $\mathcal{A} \in \mathbb{A}$. Assume Condition 3.5 holds. Let B be an $L^1(\mathbb{P})$ T-covariant β -recovering cocycle on the face \mathcal{A} . For \mathbb{P} -a.e. ω , all subsets $K \subset \mathbb{R}^d$, and all $\delta > 0$, if $\beta < \infty$,

$$\underbrace{\lim_{n \to \infty} \frac{1}{n} \log \mathcal{E}_{\mathbf{0}} \left[e^{-\beta B(\mathbf{0}, X_n) - \beta \sum_{i=0}^{n-1} V(T_{X_i} \omega, X_{i+1} - X_i)} \mathbb{1} \{ X_n / n \in K \cap \mathcal{U}_{\mathcal{A}} \} \right]}_{\xi \in \mathbb{Q}^d \cap K_{\delta} \cap \mathcal{U}_{\mathcal{A}}} \underbrace{\lim_{n \to \infty} \frac{1}{n} \log \mathcal{E}_{\mathbf{0}} \left[e^{-\beta B(\mathbf{0}, X_n) - \beta \sum_{i=0}^{n-1} V(T_{X_i} \omega, X_{i+1} - X_i)} \mathbb{1} \{ X_n = y_n(\xi) \} \right]}_{\xi \in \mathbb{Q}^d \cap K_{\delta} \cap \mathcal{U}_{\mathcal{A}}}$$

where $K_{\delta} = \{\zeta \in \mathbb{R}^d : \exists \zeta' \in K \text{ with } |\zeta - \zeta'|_1 \leq \delta\}.$ If $\beta = \infty$,

$$\lim_{n \to \infty} \frac{1}{n} \max_{x \in D_n(\mathcal{R}_{\mathcal{A}}) \cap nK} \left(F_{\mathbf{0},x,n}^{\infty} - B(\mathbf{0},x) \right) \leq \sup_{\xi \in \mathbb{Q}^d \cap K_{\delta} \cap \mathcal{U}_{\mathcal{A}}} \lim_{n \to \infty} \frac{1}{n} \left(F_{\mathbf{0},y_n(\xi),n}^{\infty} - B(\mathbf{0},y_n(\xi)) \right).$$

Proof. We follow the strategy of the proof of Lemma 2.9 in [58]. Let $R = \max\{|z|_1 : z \in \mathcal{R}_A\}$. Order $\mathcal{R}^{\mathrm{id}}_{\mathcal{A}}$ as $\{z_1, \ldots, z_m\}$ and for each of these z_i let $\hat{z}_i \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}^{\mathrm{id}}_{\mathcal{A}}$ be as in Condition 3.5. If $\mathcal{R}^{\mathrm{id}}_{\mathcal{A}} = \emptyset$, then set m = 1 and take any $\hat{z}_1 \in \mathcal{R}_{\mathcal{A}}$. Let $\varepsilon \in (0, \delta/(4R|\mathcal{R}_{\mathcal{A}}|))$. Let j be an integer with $j \ge \max\{\frac{1+2\varepsilon}{\varepsilon}|\mathcal{R}|, (2\varepsilon)^{-1}\}$. Because $(mj)^{-1}D_{mj}(\mathcal{R}_{\mathcal{A}})$ is finite, there exists an integer b such that $y_{kb} = kb\xi$ for all $k \in \mathbb{Z}_{\ge 0}$ and $\xi \in (mj)^{-1}D_{mj}(\mathcal{R}_{\mathcal{A}})$.

Next, we construct a path from each $x \in D_n(\mathcal{R}_A) \cap nK$ to a multiple of a point $\xi(n,x) \in K_\delta \cap (mj)^{-1}D_{mj}(\mathcal{R}_A)$. To do this, start by writing $x = \sum_{z \in \mathcal{R}_A} a_z z$ where each $a_z \in \mathbb{Z}_{\geq 0}$ and $\sum_{z \in \mathcal{R}_A} a_z = n$. Let $k_n = \lfloor \frac{(1+2\varepsilon)n}{mj} \rfloor$ and $s_z^{(n)} = \lfloor \frac{ja_z}{(1+2\varepsilon)n} \rfloor$. Then

$$\frac{a_z}{n} - \frac{s_z^{(n)}}{j} = \frac{a_z}{n} \left(1 - \frac{ns_z^{(n)}}{ja_z} \right) \ge \frac{a_z}{n} \left(1 - \frac{1}{1+2\varepsilon} \right) - \frac{1}{j}$$

Similarly,

$$\frac{a_z}{n} - \frac{s_z^{(n)}}{j} \leqslant \frac{a_z}{n} \left(1 - \frac{1}{1 + 2\varepsilon}\right)$$

Summing over z, we get

(5.11)
$$\frac{\varepsilon}{1+2\varepsilon} \leqslant 1 - \frac{1}{1+2\varepsilon} - \frac{|\mathcal{R}|}{j} \leqslant 1 - \frac{1}{j} \sum_{z \in \mathcal{R}_{\mathcal{A}}} s_z^{(n)} \leqslant 1 - \frac{1}{1+2\varepsilon} < \frac{\delta}{2R|\mathcal{R}_{\mathcal{A}}|} \leqslant \frac{\delta}{2R}$$

Also,

$$|n^{-1}a_z - j^{-1}s_z^{(n)}| \le \max\left\{\frac{a_z}{n}\left(1 - \frac{1}{1+2\varepsilon}\right), \frac{1}{j}\right\} \le 2\varepsilon.$$

Thus,

$$(5.12) \qquad \left|\frac{1}{j}\sum_{z\in\mathcal{R}_{\mathcal{A}}}s_{z}^{(n)}z-\frac{x}{n}\right|_{1} = \left|\frac{1}{j}\sum_{z\in\mathcal{R}_{\mathcal{A}}}\left(s_{z}^{(n)}z-\frac{j}{n}a_{z}z\right)\right|_{1} \leqslant R\sum_{z\in\mathcal{R}_{\mathcal{A}}}\left|j^{-1}s_{z}^{(n)}-n^{-1}a_{z}\right| \leqslant 2R|\mathcal{R}_{\mathcal{A}}|\varepsilon<\frac{\delta}{2}.$$

Define

(5.13)
$$\xi(n,x) = j^{-1} \sum_{z \in \mathcal{R}_{\mathcal{A}}} s_z^{(n)} z + \sum_{i=1}^m m^{-1} \left(1 - j^{-1} \sum_{z \in \mathcal{R}_{\mathcal{A}}} s_z^{(n)} \right) \hat{z}_i.$$

(5.11) and (5.12) imply that $\xi(n,x) \in K_{\delta} \cap (mj)^{-1}D_{mj}(\mathcal{R}_{\mathcal{A}})$ and the path that takes $ms_z^{(n)}$ z-steps for each $z \in \mathcal{R}_{\mathcal{A}}$ then $(j - \sum_{z \in \mathcal{R}_{\mathcal{A}}} s_z^{(n)})$ \hat{z}_i -steps, for each $i \in \{1, \ldots, m\}$, is an admissible path that takes a total of mj steps to go from **0** to $mj\xi(n,x)$.

Consider next the following admissible path starting at the origin **0**. It begins by taking a_z z-steps for each $z \in \mathcal{R}_A$, leading to the point x. Next, the path takes $(mk_n s_z^{(n)} - a_z)$ z-steps for each $z \in \mathcal{R}_A$. Then, it proceeds with $(k_n j - \sum_{z \in \mathcal{R}_A} k_n s_z^{(n)})$ \hat{z}_i -steps, for each $i \in \{1, \ldots, m\}$. This entire path is admissible because, first, $mk_n s_z^{(n)} \ge m \cdot \frac{(1+2\varepsilon)n}{mj} \cdot \frac{ja_z}{(1+2\varepsilon)n} = a_z$ and second, as previously observed, $k_n j \ge \sum_{z \in \mathcal{R}_A} k_n s_z^{(n)}$.

The above path consists of a total of $k_n m j$ steps and ends at $k_n m j \xi(n, x)$. Notably, the last $k_n m j - n$ steps give an admissible path from x to $k_n m j \xi(n, x)$. For each $i \in \{1, \ldots, m\}$, the number of \hat{z}_i steps in the path is at least

$$k_n j \left(1 - j^{-1} \sum_{z \in \mathcal{R}_{\mathcal{A}}} s_z^{(n)} \right) \ge \frac{(1 + 2\varepsilon)n}{mj} \cdot j \cdot \frac{\varepsilon}{1 + 2\varepsilon} = nm^{-1}\varepsilon.$$

Let $\ell_n = \lceil \frac{k_n}{b} \rceil$ so that $(\ell_n - 1)b < k_n \leq \ell_n b$. Repeating the steps of the path below (5.13) that goes from **0** to $mj\xi(n,x)$, one gets a path that goes from $k_n mj\xi(n,x)$ to $\ell_n bmj\xi(n,x)$. This requires repeating each step $\ell_n b - k_n < b$ times. Thus, the number of steps to go from x to $\ell_n bmj\xi(n,x)$ is $r_n = (k_n mj - n) + (\ell_n b - k_n)mj = \ell_n bmj - n$. Since b is a function solely dependent on mj and independent of n,

$$r_n \leqslant \left(\frac{k_n}{b} + 1\right) bmj - n \leqslant \left(\frac{(1+2\varepsilon)n}{mj} + 1\right) mj + bmj - n = 2\varepsilon n + (b+1)mj \leqslant 3\varepsilon n$$

for all $n \ge n_0$ with n_0 depending on a mj and ε . Denote the steps we constructed to go from x to $\ell_n bmj\xi(n, x)$ by (u_1, \ldots, u_{r_n}) and denote by $\mathbf{u}(n, x)$ the path that starts at $\mathbf{0}$ and takes these exact steps.

We are ready to estimate:

$$\begin{split} &\frac{1}{n}\log \mathbf{E_0}\left[e^{-\beta B(\mathbf{0},X_n)-\beta\sum_{i=0}^{n-1}V(T_{X_i}\omega,X_{i+1}-X_i)}\mathbbm{1}\{X_n/n\in K\cap\mathcal{U}_{\mathcal{A}}\}\right]\\ &=\frac{1}{n}\log\sum_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\mathbf{E_0}\left[e^{-\beta B(\mathbf{0},x)-\beta\sum_{i=0}^{n-1}V(T_{X_i}\omega,X_{i+1}-X_i)}\mathbbm{1}\{X_n=x\}\right]\\ &\leqslant \max_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\frac{1}{n}\log \mathbf{E_0}\left[e^{-\beta B(\mathbf{0},x)-\beta\sum_{i=0}^{n-1}V(T_{X_i}\omega,X_{i+1}-X_i)}\mathbbm{1}\{X_n=x\}\right]+\frac{C\log n}{n}\\ &\leqslant \max_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\frac{1}{n}\log \mathbf{E_0}\left[e^{-\beta B(\mathbf{0},\ell_n bmj\xi(n,x))-\beta\sum_{i=0}^{n+rn-1}V(T_{X_i}\omega,X_{i+1}-X_i)}\mathbbm{1}\{X_n=x,Z_{n+1:r_n+n}=u_{1:r_n}\}\right]\\ &-\frac{r_n}{n}\log\min_{z\in\mathcal{R}_{\mathcal{A}}}p(z)+\max_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\frac{\beta B(x,\ell_n bmj\xi(n,x))}{n}\\ &+\max_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\frac{1}{n}\log \mathbf{E_0}\left[e^{-\beta B(\mathbf{0},\ell_n bmj\xi(n,x))-\beta\sum_{i=0}^{n+rn-1}V(T_{X_i}\omega,X_{i+1}-X_i)}\mathbbm{1}\{X_{\ell_n bmj}=\ell_n bmj\xi(n,x)\}\right]\\ &\leqslant \max_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\frac{1}{n}\log \mathbf{E_0}\left[e^{-\beta B(\mathbf{0},\ell_n bmj\xi(n,x))-\beta\sum_{i=0}^{n+rn-1}V(T_{X_i}\omega,X_{i+1}-X_i)}\mathbbm{1}\{X_{\ell_n bmj}=\ell_n bmj\xi(n,x)\}\right]\\ &+\max_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\frac{1}{n}\sum_{i=0}^{r_n-1}V^+(T_{x+u_1+\ldots+u_i}\omega,u_{i+1})+\frac{C\log n}{n}\\ &\leqslant \max_{x\in D_n(\mathcal{R}_{\mathcal{A}})\cap nK}\frac{1}{n}\log \mathbbm{1}\sum_{i=0}^{r_n-1}V^+(T_{x+u_1+\ldots+u_i}\omega,u_{i+1})+\frac{C\log n}{n}. \end{split}$$

Above, we used the notation $Z_{i+1} = X_{i+1} - X_i$ and took the convention that an empty sum is 0. Hence, $x + u_1 + \ldots + u_i = x$ when i = 0.

We now explain why all of the quantities on the right-hand side converge to 0 when we send $n \to \infty$ and then $\varepsilon \to 0$. Since $\xi(n, x) \in K_{\delta} \cap (mj)^{-1} D_{mj}(\mathcal{R}_{\mathcal{A}}) \subset \mathbb{Q}^d \cap K_{\delta} \cap \mathcal{U}_{\mathcal{A}}$ and $y_{\ell_n bmj}(\xi(n, x)) = \ell_n bmj\xi(n, x)$, we have that for n large enough, the first maximum on the right-hand side is bounded above by

$$(1+3\varepsilon)\sup_{\xi\in\mathbb{Q}^d\cap K_\delta\cap\mathcal{U}_{\mathcal{A}}}\overline{\lim_{n\to\infty}}\frac{1}{n}\log \mathcal{E}_{\mathbf{0}}\left[e^{-\beta B(\mathbf{0},X_n)-\beta\sum_{i=0}^{n-1}V(T_{X_i}\omega,X_{i+1}-X_i)}\mathbb{1}\{X_n=y_n(\xi)\}\right].$$

For the second term, use $r_n \leq 3\varepsilon n$. By the shape theorem for cocycles, [38, Theorem B.3], in the limit, the next maximum is bounded in absolute value by $\beta(1 + 3\varepsilon)3\varepsilon R|\mathbf{m}(B)|_1$, where $\mathbf{m}(B)$ is the random vector which satisfies (4.3). The last term goes to 0.

For the third maximum, notice that the particular order of the steps in $\mathbf{u}(n, x)$ does not matter to this point. For each $i \in \{1, \ldots, m\}$, the ratio of z_i steps to \hat{z}_i steps is at most $r_n/(m^{-1}n\varepsilon) \leq 3m$. So order the steps of $\mathbf{u}(n, x)$ in this way. First, alternate between \hat{z}_1 steps and z_1 steps in such a way that there are no more than 3m consecutive z_1 steps. Continue this process until we have exhausted all z_1 steps and their corresponding $k_n j \left(1 - j^{-1} \sum_{z \in \mathcal{R}_A} s_z^{(n)}\right) \hat{z}_1$ steps. Repeat this procedure for $i = 2, \ldots, m$. It is important to note that if $\hat{z}_i = \hat{z}_{i'}$ for some distinct i and i', we only exhaust the corresponding number of \hat{z}_i steps when we exhaust the z_i steps, leaving the $\hat{z}_{i'}$ steps to be paired with the $z_{i'}$ steps.

Next, choose an ordering for the set $\mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_{\mathcal{A}}^{\text{id}}$, denoted as $\{z'_1, \ldots, z'_c\}$. Continue ordering the steps of $\mathbf{u}(n, x)$ by using all of the z'_1 steps first, followed by the z'_2 steps, and so on for the remaining steps. Note that this includes $(mk_n s_{\hat{z}_i}^{(n)} - a_{\hat{z}_i}) \hat{z}_i$ steps, for each $i \in \{1, \ldots, m\}$ from the first part of the path from x to $k_n m j \xi(n, x)$, which have not been used in the above part of the procedure, when we exhausted the z_i steps.

Recall that T_z is the identity map for all $z \in \mathcal{R}^{id}_{\mathcal{A}}$. Using the above ordering, we may bound

$$\max_{x \in D_n(\mathcal{R}_{\mathcal{A}}) \cap nK} \frac{1}{n} \sum_{k=0}^{r_n - 1} V^+ (T_{x+u_1+\ldots+u_k}\omega, u_{k+1}) \leq \frac{|\mathcal{R}_{\mathcal{A}}|}{n} \max_{x \in D_n(\mathcal{R}_{\mathcal{A}}) \cap nK} \max_{y \in x+\mathbf{u}(n,x)} \max_{z \in \mathcal{R}_{\mathcal{A}} \setminus \mathcal{R}_{\mathcal{A}}^{\mathrm{id}}} \sum_{k=0}^{r_n - 1} V^+ (T_{y+kz}\omega, z) + \sum_{i=1}^m \frac{3m}{n} \max_{x \in D_n(\mathcal{R}_{\mathcal{A}}) \cap nK} \max_{y \in x+\mathbf{u}(n,x)} \sum_{k=0}^{r_n - 1} V^+ (T_{y+k\hat{z}_i}\omega, z_i).$$

Since $V^+(\omega, z) \in \mathcal{L}_{z,\mathcal{R}_{\mathcal{A}}}$ for all $z \in \mathcal{R}_{\mathcal{A}}$ and $V^+(\omega, z_i) \in \mathcal{L}_{\hat{z}_i,\mathcal{R}_{\mathcal{A}}}$, for each $i \in \{1, \ldots, m\}$, the right-hand side converges to 0 when we take $n \to \infty$ and then $\varepsilon \to \infty$.

At zero temperature, the equivalent estimate is

$$\frac{1}{n} \max_{x \in D_n(\mathcal{R}_A) \cap nK} \left(F_{\mathbf{0},x,n}^{\infty} - B(\mathbf{0},x) \right) \\
\leq \frac{1}{n} \max_{x \in D_n(\mathcal{R}_A) \cap nK} \left(F_{\mathbf{0},x,n}^{\infty} - \sum_{i=0}^{r_n-1} V(T_{x+u_1+\ldots+u_i}\omega, u_{i+1}) - B(\mathbf{0}, \ell_n bmj\xi(n,x)) \right) \\
+ \max_{x \in D_n(\mathcal{R}_A) \cap nK} \frac{B(x, \ell_n bmj\xi(n,x))}{n} + \max_{x \in D_n(\mathcal{R}_A) \cap nK} \frac{1}{n} \sum_{i=0}^{r_n-1} V^+(T_{x+u_1+\ldots+u_i}\omega, u_{i+1}) \\
\leq \frac{1}{n} \max_{x \in D_n(\mathcal{R}_A) \cap nK} \left(F_{\mathbf{0},\ell_n bmj\xi(n,x),\ell_n bmj}^{\infty} - B(\mathbf{0}, \ell_n bmj\xi(n,x)) \right) \\
+ \max_{x \in D_n(\mathcal{R}_A) \cap nK} \frac{B(x, \ell_n bmj\xi(n,x))}{n} + \max_{x \in D_n(\mathcal{R}_A) \cap nK} \frac{1}{n} \sum_{i=0}^{r_n-1} V^+(T_{x+u_1+\ldots+u_i}\omega, u_{i+1}).$$

The two terms on the last line converge to 0 after sending $n \to \infty$ and $\varepsilon \to 0$, as in the positive temperature case. The term in the second-to-last line is bounded above by

$$(1+3\varepsilon)\sup_{\xi\in\mathbb{Q}^d\cap K_{\delta}\cap\mathcal{U}_{\mathcal{A}}}\overline{\lim_{n\to\infty}}\frac{1}{n}\left(F^{\infty}_{\mathbf{0},y_n(\xi),n}-B(\mathbf{0},y_n(\xi))\right).$$

Take $\varepsilon \to 0$.

(5.16)

Recall the restricted-length limiting quenched point-to-point free energy $\Lambda_{\rm res}^{\beta}$ in (2.4), its restriction $\Lambda_{\mathcal{A},\rm res}^{\beta}$ to the face \mathcal{A} , and the upper semicontinuous regularization $\Lambda_{\mathcal{A},\rm res}^{\beta,\rm usc}$. Define

$$I_B(\xi) = \begin{cases} \beta \mathbf{m}(B) \cdot \xi - \beta \Lambda_{\mathcal{A}, \mathrm{res}}^{\beta, \mathrm{usc}}(\xi) & \text{if } \xi \in \mathcal{U}_{\mathcal{A}}, \\ \infty & \text{otherwise.} \end{cases}$$

Theorem 5.21. Assume the setting of Theorem 5.8 with $\beta < \infty$. There exists an event $\Omega_B \subset \Omega$ with $\mathbb{P}(\Omega_B) = 1$ such that for each $\omega \in \Omega_B$, the distribution of X_n/n under $Q_x^{\beta,B,\omega}$ satisfies a large deviation principle with rate function I_B . This means the following bounds hold

(5.14)
$$\overline{\lim}_{n \to \infty} n^{-1} \log \mathbf{Q}_x^{\beta, B, \omega}(X_n/n \in K) \leqslant -\inf_{\zeta \in K} I_B(\zeta) \text{ for closed } K \subset \mathbb{R}^d \quad and$$

(5.15)
$$\lim_{n \to \infty} n^{-1} \log \mathbf{Q}_x^{\beta,B,\omega}(X_n/n \in O) \ge -\inf_{\zeta \in O} I_B(\zeta) \text{ for open } O \subset \mathbb{R}^d.$$

Proof. We follow the strategy of the proof of [58, Theorem 4.1]. First, observe that for $y - x \in \mathcal{G}_{\mathcal{A}}^+$, using the cocycle property, we get

$$Q_x^{\beta,B,\omega}(X_n = y) = \sum_{x_{0:n} \in \mathbb{X}_{x,y}^n} p(x_{0:n}) e^{-\beta \sum_{i=0}^{n-1} B(x_i, x_{i+1}, \omega) - \beta \sum_{i=0}^{n-1} V(T_{x_i}, \omega, x_{i+1}, -x_i)}$$
$$= e^{-\beta B(x,y,\omega)} \sum_{x_{0:n} \in \mathbb{X}_{x,y}^n} p(x_{0:n}) e^{-\beta \sum_{i=0}^{n-1} V(T_{x_i}, \omega, x_{i+1}, -x_i)}$$
$$= e^{-\beta B(x,y,\omega)} Z_{x,y,n}^{\beta,\omega}.$$

Then using [38, Theorem B.3] and Theorem 2.17, we have for \mathbb{P} -almost every ω , for any sequence $x_n \in D_n(\mathcal{R}_A) + x$, $n \in \mathbb{Z}_{>0}$, such that $x_n/n \to \xi \in \mathrm{ri}\,\mathcal{U}_A$ as $n \to \infty$,

$$\frac{1}{n}\log \mathcal{Q}_x^{\beta,B,\omega}(X_n=x_n) = -\frac{1}{n}\beta B(x,x_n,\omega) + \frac{1}{n}\log Z_{x,x_n,n}^{\beta,\omega} \xrightarrow[n\to\infty]{} -\beta \mathbf{m}(B)\cdot\xi + \beta\Lambda_{\mathcal{A},\mathrm{res}}^{\beta,\mathrm{usc}}(\xi) = -I_B(\xi).$$

Let $O \subset \mathbb{R}^d$ be an open set. Take $\xi \in O \cap \mathrm{ri} \mathcal{U}_{\mathcal{A}}$ and take x_n as above. Then $x_n \in nO$ for all n large enough and

$$\underbrace{\lim_{n \to \infty} \frac{1}{n} \log \mathbf{Q}_x^{\beta, B, \omega}(X_n/n \in O)}_{n \to \infty} \ge \underbrace{\lim_{n \to \infty} \frac{1}{n} \log \mathbf{Q}_x^{\beta, B, \omega}(X_n = x_n)}_{n \to \infty} = -I_B(\xi).$$

By Theorem 2.17, I_B is continuous on $\mathcal{U}_{\mathcal{A}}$ and, therefore, the bound, in fact, holds for all $\xi \in O \cap \mathcal{U}_{\mathcal{A}}$. This bound also holds trivially for $\xi \in O \setminus (\mathcal{U}_{\mathcal{A}})$. Take the supremum over all $\xi \in O$ to get (5.15).

Next, Lemma 5.20 says that for \mathbb{P} -almost every ω , for any closed set $K \subset \mathbb{R}^d$ and $\delta > 0$, for any $x \in \mathbb{Z}^d$,

$$\underbrace{\lim_{n \to \infty} \frac{1}{n} \log Q_x^{\beta, B, \omega} \left(\frac{X_n}{n} \in K \right)}_{n \to \infty} = \underbrace{\lim_{n \to \infty} \frac{1}{n} \log E_0 \left[e^{-\beta B(\mathbf{0}, X_n) - \beta \sum_{i=0}^{n-1} V(T_{X_i} \omega, X_{i+1} - X_i)} \mathbb{1} \left\{ \frac{X_n}{n} \in K \cap \mathcal{U}_{\mathcal{A}} \right\} \right] \\ \leqslant \sup_{\xi \in \mathbb{Q}^d \cap K_\delta \cap \mathcal{U}_{\mathcal{A}}} \underbrace{\lim_{n \to \infty} \frac{1}{n} \log Q_x^{\beta, B, \omega} (X_n = y_n(\xi))}_{n \to \infty} \\ = - \inf_{\xi \in \mathbb{Q}^d \cap K_\delta \cap \mathcal{U}_{\mathcal{A}}} I_B(\xi) \leqslant - \inf_{\xi \in K_\delta \cap \mathcal{U}_{\mathcal{A}}} I_B(\xi).$$
(5.17)

Since (5.17) holds for all $\delta > 0$, $K_{\delta} \cap \mathcal{U}_{\mathcal{A}}$ is compact and decreases to $K \cap \mathcal{U}_{\mathcal{A}}$ as δ decreases to 0, and since I_B is lower semicontinuous, we obtain the bound

$$\overline{\lim_{n \to \infty}} \frac{1}{n} \log \mathcal{Q}_x^{\beta, B, \omega} \left(\frac{X_n}{n} \in K \right) \leq -\lim_{\delta \to 0} \inf_{\xi \in K_\delta \cap \mathcal{U}_A} I_B(\xi) = -\inf_{\xi \in K \cap \mathcal{U}_A} I_B(\xi) = -\inf_{\xi \in K} I_B(\xi).$$

We are now ready to prove Theorem 5.8.

Proof of Theorem 5.8. Parts (a), (b), (c), (d), (f), (g), and (h) were proved in Lemmas 5.10, 5.11, 5.12, 5.13, 5.14, 5.15, and Theorem 5.19, respectively. Next, we will prove part (e) in the positive temperature case.

Write the complement $\mathbb{R}^d \setminus \mathcal{U}^{\beta}_{\mathbf{m}(B),\mathcal{A}}$ as a countable union of non-decreasing compact sets K_j . Explicitly, we define

$$K_j = \{ v \in \mathbb{R}^d : |v|_1 \leq j \text{ and } d(v, \mathcal{U}^{\beta}_{\mathbf{m}(B), \mathcal{A}}) \geq j^{-1} \}.$$

Then in fact $\mathbb{R}^d \setminus \mathcal{U}^{\beta}_{\mathbf{m}(B),\mathcal{A}} = \bigcup_{j=1}^{\infty} \mathring{K}_j$, where \mathring{A} is the interior of the set $A \subset \mathbb{R}^d$.

We now show that for each j, $Q_x^{\beta,B,\omega}(X_n/n \text{ has no limit points in } \mathring{K}_j) = 1$. From the large deviation principle in Theorem 5.21,

$$\overline{\lim_{n \to \infty}} n^{-1} \log \mathbf{Q}_x^{\beta,B,\omega}(X_n/n \in K_j) \leqslant -\inf_{\zeta \in K_j} I_B(\zeta).$$

The rate function $I_B(\zeta) = 0$ precisely when $\zeta \in \mathcal{U}_{\mathbf{m}(B),\mathcal{A}}^{\beta}$. Because K_j is compact, $K_j \subset \mathbb{R}^d \setminus \mathcal{U}_{\mathbf{m}(B),\mathcal{A}}^{\beta}$, and I_B is continuous, we see that $-\inf_{\zeta \in K_j} I_B(\zeta) < 0$. Therefore $\mathbb{Q}_x^{\beta,B,\omega}(X_n/n \in K_j)$ is summable in n, and by Borel-Cantelli, $\mathbb{Q}_x^{\beta,B,\omega}(X_n/n \in K_j$ for only finitely many n) = 1. Therefore, X_n/n has no limit points in \mathring{K}_j , $\mathbb{Q}_x^{\beta,B,\omega}$ -almost surely.

Then by a union bound in j, since K_j are non-decreasing, we see $Q_x^{\beta,B,\omega}$ -almost surely, X_n/n has no limit points in $\bigcup_{j=1}^{\infty} \mathring{K_j} = \mathbb{R}^d \setminus \mathcal{U}_{\mathbf{m}(B),\mathcal{A}}^{\beta}$. So we conclude that $Q_x^{\beta,B,\omega}(X_n \text{ is directed into } \mathcal{U}_{\mathbf{m}(B),\mathcal{A}}^{\beta}) = 1$.

Lastly, we part (e) in the zero temperature case. For any sequence $x_n \in D_n(\mathcal{R}_{\mathcal{A}}) + x$, $n \in \mathbb{Z}_{>0}$, such that $x_n/n \to \xi \in \mathrm{ri}\,\mathcal{U}_{\mathcal{A}}$ as $n \to \infty$, the cocycle and recovery properties imply that $n^{-1}B(x, x_n) \ge n^{-1}F_{x,x_n,n}^{\infty}$. The shape theorem [38, Theorem B.3] and Theorem 2.17 give that $\mathbf{m}(B) \cdot \xi \ge \Lambda_{\mathcal{A},\mathrm{res}}^{\infty,\mathrm{usc}}(\xi)$, for all $\xi \in \mathrm{ri}\,\mathcal{U}_{\mathcal{A}}$.

Consider X_n under $Q_x^{\infty,B,t,\omega}$. Using $n^{-1}B(x,X_n) = n^{-1}F_{x,X_n,n}^{\infty}$, the shape theorem [38, Theorem B.3], and Theorem 2.17, we get $\mathbf{m}(B) \cdot \zeta \leq \Lambda_{\mathcal{A}, \text{res}}^{\infty, \text{usc}}(\zeta)$, for any limit point ζ of X_n/n . Combined with the inequality above, we see $\mathbf{m}(B) \cdot \zeta = \Lambda_{\mathcal{A}, \text{res}}^{\infty, \text{usc}}(\zeta)$, which then implies that $\zeta \in \mathcal{U}_{\mathbf{m}(B), \mathcal{A}}^{\infty}$.

6. Proof of Theorem 3.14

Apply Theorem 4.5 to obtain, for each $\mathcal{A} \in \mathbb{A}$, $(\beta, m) \in \mathcal{D}_{\mathcal{A}}$, and $x \in \mathbb{Z}^d$, the $L^1(\widehat{\mathbb{P}})$, \widehat{T} -covariant, β -recovering cocycle $B^{\mathcal{A},\beta,m}(x,y,\hat{\omega})$ on \mathcal{A} with mean $\widehat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})] = m$. Next, apply Theorem 5.8 to obtain the semi-infinite path measures $Q_x^{\mathcal{A},\beta,B^{\mathcal{A},\beta,m},\hat{\omega}}$ for triples (\mathcal{A},β,m) with $\beta < \infty$. For the triples (\mathcal{A},∞,m) , Lemma 4.15 verifies that Condition 5.5 holds for $B^{\mathcal{A},\infty,m}$. Therefore, Theorem 5.8 produces the measures $Q_x^{\mathcal{A},\infty,B^{\mathcal{A},\infty,m},\mathfrak{t},\hat{\omega}}$, where \mathfrak{t} is the tie-breaker mentioned in part (g) of the theorem. We will abbreviate this family of measures as $\{Q_x^{\mathcal{A},\beta,m,\hat{\omega}} : \mathcal{A} \in \mathbb{A}, (\beta,m) \in \mathcal{D}_{\mathcal{A}}, x \in \mathbb{Z}^d\}$. Let $\widehat{\Omega}_0 = \widehat{\Omega}_{coc} \cap \bigcap_{\mathcal{A} \in \mathbb{A}, (\beta,m) \in \mathcal{D}_{\mathcal{A}}, \widehat{\Omega}, m},$ where $\widehat{\Omega}_{coc}$ is the full-probability event from Theorem 4.5(b) and $\widehat{\Omega}_{B^{\mathcal{A},\beta,m}}$ are the full-probability events from Theorem 5.8. Then the measures $Q_x^{\mathcal{A},\beta,m,\hat{\omega}}$ satisfy the consistency properties in parts (a), (b), (c), and (d), for all $\hat{\omega} \in \widehat{\Omega}_0$.

By parts (f), (g), and (h) of Theorem 5.8, we have that under the conditions of Theorem 3.14(e), for all $x \in \mathbb{Z}^d$ and $\hat{\omega} \in \hat{\Omega}_0$,

$$\mathbf{Q}_{x}^{\mathcal{A},\beta,m,\hat{\omega}}\{|X_{n}|_{1} \to \infty\} = \mathbf{Q}_{x}^{\mathcal{A},\beta,m,\hat{\omega}}\{X_{0:\infty} \text{ is directed into } \mathcal{F}_{\mathbf{m}(B^{\mathcal{A},\beta,m})}^{\beta}\} = 1.$$

(Recall that Lemma 4.15 verifies that Condition 5.5 holds for $B^{\mathcal{A},\infty,m}$.) By Theorem 5.8(e), for all $x \in \mathbb{Z}^d$ and $\hat{\omega} \in \hat{\Omega}_0$,

 $Q_x^{\mathcal{A},\beta,m,\hat{\omega}}$ {all limit points of X_n/n are contained in $\mathcal{U}^{\beta}_{\mathbf{m}(B^{\mathcal{A},\beta,m})}$ } = 1.

By Theorem 5.8(c) and (3.8), $\hat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})] = m \in \operatorname{ext}(W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta,\operatorname{usc}}(\xi))$ for some $\xi \in \operatorname{ri} \mathcal{A}$. By Lemma 4.4, $\mathbf{m}(B^{\mathcal{A},\beta,m}) \in W_{\mathcal{A}} \cap \partial \Lambda_{\mathcal{A}}^{\beta,\operatorname{usc}}(\xi)$, $\hat{\mathbb{P}}$ -almost surely. It must be then that $m = \hat{\mathbb{E}}[\mathbf{m}(B^{\mathcal{A},\beta,m})] = \mathbf{m}(B^{\mathcal{A},\beta,m})$, $\hat{\mathbb{P}}$ -almost surely by the definition of an extreme point. So there exists a \hat{T} -invariant event of $\hat{\mathbb{P}}$ -probability one, on which $\mathcal{U}_{\mathbf{m}(B^{\mathcal{A},\beta,m})}^{\beta} = \mathcal{U}_{m,\mathcal{A}}^{\beta}$ and $\mathcal{F}_{\mathbf{m}(B^{\mathcal{A},\beta,m})}^{\beta} = \mathcal{F}_{m,\mathcal{A}}^{\beta}$. Let $\hat{\Omega}_{\operatorname{dir}}$ be the intersection of this event with $\hat{\Omega}_{0}$. All the claims of the theorem are now verified, but with the quenched measures being measurable functions of $\hat{\omega}$.

It remains to construct the family of measures on the original space Ω , using a standard argument in measure theory. To this end, recall that π_{Ω} is the projection from $\widehat{\Omega}$ to Ω . By [11, Example 10.4.11], there exist a *T*-invariant Borel set $\Omega_{\text{reg}} \subset \Omega$ and a family of regular conditional distributions $\mu_{\omega}(\cdot) = \widehat{\mathbb{P}}(\cdot | \pi_{\Omega}^{-1}(\omega))$ such that $\mathbb{P}(\Omega_{\text{reg}}) = 1$ and $\mu_{\omega}(\pi_{\Omega}^{-1}(\omega)) = 1$ for all $\omega \in \Omega_{\text{reg}}$. For \mathbb{P} -a.e. $\omega \in \Omega$, $\mu_{\omega}(\widehat{\Omega}_{\text{dir}}) = 1$ since

$$\int \mu_{\omega}(\widehat{\Omega}_{\mathrm{dir}}) \mathbb{P}(d\omega) = \widehat{\mathbb{P}}(\widehat{\Omega}_{\mathrm{dir}}) = 1.$$

Define the event $\Omega_{\text{dir}} = \Omega_{\text{reg}} \cap \{\omega \in \Omega : \mu_{\omega}(\widehat{\Omega}_0) = 1\}$. Then $\mathbb{P}(\Omega_{\text{dir}}) = 1$. For each $\omega \in \Omega_{\text{dir}}, \mu_{\omega}(\pi_{\Omega}^{-1}(\omega) \cap \widehat{\Omega}_{\text{dir}}) = 1$, so there exists $\widehat{\omega} \in \widehat{\Omega}_0$ such that $\pi_{\Omega}(\widehat{\omega}) = \omega$. For each $\omega \in \Omega_{\text{dir}}$, define $Q_x^{\mathcal{A},\beta,m,\omega} = Q_x^{\mathcal{A},\beta,m,\widehat{\omega}}$. This family satisfies all the desired claims. \Box

Appendix A. Basic convex analysis and linear algebra facts

In this short appendix, we recall some convex analysis facts and prove some lemmas that are of use to us. For a convex set K, A is called a face of K if for all $\xi, \eta \in K$ and $t \in (0, 1), t\xi + (1 - t)\eta \in A$ implies that $\xi, \eta \in A$. The intersection of faces is a face. K itself is a face, and, by [62, Corollary 18.1.3], all other faces are contained in the relative boundary of K. Extreme points are the zero-dimensional faces. If $\xi \in A$ can be written as a convex combination of $\eta, \zeta \in K$ then $\eta, \zeta \in A$. The relative interiors of the non-empty faces of K form a partition of K by [62, Theorem 18.2]. Thus, every $\xi \in K$ has a unique face K_{ξ} such that $\xi \in \operatorname{ri} K_{\xi}$. If K is in the convex set (respectively convex cone) generated by a set R, then by [62, Theorem 18.3] a face A of K is in the convex set (convex cone) generated by $R \cap A$.

Lemma A.1 (Euler's homogeneous concave function theorem). Let \mathcal{X} and \mathcal{Y} be real vector spaces with a bilinear function $\langle \cdot, \cdot \rangle : \mathcal{X} \times \mathcal{Y} \to \mathbb{R}$. Let $f : \mathcal{X} \to [-\infty, \infty)$ be a proper, concave, positively 1-homogeneous function. The superdifferential at $x \in \mathcal{X}$ is

$$\partial f(x) = \{ y \in \mathcal{Y} : \forall u \in \mathcal{X}, f(u) \leq f(x) + \langle u - x, y \rangle \}.$$

For $x \in \mathcal{X}$ and $y \in \mathcal{Y}$, if $y \in \partial f(x)$ then $f(x) = \langle x, y \rangle$.

Proof. Let $x \in \mathcal{X}$ and $y \in \mathcal{Y}$ such that $y \in \partial f(x)$. Let $\lambda > 0$ and $u = \lambda x$. By homogeneity,

$$(\lambda - 1)f(x) = f(u) - f(x) \leq \langle u, y \rangle - \langle x, y \rangle = (\lambda - 1)\langle x, y \rangle$$

With $\lambda < 1$, this implies $f(x) \ge \langle x, y \rangle$. With $\lambda > 1$, this implies $f(x) \le \langle x, y \rangle$. Therefore, $f(x) = \langle x, y \rangle$. \Box

Lemma A.2. Let $d \in \mathbb{Z}_{>0}$ and $\mathcal{R} \subset \mathbb{Z}^d$. The following are equivalent:

- (a) For any $x, y \in \mathbb{Z}^d$ with $y x \in \mathcal{G}$, all paths in $\mathbb{X}_{x,y}$ have the same length.
- (b) There exists a vector $\hat{u} \in \mathbb{R}^d$ such that $\hat{u} \cdot z = 1$ for all $z \in \mathcal{R}$.

Proof. Suppose (b) holds. Then any path $x_{0:k} \in \mathbb{X}_{x,y}$ satisfies $(y-x) \cdot \hat{u} = \sum_{i=0}^{k-1} (x_{i+1} - x_i) \cdot \hat{u} = k$. Thus, (a) holds.

Now suppose (a) holds. Let \mathbb{V} denote the linear span of \mathcal{R} . Let $k \in [1,d]$ be the dimension of this vector space and let $z_1, \ldots, z_k \in \mathcal{R}$ be a basis for it. Augment this set to a basis $\{z_1, \ldots, z_k, z_{k+1}, \ldots, z_d\}$ of \mathbb{R}^d ,

where z_{k+1}, \ldots, z_d are also integer vectors. Let A be the unique invertible linear transformation such that $Az_i = e_i$ for $1 \le i \le d$. In the standard basis, the matrix of A is the inverse of the matrix $B = [z_1, \ldots, z_d]$; hence this matrix has rational entries.

Take $z \in \mathcal{R}$. There is a unique set of numbers $\{a_1, \ldots, a_k\}$ such that $z = \sum_{i=1}^k a_i z_i$. Applying A gives $Az = \sum_{i=1}^k a_i e_i$. Since the left-hand side has rational coordinates, we get that a_i is rational for $1 \leq i \leq k$. Let $n \in \mathbb{Z}_{>0}$ be such that $na_i \in \mathbb{Z}$ for all $i \leq k$.

Since $nz + \sum_{i=1}^{k} na_i^- z_i = \sum_{i=1}^{k} na_i^+ z_i$, (a) implies that $n + \sum_{i=1}^{k} na_i^- = \sum_{i=1}^{k} na_i^+$, which implies that $\sum_{i=1}^{k} a_i = 1$. Consequently, we have $z - z_1 = \sum_{i=2}^{k} a_i(z_i - z_1)$. We have thus shown that the linear span of $\{z - z_1 : z \in \mathcal{R}\}$ has dimension at most k - 1. As this is a subspace of \mathbb{V} and \mathbb{V} has dimension k, there exists a vector $\hat{u}' \in \mathbb{V} \setminus \{0\}$ that is orthogonal to $z - z_1$ for all $z \in \mathcal{R}$. This implies that $\hat{u}' \cdot z = \hat{u}' \cdot z'$ for all $z, z' \in \mathcal{R}$. Denote this common number by c. If c = 0, then \hat{u}' is perpendicular to all $z \in \mathcal{R}$ and is hence perpendicular to \mathbb{V} . However, then \hat{u}' is perpendicular to itself, and thus $\hat{u}' = \mathbf{0}$. This contradicts the choice of \hat{u}' and proves that $c \neq 0$. Taking $\hat{u} = c^{-1}\hat{u}'$ satisfies (b).

Let $\mathcal{R} \subset \mathbb{Z}^d$ and $\mathcal{C} = \left\{ \sum_{z \in \mathcal{R}} b_z z : b_z \in \mathbb{R}_+ \right\}$. Fix a face \mathcal{A} of \mathcal{C} and let $\overline{\mathcal{G}}_{\mathcal{A}}$ be the subgroup of \mathbb{Z}^{d+1} generated by $\{\langle z, 1 \rangle : z \in \mathcal{R}_{\mathcal{A}}\}$. Let $\mathcal{G}_{\mathcal{A}}^{(0)}$ be the group generated by $\{z - z' : z, z' \in \mathcal{R}_{\mathcal{A}}\}$. Take $z_0 \in \mathcal{R}_{\mathcal{A}}$ and for $j \in \mathbb{Z}$ let $\mathcal{G}_{\mathcal{A}}^{(j)} = jz_0 + \mathcal{G}_{\mathcal{A}}^{(0)}$. Note that $\mathcal{G}_{\mathcal{A}}^{(j)}$ does not depend on the choice of z_0 .

Lemma A.3. We have

$$\overline{\mathcal{G}}_{\mathcal{A}} = \bigcup_{j \in \mathbb{Z}} \{ \langle x, j \rangle : x \in \mathcal{G}_{\mathcal{A}}^{(j)} \}.$$

Proof. If $x \in \mathcal{G}_{\mathcal{A}}^{(0)}$, then $x = \sum_{i=1}^{k} (z_i - z'_i)$ for some $z_1, \dots, z_k, z'_1, \dots, z'_k \in \mathcal{R}_{\mathcal{A}}$ and, consequently, $\langle x, 0 \rangle = \sum_{i=1}^{k} (\langle z_i, 1 \rangle - \langle z'_i, 1 \rangle) \in \overline{\mathcal{G}}_{\mathcal{A}}.$

If $j \in \mathbb{Z}$, then

 $\langle jz_0 + x, j \rangle = \langle x, 0 \rangle + j \langle z_0, 1 \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}.$

For the other direction, take $x \in \mathbb{Z}^d$ and $j \in \mathbb{Z}$ such that $\langle x, j \rangle \in \overline{\mathcal{G}}_{\mathcal{A}}$. Then

$$\langle x, j \rangle = \sum_{i=1}^{k} \langle z_i, 1 \rangle - \sum_{i=1}^{\ell} \langle z'_i, 1 \rangle,$$

for some $z_1, \ldots, z_k, z'_1, \ldots, z'_\ell \in \mathcal{R}_A$ and $k, \ell \in \mathbb{Z}_{\geq 0}$ with $k - \ell = j$. From this, we get

$$x = \sum_{i=1}^{k} z_i + \ell z_0 - k z_0 - \sum_{i=1}^{\ell} z'_i + j z_0 \in \mathcal{G}_{\mathcal{A}}^{(j)}.$$

Proof of (4.15). Note that $\overline{m} \in \partial \overline{\Lambda}_{\mathcal{A}}^{\beta, \text{usc}}(\langle \xi, 1 \rangle)$ is equivalent to having

$$t\overline{\Lambda}_{\overline{\mathcal{A}}}^{\beta,\mathrm{usc}}(\langle \zeta,1\rangle) - \overline{\Lambda}_{\overline{\mathcal{A}}}^{\beta,\mathrm{usc}}(\langle \xi,1\rangle) \leqslant m \cdot (t\zeta - \xi) + c(t-1)$$

for all $\zeta \in \mathcal{U}'$ and t > 0. Rearranging and reverting back to restricted-length gives

$$t\left(\Lambda_{\mathcal{U}',\mathrm{res}}^{\beta,\mathrm{usc}}(\zeta) - m \cdot \zeta - c\right) \leqslant \Lambda_{\mathcal{U}',\mathrm{res}}^{\beta,\mathrm{usc}}(\xi) - m \cdot \xi - c.$$

Taking $t \to 0$ and $t \to \infty$ gives

$$\Lambda_{\mathcal{U}',\mathrm{res}}^{\beta,\mathrm{usc}}(\zeta) - m \cdot \zeta \leqslant c \leqslant \Lambda_{\mathcal{U}',\mathrm{res}}^{\beta,\mathrm{usc}}(\xi) - m \cdot \xi,$$

for all $\zeta \in \mathcal{U}'$. This gives

$$c = \Lambda_{\mathcal{U}', \mathrm{res}}^{\beta, \mathrm{usc}}(\xi) - m \cdot \xi$$

and implies that m is in the superdifferential at ξ of the concave function that is equal to $\Lambda_{\mathcal{A}}^{\beta, \text{usc}}$ on \mathcal{U}' and is set to $-\infty$ outside \mathcal{U}' .

We close this section the proof of an observation made just prior to Theorem 2.15.

Lemma A.4. Fix $x \in C$ and let A be the unique face of C for which $x \in \operatorname{ri} A$. Let $x_{0:n}$ be any path with $x_0 = 0$, $x_n = x$, and $z_i = x_i - x_{i-1} \in \mathcal{R}$ for $i = 1, \ldots, n$. Then $x_i \in A$ for all $i = 0, \ldots, n$.

Proof. We begin by noting that $\mathbf{0} \in \mathcal{A}$. If $\mathcal{C} \neq \mathbb{R}^d$, then \mathcal{C} is polyhedral ([62, Theorem 19.1]), i.e. equal to the intersection of finitely many closed half-spaces whose boundary hyperplanes pass through the origin. The facets of such a set are obtained by intersecting with these hyperplanes and so in particular $\mathbf{0} \in \mathcal{A}$.

Next, denote by S the set of steps $x_i - x_{i-1} = z_i \in \mathcal{R}$, i = 1, ..., n used in this path. By definition, x lies in the relative interior of the convex hull of nS, which is a convex set in \mathcal{C} . Because x lies in the relative interior of the face \mathcal{A} , it follows from [62, Theorem 18.1] that the convex hull of nS is a subset of \mathcal{A} . For each i = 1, ..., n, x_i is a convex combination of **0** and points in nS, so $x_i \in \mathcal{A}$ for all such i.

APPENDIX B. GIBBS CONSISTENCY

In this appendix, we discuss the Gibbs consistency of the various polymer measures to place our results within the framework of Gibbs specifications and solutions to the Dobrushin-Landford-Ruelle (DLR) equations.

As stated in the next lemma, the restricted-length finite path measures are consistent in the sense of conditioning, indicating that this family forms a Gibbs specification. See Section 2.4 in [38].

Lemma B.1. For $j, k, n \in \mathbb{Z}_{\geq 0}$ with $j \leq k \leq n$, for $u, v, x, y \in \mathbb{Z}^d$ such that $x - u \in D_j$, $y - x \in D_{k-j}$, and $v - y \in D_{n-k}$, and for $\beta \in (0, \infty)$, $\omega \in \Omega$, $x_{0:j} \in \mathbb{X}_{u,x}^{0,j}$, $x_{j:k} \in \mathbb{X}_{x,y}^{j,k}$, and $x_{k:n} \in \mathbb{X}_{y,v}^{k,n}$,

$$Q_{u,v,n}^{\beta,\omega}(X_{0:n} = x_{0:n} \mid X_{0:j} = x_{0:j}, X_{k:n} = x_{k:n}) = Q_{u,v,n}^{\beta,\omega}(X_{j:k} = x_{j:k} \mid X_j = x, X_k = y)$$
$$= Q_{x,y,k-j}^{\beta,\omega}(X_{0:k-j} = x_{j:k}).$$

Proof. Let $j, k, n, u, x, y, v, \beta, \omega$ be as in the statement. Let $x_{0:j} \in \mathbb{X}_{u,x}^{0,j}, x_{j:k} \in \mathbb{X}_{x,y}^{j,k}$, and $x_{k:n} \in \mathbb{X}_{y,v}^{k,n}$. Then

$$\begin{aligned} Q_{u,v,n}^{\beta,\omega}(X_{0:n} = x_{0:n} | X_{0:j} = x_{0:j}, X_{k:n} = x_{k:n}) \\ &= \frac{Q_{u,v,n}^{\beta,\omega}(X_{0:n} = x_{0:n})}{Q_{u,v,n}^{\beta,\omega}(X_{0:j} = x_{0:j}, X_{k:n} = x_{k:n})} \\ &= \frac{\prod_{i=0}^{n-1} p(x_{i+1} - x_i) e^{-\beta V(T_{x_i}\omega, x_{i+1} - x_i)}}{\prod_{i=0}^{j-1} p(x_{i+1} - x_i) e^{-\beta V(T_{x_i}\omega, x_{i+1} - x_i)}} \\ &= \frac{\prod_{i=j}^{k-1} p(x_{i+1} - x_i) e^{-\beta V(T_{x_i}\omega, x_{i+1} - x_i)}}{Z_{x,y,k-j}^{\beta,\omega}} \\ &= \frac{Q_{x,y,k-j}^{\beta,\omega}(X_{0:k-j} = x_{j:k}). \end{aligned}$$

Similarly,

$$Q_{u,v,n}^{\beta,\omega}(X_{j:k} = x_{j:k} \mid X_j = x, X_k = y) = Q_{x,y,k-j}^{\beta,\omega}(X_{0:k-j} = x_{j:k}).$$

Let \mathcal{U} be the convex hull of \mathcal{R} , with ri \mathcal{U} denoting its relative interior. The next result shows that the unrestricted-length finite path measures are also Gibbs consistent if $\mathbf{0} \notin \mathcal{U}$.

Lemma B.2. Assume $\mathbf{0} \notin \mathcal{U}$. Let $\beta \in (0, \infty)$, $\omega \in \Omega$, and $u, v, x, y \in \mathbb{Z}^d$ such that $x - u, y - x, v - y \in \mathcal{G}^+$. Let $x_{0:n} \in \overline{\mathbb{X}}_{u,v}$, such that $x_j = x$ and $x_k = y$ for some $0 \leq j \leq k \leq n$. Then,

$$Q_{u,v}^{\beta,\omega}(X_{0:\tau_v} = x_{0:n} \mid X_{0:\tau_x} = x_{0:j}, X_{\tau_y:\tau_v} = x_{k:n}) = Q_{u,v}^{\beta,\omega}(X_{\tau_x:\tau_y} = x_{j:k} \mid \tau_x \leqslant \tau_y < \infty)$$
$$= Q_{x,v}^{\beta,\omega}(X_{0:\tau_v} = x_{j:k}).$$

Proof. Assume $\mathbf{0} \notin \mathcal{U}$ and let $\beta \in (0, \infty)$, $\omega \in \Omega$, and $u, v, x, y \in \mathbb{Z}^d$ such that $x - u, y - x, v - y \in \mathcal{G}^+$. Let $x_{0:n} \in \overline{\mathbb{X}}_{u,v}$ such that $x_j = x$ and $x_k = y$. Since there are no loops and $y - x \in \mathcal{G}^+$, it must be that $0 \leq j \leq k \leq n$. Furthermore, on the event $\{\tau_x < \infty, \tau_y < \infty\}$, it must be that $\tau_x \leq \tau_y \leq \tau_v$, for otherwise we can construct a loop. Then,

$$\begin{aligned} Q_{u,v}^{\beta,\omega}(X_{0:\tau_{v}} = x_{0:n} \mid X_{0:\tau_{x}} = x_{0:j}, X_{\tau_{y}:\tau_{v}} = x_{k:n}) \\ &= \frac{Q_{u,v}^{\beta,\omega}(X_{0:\tau_{v}} = x_{0:n})}{Q_{u,v}^{\beta,\omega}(X_{0:\tau_{x}} = x_{0:j}, X_{\tau_{y}:\tau_{v}} = x_{k:n})} \\ &= \frac{\prod_{i=0}^{n-1} p(x_{i+1} - x_{i})e^{-\beta V(T_{x_{i}}\omega, x_{i+1} - x_{i})}}{\prod_{i=0}^{j-1} p(x_{i+1} - x_{i})e^{-\beta V(T_{x_{i}}\omega, x_{i+1} - x_{i})}} \\ &= \frac{\prod_{i=j}^{k-1} p(x_{i+1} - x_{i})e^{-\beta V(T_{x_{i}}\omega, x_{i+1} - x_{i})}}{Z_{x,y}^{\beta,\omega}} \\ &= \frac{Q_{x,y}^{\beta,\omega}(X_{0:\tau_{y}} = x_{j:k}). \end{aligned}$$

Similarly,

$$\begin{aligned} Q_{u,v}^{\beta,\omega}(X_{\tau_x:\tau_y} &= x_{j:k} \mid \tau_x \leqslant \tau_y < \infty) \\ &= \frac{Q_{u,v}^{\beta,\omega}(\tau_x \leqslant \tau_y < \infty, X_{\tau_x:\tau_y} = x_{j:k})}{Q_{u,v}^{\beta,\omega}(\tau_x \leqslant \tau_y < \infty)} \\ &= \frac{Z_{u,x}^{\beta,\omega} \prod_{i=j}^{k-1} p(x_{i+1} - x_i) e^{-\beta V(T_{x_i}\omega, x_{i+1} - x_i)} Z_{y,v}^{\beta,\omega}}{Z_{u,x}^{\beta,\omega} Z_{x,y}^{\beta,\omega} Z_{y,v}^{\beta,\omega}} \\ &= \frac{\prod_{i=j}^{k-1} p(x_{i+1} - x_i) e^{-\beta V(T_{x_i}\omega, x_{i+1} - x_i)}}{Z_{x,y}^{\beta,\omega}} \\ &= Q_{x,y}^{\beta,\omega}(X_{0:\tau_y} = x_{j:k}). \end{aligned}$$

The measures $Q_{u,v}^{\beta,\omega}$ are not consistent in general if $\mathbf{0} \in \mathcal{U}$, but they are asymptotically consistent as $|v|_1 \to \infty$. To see this, we introduce some more notation and definitions. Let $x - u, y - x, v - y \in \mathcal{G}^+$ with $y \neq v$. Define the partition functions

$$Z_{u,x}^{\beta,\omega}(\tau_x \leqslant \min\{\tau_y, \tau_v\}) = \mathbf{E}_u \bigg[e^{-\beta \sum_{i=0}^{\tau_x - 1} V(T_{X_i}\omega, X_{i+1} - X_i)} \mathbb{1}_{\{\tau_x < \infty, \tau_x \leqslant \min\{\tau_y, \tau_v\}\}} \bigg] \text{ and } Z_{x,y}^{\beta,\omega}(\tau_y \leqslant \tau_v) = \mathbf{E}_x \bigg[e^{-\beta \sum_{i=0}^{\tau_y - 1} V(T_{X_i}\omega, X_{i+1} - X_i)} \mathbb{1}_{\{\tau_y < \infty, \tau_y \leqslant \tau_v\}} \bigg].$$

Take $j \leq k$ in \mathbb{Z} and $x_{j:k} \in \mathbb{X}^{j,k}_{x,y}$ such that $x_i \notin \{y, v\}$ for all integers $i \in [j,k)$ and $x_k = y$. Then,

$$Q_{u,v}^{\beta,\omega}(X_{\tau_x:\tau_y} = x_{j:k}, \tau_x = j, \tau_y = k) = \frac{Z_{u,x}^{\beta,\omega}(\tau_x \le \min\{\tau_y, \tau_v\})e^{-\beta\sum_{i=j}^{k-1}V(T_{x_i}\omega, x_{i+1}-x_i)}Z_{y,v}^{\beta,\omega}}{Z_{u,v}^{\beta,\omega}}$$

whereas,

$$\begin{aligned} \mathbf{Q}_{u,v}^{\beta,\omega}(\tau_x \leqslant \tau_y \leqslant \tau_v) \cdot \mathbf{Q}_{x,y}^{\beta,\omega}(X_{0:k-j} = x_{j:k}) \\ &= \frac{Z_{u,x}^{\beta,\omega}(\tau_x \leqslant \min\{\tau_y,\tau_v\}) Z_{x,y}^{\beta,\omega}(\tau_y \leqslant \tau_v) Z_{y,v}^{\beta,\omega}}{Z_{u,v}^{\beta,\omega}} \cdot \frac{e^{-\beta \sum_{i=j}^{k-1} V(T_{x_i}\omega, x_{i+1} - x_i)}}{Z_{x,y}^{\beta,\omega}}. \end{aligned}$$

Thus, we do not get exact consistency, since $Z_{x,y}^{\beta,\omega}(\tau_y \leq \tau_v)/Z_{x,y}^{\beta,\omega}$ is not equal to 1 until we take $|v|_1 \to \infty$. When $\mathbf{0} \notin \mathcal{U}$, this issue is resolved since when $v - y \in \mathcal{G}^+$ having $\tau_y < \infty$ implies $\tau_y \leq \tau_v$.

It is shown in Theorem 3.14 above that there exist measures on semi-infinite paths that are consistent with the point-to-point measures in the sense that

$$\mathbf{Q}_{u}^{\beta,\omega}(X_{\tau_{x}:\tau_{y}}=x_{j:k} \mid \tau_{x} \leqslant \tau_{y} < \infty) = \mathbf{Q}_{x,y}^{\beta,\omega}(X_{0:\tau_{y}}=x_{j:k})$$

At zero temperature, the consistency properties become the following facts about geodesics. Given a geodesic $x_{0:n}$, for any integers $0 \leq j \leq k \leq n$, $x_{j:k}$ optimizes the passage time among all paths in $\mathbb{X}_{x_j,x_k}^{k-j}$ in the restricted-length case, and among all paths in \mathbb{X}_{x_j,x_k} in the unrestricted-length case. The question about the existence of semi-infinite polymer measures becomes one of the existence of restricted-length or, respectively, unrestricted-length semi-infinite geodesics. These are semi-infinite admissible paths $x_{0:\infty}$ with

the property that for all integers $0 \leq j < k$, $x_{j:k}$ is a restricted-length or, respectively, an unrestricted-length geodesic from x_j to x_k .

Appendix C. Shape theorems

In this appendix, we prove the shape theorems which were stated in Section 2.3. These results play a key role in the body of the paper when we prove the duality between the Busemann function mean vector and the direction of the associated Gibbs measure.

Proof of Theorem 2.15. The finiteness of $\Lambda_{\mathcal{A}}^{\beta}$ comes from [37, Theorem 3.10], after adjusting their proof as in Lemma 5.20 to accommodate our weaker Condition 3.5. As a finite concave function, $\Lambda_{\mathcal{A}}^{\beta}$ is continuous on the convex open set ri \mathcal{A} . Lower semicontinuity implies that $\Lambda_{\mathcal{A}}^{\beta}$ is bounded below (with a bound that can depend on ω), uniformly over any bounded subset of \mathcal{A} . Then [62, Theorem 10.3] implies that $\Lambda_{\mathcal{A}}^{\beta}$ has a unique continuous extension from the relative interior to the whole of \mathcal{A} . The general argument on page 726 of [58] shows that this extension agrees with the upper semicontinuous regularization. (In [58], the argument is made for functions on \mathcal{U} , but it works word for word for functions on \mathcal{A} .)

The second inequality in (2.8) is implied by (2.7). The argument for both the first inequality in (2.8) and for (2.7) follows the proof of [37, Theorem 3.10], with minor modifications that we will highlight. Consider the following two cases. In Case 1, assume that with positive probability there exists an $\varepsilon > 0$ and a sequence $x_n \in \mathcal{G}^+_{\mathcal{A}}$ such that $|x_n|_1 \to \infty$ and $F^{\beta}_{\mathbf{0},x_n} - \Lambda^{\beta,\text{usc}}_{\mathcal{A}}(x_n) \ge \varepsilon |x_n|_1$ for all n. In Case 2, assume that with positive probability there exists a $\xi \in \mathcal{A}$ with $|\xi|_1 = 1$ and a sequence $x_n \in \mathcal{G}^+_{\mathcal{A}}$ such that $|x_n|_1 \to \infty$, $x_n/|x_n|_1 \to \xi$, and

(C.1)
$$|x_n|_1^{-1} F_{\mathbf{0}, x_n}^{\beta} - \Lambda^{\beta}(\xi) \leqslant -\varepsilon \quad \text{for all } n.$$

In the first case, we can follow the approach in [37] and extract a subsequence (which we denote as x_n again) such that $x_n/|x_n|_1$ converges to some $\xi \in \mathcal{A}$. Since $\Lambda_{\mathcal{A}}^{\beta, \text{usc}}$ is continuous on \mathcal{A} we have

(C.2)
$$|x_n|_1^{-1} F_{\mathbf{0},x_n}^{\beta} - \Lambda_{\mathcal{A}}^{\beta,\text{usc}}(\xi) \ge \varepsilon/2 \text{ for all } n \text{ large enough.}$$

Now, in either case, follow the argument in [37, Theorem 3.10], specifically below their equation (3.10), adjusting it as shown in the proof of Lemma 5.20 to accommodate our weaker Condition 3.5. Following this adjusted argument, we reach the conclusion that \mathbb{P} -almost surely, on the event where either (C.1) holds or (C.2) holds, we have

$$-\varepsilon_2 + \Lambda^{\beta}(\xi) \leq \lim_{n \to \infty} |x_n|^{-1} F^{\beta}_{\mathbf{0}, x_n} \leq \overline{\lim_{n \to \infty}} |x_n|^{-1} F^{\beta}_{\mathbf{0}, x_n} \leq \Lambda^{\beta} \left(\xi + \varepsilon_1 \sum_{z \in \mathcal{R}_{\mathcal{A}}} z\right) + \varepsilon_2,$$

where $\varepsilon_2 > 0$ is arbitrary, and $\varepsilon_1 > 0$ can be chosen to be arbitrarily small, depending on ε_2 . Take $\varepsilon_1 \to 0$ then $\varepsilon_2 \to 0$ and use the facts that $\Lambda^{\beta} \left(\xi + \varepsilon_1 \sum_{z \in \mathcal{R}_{\mathcal{A}}} z \right) = \Lambda^{\beta, \text{usc}}_{\mathcal{A}} \left(\xi + \varepsilon_1 \sum_{z \in \mathcal{R}_{\mathcal{A}}} z \right)$ and $\Lambda^{\beta, \text{usc}}_{\mathcal{A}}$ is continuous on \mathcal{A} to get that

$$\Lambda^{\beta}(\xi) \leqslant \lim_{n \to \infty} |x_n|^{-1} F^{\beta}_{\mathbf{0}, x_n} \leqslant \overline{\lim_{n \to \infty}} |x_n|^{-1} F^{\beta}_{\mathbf{0}, x_n} \leqslant \Lambda^{\beta, \text{usc}}_{\mathcal{A}}(\xi),$$

which contradicts both (C.1) and (C.2). This proves the desired inequalities. The ergodicity claim is already in Theorem 3.10 of [37]. The theorem is proved. \Box

Proof of Theorem 2.17. Write the restricted-length model as an unrestricted-length model as in Remark 2.2. Then the conditions on V^+ in the statement of the theorem imply that \overline{V}^+ satisfies the hypotheses of Theorem 2.15 and, as explained in Remark 3.19, $\overline{\Lambda}_{\overline{\mathcal{A}}}^{\beta,\text{usc}}(\langle \zeta, t \rangle) = t \Lambda_{\mathcal{U}',\text{res}}^{\beta,\text{usc}}(\zeta/t)$ for all $\zeta \in \mathcal{A}$ and t > 0 with $\zeta/t \in \mathcal{U}'$. Applying (2.7) and (2.8) to the unrestricted-length model gives (2.9) and (2.10) for the restricted-length model. The ergodicity of \mathbb{P} under $\{T_z : z \in \mathcal{R}_{\mathcal{A}}\}$ is equivalent to its ergodicity under $\{T_{\langle z,1 \rangle} : \langle z,1 \rangle \in \overline{\mathcal{R}}_{\overline{\mathcal{A}}}\}$, which by Theorem 2.15 implies that $\overline{\Lambda}_{\overline{\mathcal{A}}}^{\beta,\text{usc}}$ is deterministic on $\overline{\mathcal{A}}$ and, therefore, in this case, $\Lambda_{\mathcal{U}',\text{res}}^{\beta,\text{usc}}$ is deterministic on \mathcal{U}' .

Appendix D. Relationship between restricted and unrestricted-length

We prove a theorem relating the restricted-length and unrestricted-length limiting free energies in the directed setting where $\mathbf{0} \notin \mathcal{U}$. This connection is also known to hold in the case of the standard first-passage percolation model. See [45, Equation (2.37)]. It is natural to expect that our theorem continues to hold under appropriate hypotheses when the model has loops, i.e., $\mathbf{0} \in \mathcal{U}$. We leave this for future work.

Theorem D.1. Fix a face $\mathcal{A} \in \mathbb{A}$ (possibly \mathcal{C} itself). Assume $\mathbf{0} \notin \mathcal{U}_{\mathcal{A}}$, Conditions 3.2 and 3.5 hold on \mathcal{A} , and \mathbb{P} is ergodic under $\{T_x : x \in \mathcal{G}_{\mathcal{A}}\}$. Then for each $\beta \in (0, \infty]$ and $\xi \in (\mathrm{ri} \mathcal{A}) \setminus \{\mathbf{0}\}$,

(D.1)
$$\Lambda^{\beta}(\xi) = \sup_{\substack{s>0\\\xi/s\in\mathcal{U}_{\mathcal{A}}}} \{s\Lambda^{\beta,\mathrm{usc}}_{\mathrm{res}}(\xi/s)\}, \text{ and the supremum is achieved.}$$

Proof. For each $\zeta \in \mathcal{U}$ and $n \in \mathbb{Z}_{>0}$, define the lattice point $\hat{x}_n(\zeta)$ as in [58, (2.1)]. The point $\hat{x}_n(\zeta)$ approximates $n\zeta$. In particular, $\hat{x}_n(\zeta)/n \to \zeta$ as $n \to \infty$, and if $\zeta \in \mathcal{U}_A$, then $\hat{x}_n(\zeta) \in D_n(\mathcal{R}_A)$.

For the lower bound, note that there are no loops since $\mathbf{0} \notin \mathcal{U}_{\mathcal{A}}$. Let $\xi \in (\mathrm{ri} \,\mathcal{A}) \setminus \{\mathbf{0}\}$. For any s > 0 such that $\xi/s \in \mathrm{ri} \,\mathcal{U}_{\mathcal{A}}$,

$$\frac{1}{n}\log Z^{\beta}_{\mathbf{0},\hat{x}_{\lfloor sn \rfloor}(\xi/s)} = \frac{1}{n}\log \sum_{k=0}^{\infty} Z^{\beta}_{\mathbf{0},\hat{x}_{\lfloor sn \rfloor}(\xi/s),k} \ge \frac{1}{n}\log Z^{\beta}_{\mathbf{0},\hat{x}_{\lfloor sn \rfloor}(\xi/s),\lfloor sn \rfloor}$$

The equivalent bound at zero temperature is $n^{-1}F^{\infty}_{\mathbf{0},\hat{x}_{\lfloor sn \rfloor}(\xi/s)} \ge n^{-1}F^{\infty}_{\mathbf{0},\hat{x}_{\lfloor sn \rfloor}(\xi/s),\lfloor sn \rfloor}$. Take limits, apply Theorems 2.15 and 2.17, and take the supremum over s > 0 to get

(D.2)
$$\Lambda^{\beta}_{\mathcal{A}}(\xi) = \Lambda^{\beta, \text{usc}}_{\mathcal{A}}(\xi) \ge \sup_{\substack{s>0\\ \xi/s \in \text{ri}\,\mathcal{U}_{\mathcal{A}}}} \{s\Lambda^{\beta}_{\text{res}}(\xi/s)\} = \sup_{\substack{s>0\\ \xi/s \in \mathcal{U}_{\mathcal{A}}}} \{s\Lambda^{\beta, \text{usc}}_{\text{res}}(\xi/s)\}.$$

For the upper bound, first note that by [37, Lemma A.1] there exist a $\hat{u} \in \mathbb{R}^d$ and a $\delta > 0$ such that $\hat{u} \cdot z \ge \delta > 0$ for all $z \in \mathcal{R}_A$. Any admissible path from **0** to x_n must take at least $|x_n|_1 \min_{z \in \mathcal{R}_A} |z|_1^{-1}$ steps and at most $\delta^{-1}x_n \cdot \hat{u}$ steps. Thus, for *n* large enough, the number of steps is between εn and Cn, for some finite positive constants $C = C(\xi)$ and $\varepsilon = \varepsilon(\xi)$. For such *n*, we have

$$\frac{1}{n}\log Z_{\mathbf{0},x_n}^{\beta} = \frac{1}{n}\log \mathbf{E}_{\mathbf{0}} \Big[e^{-\beta \sum_{i=0}^{\tau_{x_n}-1} V(T_{X_i}\omega, X_{i+1}-X_i)} \mathbb{1}_{\{\tau_{x_n}<\infty\}} \Big]$$
$$= \frac{1}{n}\log \sum_{k=0}^{\infty} \mathbf{E}_{\mathbf{0}} \Big[e^{-\beta \sum_{i=0}^{k-1} V(T_{X_i}\omega, X_{i+1}-X_i)} \mathbb{1}_{\{\tau_{x_n}=k\}} \mathbb{1}_{\{X_k=x_n\}} \Big]$$
$$\leqslant \frac{1}{n}\log \sum_{\varepsilon n \leqslant k \leqslant Cn} Z_{\mathbf{0},x_n,k}^{\beta}$$
$$\leqslant \max_{\varepsilon n \leqslant k \leqslant Cn} \frac{1}{n}\log Z_{\mathbf{0},x_n,k}^{\beta} + \frac{1}{n}\log Cn.$$

The equivalent bound at zero temperature is

$$\frac{1}{n}F^{\infty}_{\mathbf{0},x_n}\leqslant \max_{\varepsilon n\leqslant k\leqslant Cn}\frac{1}{n}F^{\infty}_{\mathbf{0},x_n,k}.$$

In either case, let $k_n \in [\varepsilon n, Cn]$ be the integer that achieves the maximum. Then

$$\frac{1}{n}F_{\mathbf{0},x_n}^{\beta} = \frac{k_n}{n} \cdot \frac{1}{k_n}F_{\mathbf{0},x_n,k_n}^{\beta}$$

Take a subsequence such that k_n/n converges to some $t \in [\varepsilon, C]$. In particular, $k_n \to \infty$ and $x_n/k_n \to \xi/t \in \mathcal{U}_A$ as $n \to \infty$. Take $n \to \infty$ along this subsequence and apply Theorems 2.15 and 2.17 to get

$$\Lambda^{\beta}_{\mathcal{A}}(\xi) \leqslant t\Lambda^{\beta,\mathrm{usc}}_{\mathcal{A},\mathrm{res}}(\xi/t) \leqslant \sup_{\substack{s>0\\ \xi/s \in \mathcal{U}_{\mathcal{A}}}} s\Lambda^{\beta,\mathrm{usc}}_{\mathcal{A},\mathrm{res}}(\xi/s).$$

Together with the lower bound (D.2) we get that the inequalities above are, in fact, all equalities.

Remark D.2. Theorem 3.14 shows that the set $\mathcal{U}_{m,\mathcal{A}}^{\beta}$ must be non-empty. One can see this directly if one first proves the relationship in (D.1) on \mathcal{A} . Indeed, if $m \in \partial \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi)$ for $\xi \in (\text{ri }\mathcal{A}) \setminus \{\mathbf{0}\}$, then $m \cdot \xi = \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi)$, and using (D.1), one would get

$$m \cdot \xi = \Lambda_{\mathcal{A}}^{\beta, \text{usc}}(\xi) = \Lambda_{\mathcal{A}}^{\beta}(\xi) = s \Lambda_{\mathcal{A}, \text{res}}^{\beta, \text{usc}}(\xi/s)$$

for some s > 0 with $\xi/s \in \mathcal{U}_{\mathcal{A}}$. Therefore, $\Lambda_{\mathcal{A}, \operatorname{res}}^{\beta, \operatorname{usc}}(\xi/s) = m \cdot \xi/s$, and $\mathcal{U}_{m, \mathcal{A}}^{\beta}$ is non-empty.

Acknowledgment. The authors extend their gratitude to the anonymous referees for their meticulous and detailed reviews, as well as for their insightful and constructive comments.

References

- T. Alberts, F. Rassoul-Agha, and M. Simper. Busemann functions and semi-infinite O'Connell-Yor polymers. *Bernoulli*, 26(3):1927–1955, 2020.
- [2] K. S. Alexander and Q. Berger. Geodesics toward corners in first passage percolation. J. Stat. Phys., 172(4):1029–1056, 2018.
- [3] A. Auffinger, J. Hanson, and M. Damron. 50 years of first passage percolation, volume 68 of University Lecture Series. American Mathematical Society, Providence, RI, 2017.
- [4] Y. Bakhtin. The Burgers equation with Poisson random forcing. Ann. Probab., 41(4):2961–2989, 2013.
- [5] Y. Bakhtin. Inviscid Burgers equation with random kick forcing in noncompact setting. *Electron. J. Probab.*, 21:Paper No. 37, 50, 2016.
- [6] Y. Bakhtin, E. Cator, and K. Khanin. Space-time stationary solutions for the Burgers equation. J. Amer. Math. Soc., 27(1):193–238, 2014.
- [7] Y. Bakhtin and K. Khanin. On global solutions of the random Hamilton-Jacobi equations and the KPZ problem. *Nonlinearity*, 31(4):R93–R121, 2018.
- [8] Y. Bakhtin and L. Li. Thermodynamic limit for directed polymers and stationary solutions of the Burgers equation. *Comm. Pure Appl. Math.*, 72(3):536–619, 2019.
- [9] M. Balázs, F. Rassoul-Agha, and T. Seppäläinen. Large deviations and wandering exponent for random walk in a dynamic beta environment. Ann. Probab., 47(4):2186–2229, 2019.
- [10] E. Bates. Empirical measures, geodesic lengths, and a variational formula in first-passage percolation. Mem. Amer. Math. Soc., 293(1460):xi+92, 2024.
- [11] V. I. Bogachev. Measure theory. Vol. I, II. Springer-Verlag, Berlin, 2007.
- [12] R. C. Bradley. Basic properties of strong mixing conditions. A survey and some open questions. Probab. Surv., 2:107–144, 2005. Update of, and a supplement to, the 1986 original.
- [13] G. Brito and C. Hoffman. Geodesic rays and exponents in ergodic planar first passage percolation. In In and out of equilibrium 3. Celebrating Vladas Sidoravicius, volume 77 of Progr. Probab., pages 163–186. Birkhäuser/Springer, Cham, [2021] ©2021.
- [14] O. Busani, T. Seppäläinen, and E. Sorensen. The stationary horizon and semi-infinite geodesics in the directed landscape. Ann. Probab., 52(1):1–66, 2024.
- [15] P. Cardaliaguet and P. E. Souganidis. On the existence of correctors for the stochastic homogenization of viscous Hamilton-Jacobi equations. C. R. Math. Acad. Sci. Paris, 355(7):786–794, 2017.
- [16] E. Cator and L. P. R. Pimentel. A shape theorem and semi-infinite geodesics for the Hammersley model with random weights. ALEA Lat. Am. J. Probab. Math. Stat., 8:163–175, 2011.
- [17] E. Cator and L. P. R. Pimentel. Busemann functions and equilibrium measures in last passage percolation models. *Probab. Theory Related Fields*, 154(1-2):89–125, 2012.
- [18] E. Cator and L. P. R. Pimentel. Busemann functions and the speed of a second class particle in the rarefaction fan. Ann. Probab., 41(4):2401–2425, 2013.
- [19] J. Chaika and A. Krishnan. Stationary coalescing walks on the lattice. Probability Theory and Related Fields, 175(3):655–675, 2019.
- [20] J. Chaika and A. Krishnan. Stationary coalescing walks on the lattice II: entropy. Nonlinearity, 34(10):7045–7063, 2021.
- [21] F. Comets. Directed polymers in random environments, volume 2175 of Lecture Notes in Mathematics. Springer, Cham, 2017. Lecture notes from the 46th Probability Summer School held in Saint-Flour, 2016.

 \triangle

- [22] F. Comets and O. Zeitouni. A law of large numbers for random walks in random mixing environments. Ann. Probab., 32(1B):880–914, 2004.
- [23] M. Damron and J. Hanson. Busemann functions and infinite geodesics in two-dimensional first-passage percolation. Comm. Math. Phys., 325(3):917–963, 2014.
- [24] M. Damron and J. Hanson. Bigeodesics in First-Passage Percolation. Comm. Math. Phys., 349(2):753– 776, 2017.
- [25] O. Garet and R. Marchand. Coexistence in two-type first-passage percolation models. Ann. Appl. Probab., 15(1A):298–330, 2005.
- [26] N. Georgiou, F. Rassoul-Agha, and T. Seppäläinen. Variational formulas and cocycle solutions for directed polymer and percolation models. *Comm. Math. Phys.*, 346(2):741–779, 2016.
- [27] N. Georgiou, F. Rassoul-Agha, and T. Seppäläinen. Geodesics and the competition interface for the corner growth model. *Probab. Theory Related Fields*, 169(1-2):223–255, 2017.
- [28] N. Georgiou, F. Rassoul-Agha, and T. Seppäläinen. Stationary cocycles and Busemann functions for the corner growth model. *Probab. Theory Related Fields*, 169(1-2):177–222, 2017.
- [29] N. Georgiou, F. Rassoul-Agha, T. Seppäläinen, and A. Yilmaz. Ratios of partition functions for the log-gamma polymer. Ann. Probab., 43(5):2282–2331, 2015.
- [30] J.-B. Gouéré. Shape of territories in some competing growth models. Ann. Appl. Probab., 17(4):1273– 1305, 2007.
- [31] A. Greven and F. den Hollander. Large deviations for a random walk in random environment. Ann. Probab., 22(3):1381–1428, 1994.
- [32] S. A. Groathouse. Infinite-Length Random Polymers. ProQuest LLC, Ann Arbor, MI, 2023. Thesis (Ph.D.)–The University of Utah.
- [33] O. Häggström and R. Meester. Asymptotic shapes for stationary first passage percolation. Ann. Probab., 23(4):1511–1522, 1995.
- [34] C. Hoffman. Geodesics in first passage percolation. Ann. Appl. Probab., 18(5):1944–1969, 2008.
- [35] C. D. Howard and C. M. Newman. Geodesics and spanning trees for Euclidean first-passage percolation. Ann. Probab., 29(2):577–623, 2001.
- [36] D. Ioffe and Y. Velenik. Stretched polymers in random environment. In *Probability in complex physical systems*, volume 11 of *Springer Proc. Math.*, pages 339–369. Springer, Heidelberg, 2012.
- [37] C. Janjigian, S. Nurbavliyev, and F. Rassoul-Agha. A shape theorem and a variational formula for the quenched Lyapunov exponent of random walk in a random potential. Ann. Inst. Henri Poincaré Probab. Stat., 58(2):1010–1040, 2022.
- [38] C. Janjigian and F. Rassoul-Agha. Busemann functions and Gibbs measures in directed polymer models on Z². 2018. Extended version (arXiv 1810.03580v2).
- [39] C. Janjigian and F. Rassoul-Agha. Busemann functions and Gibbs measures in directed polymer models on Z². Ann. Probab., 48(2):778–816, 2020.
- [40] C. Janjigian and F. Rassoul-Agha. Uniqueness and Ergodicity of Stationary Directed Polymers on Z². J. Stat. Phys., 179(3):672–689, 2020.
- [41] C. Janjigian, F. Rassoul-Agha, and T. Seppäläinen. Ergodicity and synchronization of the Kardar-Parisi-Zhang equation. 2022. Preprint (arXiv:2211.06779).
- [42] C. Janjigian, F. Rassoul-Agha, and T. Seppäläinen. Geometry of geodesics through Busemann measures in directed last-passage percolation. J. Eur. Math. Soc. (JEMS), 25(7):2573–2639, 2023.
- [43] S. A. Kalikow. Generalized random walk in a random environment. Ann. Probab., 9(5):753–768, 1981.
- [44] H. Kesten. Aspects of first passage percolation. In École d'été de probabilités de Saint-Flour, XIV—1984, volume 1180 of Lecture Notes in Math., pages 125–264. Springer, Berlin, 1986.
- [45] A. Krishnan, F. Rassoul-Agha, and T. Seppäläinen. Geodesic length and shifted weights in first-passage percolation. Comm. Amer. Math. Soc., 3:209–289, 2023.
- [46] J. Krug, P. Meakin, and T. Halpin-Healy. Amplitude universality for driven interfaces and directed polymers in random media. *Phys. Rev. A*, 45:638–653, 1992.
- [47] N. Kryloff and N. Bogoliouboff. La théorie générale de la mesure dans son application à l'étude des systèmes dynamiques de la mécanique non linéaire. Ann. of Math. (2), 38(1):65–113, 1937.
- [48] C. Licea and C. M. Newman. Geodesics in two-dimensional first-passage percolation. Ann. Probab., 24(1):399–410, 1996.
- [49] T. M. Liggett. An improved subadditive ergodic theorem. Ann. Probab., 13(4):1279–1285, 1985.

- [50] P.-L. Lions, G. Papanicolaou, and S. Varadhan. Homogenization of Hamilton-Jacobi equation. pages 1-34, 1987. Unpublished manuscript.
- [51] J. Mairesse and B. Prabhakar. The existence of fixed points for the ·/GI/1 queue. Ann. Probab., 31(4):2216-2236, 2003.
- [52] J. B. Martin. Linear growth for greedy lattice animals. Stochastic Process. Appl., 98(1):43–66, 2002.
- [53] J. B. Martin. Limiting shape for directed percolation models. Ann. Probab., 32(4):2908–2937, 2004.
- [54] C. M. Newman. A surface view of first-passage percolation. In Proceedings of the International Congress of Mathematicians, Vol. 1, 2 (Zürich, 1994), pages 1017–1023, Basel, 1995. Birkhäuser.
- [55] F. Rassoul-Agha. The point of view of the particle on the law of large numbers for random walks in a mixing random environment. Ann. Probab., 31(3):1441–1463, 2003.
- [56] F. Rassoul-Agha. On the zero-one law and the law of large numbers for random walk in mixing random environment. *Electron. Comm. Probab.*, 10:36–44 (electronic), 2005.
- [57] F. Rassoul-Agha and T. Seppäläinen. Process-level quenched large deviations for random walk in random environment. Ann. Inst. Henri Poincaré Probab. Stat., 47(1):214–242, 2011.
- [58] F. Rassoul-Agha and T. Seppäläinen. Quenched point-to-point free energy for random walks in random potentials. Probab. Theory Related Fields, 158(3-4):711-750, 2014.
- [59] F. Rassoul-Agha and T. Seppäläinen. A course on large deviations with an introduction to Gibbs measures, volume 162 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2015.
- [60] F. Rassoul-Agha, T. Seppäläinen, and A. Yilmaz. Quenched free energy and large deviations for random walks in random potentials. *Comm. Pure Appl. Math.*, 66(2):202–244, 2013.
- [61] F. Rassoul-Agha, T. Seppäläinen, and A. Yilmaz. Variational formulas and disorder regimes of random walks in random potentials. *Bernoulli*, 23(1):405–431, 2017.
- [62] R. T. Rockafellar. Convex analysis. Princeton Mathematical Series, No. 28. Princeton University Press, Princeton, N.J., 1970.
- [63] T. Seppäläinen and E. Sorensen. Global structure of semi-infinite geodesics and competition interfaces in Brownian last-passage percolation. *Probab. Math. Phys.*, 4(3):667–760, 2023.
- [64] D. J. Slonim. A zero-one law for random walks in random environments on \mathbb{Z}^2 with bounded jumps. 08 2021.
- [65] H. Spohn. KPZ scaling theory and the semidiscrete directed polymer model. In Random matrix theory, interacting particle systems, and integrable systems, volume 65 of Math. Sci. Res. Inst. Publ., pages 483–493. Cambridge Univ. Press, New York, 2014.
- [66] A.-S. Sznitman and M. Zerner. A law of large numbers for random walks in random environment. Ann. Probab., 27(4):1851–1869, 1999.
- [67] L. Tournier. Asymptotic direction of random walks in Dirichlet environment. Ann. Inst. Henri Poincaré Probab. Stat., 51(2):716–726, 2015.
- [68] S. R. S. Varadhan. Large deviations for random walks in a random environment. Comm. Pure Appl. Math., 56(8):1222–1245, 2003. Dedicated to the memory of Jürgen K. Moser.
- [69] A. Yilmaz. Large deviations for random walk in a space-time product environment. Ann. Probab., 37(1):189–205, 2009.
- [70] A. Yilmaz. Quenched large deviations for random walk in a random environment. Comm. Pure Appl. Math., 62(8):1033–1075, 2009.
- [71] A. Yilmaz. Harmonic functions, h-transform and large deviations for random walks in random environments in dimensions four and higher. Ann. Probab., 39(2):471–506, 2011.
- [72] M. P. W. Zerner. Directional decay of the Green's function for a random nonnegative potential on \mathbb{Z}^d . Ann. Appl. Probab., 8(1):246–280, 1998.
- [73] M. P. W. Zerner. The zero-one law for planar random walks in i.i.d. random environments revisited. *Electron. Comm. Probab.*, 12:326–335 (electronic), 2007.
- [74] M. P. W. Zerner and F. Merkl. A zero-one law for planar random walks in random environment. Ann. Probab., 29(4):1716–1732, 2001.

SEAN GROATHOUSE, UNIVERSITY OF UTAH, DEPARTMENT OF MATHEMATICS, 155 S 1400 E, SALT LAKE CITY, UT 84112 Email address: sean.groathouse@utah.edu URL: http://www.math.utah.edu/~sean/

Christopher Janjigian, Purdue University, Department of Mathematics, 150 N University St, West Lafayette, IN 47907

Email address: cjanjigi@purdue.edu URL: http://www.math.purdue.edu/~cjanjigi

FIRAS RASSOUL-AGHA, UNIVERSITY OF UTAH, DEPARTMENT OF MATHEMATICS, 155 S 1400 E, SALT LAKE CITY, UT 84112 Email address: firas@math.utah.edu URL: http://www.math.utah.edu/~firas